

Safe and energy efficient decision/control architecture for a formation of multi-vehicle system in dynamic environment

Lyes Saidi

► To cite this version:

Lyes Saidi. Safe and energy efficient decision/control architecture for a formation of multi-vehicle system in dynamic environment. Automatic. Université de Technologie de Compiègne, 2024. English. NNT : 2024COMP2791 . tel-04670065

HAL Id: tel-04670065 https://theses.hal.science/tel-04670065v1

Submitted on 11 Aug 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Par Lyes SAIDI

Safe and energy efficient decision/control architecture for a formation of multi-vehicle system in dynamic environment

Thèse présentée pour l'obtention du grade de Docteur de l'UTC



Soutenue le 10 janvier 2024 **Spécialité :** Automatique et Robotique : Unité de recherche Heudyasic (UMR-7253) D2791

Safe and Energy Efficient Decision/Control Architecture for a Formation of Multi-Vehicle System in Dynamic Environment

By Lyes SAIDI

Spécialité : Automatique et Robotique

PhD defense on January 10, 2024. In front of the jury composed of:

Reviewers

Michel BASSET Professeur des Universités Univ. Haute-Alsace *Philippe MARTINET* Directeur de recherche Inria Sophia Antipolis

Examiners

Olivier SIMONIN Professeur des Universités INSA Lyon, Univ. de Lyon Maan EL BADAOUI EL NAJJAR Professeur des Universités CRIStAL, Univ. de Lille

Véronique CHERFAOUI Professeure des Universités Heudiasyc, UTC

Supervisors

Lounis ADOUANEReine TALJProfesseur des UniversitésChargée de Recherche CNRS, HDRHeudiasyc, UTCHeudiasyc, UTC

PhD Thesis prepared at the Heudiasyc Laboratory

UMR UTC/CNRS 7253





Abstract

The widespread use of Intelligent Transportation Systems (ITS) primarily stems from the imperative to mitigate human errors contributing to accidents. However, addressing the "individual" Automated Vehicles (AVs) alone remains insufficient, given that a number of situations require the "coordination" of the relative movements of AVs.

Within the Multi-Vehicle System (MVS) paradigm, AVs derive advantages from connectivity and road preview information. Consequently, they can sense more accurately, process more information, and can be more tightly controlled. The collaborative assessment of safety within MVS allows for the establishment of advanced collision-avoidance strategies, particularly in challenging scenarios such as intersection crossings and onramp mergings. Moreover, MVS technology facilitates the reduction of gaps between vehicles, enhancing road capacity and traffic flow. The MVS's shorter response time enables improved control of AV dynamics, paving the way for promising energy-centric strategies.

The main aim of the research done in this Ph.D. manuscript is to propose a safe and energy efficient decision/control architecture for MVS that navigates in dynamic and complex environment such as on-ramp merging and multi-road navigation in highway. Based on the multi-controller foundational control architecture a Cooperative Multi-Controller Architecture (C-MCA) is proposed.

The first part of the proposed C-MCA deals with the decision-making level. This level involves a multi-behavior decision-making strategy, responsible for activating the MVS' suitable behavior based on the safety metric. Two distinct behaviors are proposed: (a) The nominal behavior, designed for executing the merging scenario while adhering to the individual goals of the MVS vehicles. It is activated when no collision risk is detected, (b) The cooperative behavior, activated by the decision-making level when both of the safety and the energy efficiency requirement are not satisfied by the nominal behavior. Its objective is to solve the merging conflict by generating a safe and energy efficient passing order of the MVS vehicles in the merging zone.

The second part of the proposed C-MCA focuses on the local trajectory planning level. Each behavior is assigned a dedicated controller tailored to meet its specific requirements. For instance, the cooperative behavior controller has the responsibility of translating the vehicle's passing order into feasible dynamics (i.e., trajectory, velocity, etc.). The translation task is facilitated through the proposed dynamic formation reconfiguration strategy. In essence, the approach leverages the multi-vehicle system's formation navigation capabilities to conceptualize the merging challenge as a formation reconfiguration problem. A formalization of the formation reconfiguration problem is presented, employing a flexible and generic formal approach. The proposed dynamic formation reconfiguration strategy, employs virtual dynamic targets to ensure a secure reshaping of the formation toward the desired configuration based on the selected passing order. The overall architecture's performance is assessed through co-simulation using Matlab/Simulink and SCANeR studio.

Keywords: Highway cooperative navigation, Multi-vehicle system, Multi-controller architecture, Multi-behavior decision-making strategy, Dynamic formation reconfiguration strategy.

Résumé

L'engouement important pour les systèmes de transport intelligent est justifié principalement par l'impératif de réduire, voire annihiler, les erreurs humaines induisant les accidents. Cependant, s'attaquer uniquement aux Véhicules Autonomes (VAs) "individuels" demeure insuffisant, étant donné que plusieurs situations nécessitent la "coordination" des mouvements relatifs des VAs. Dans le paradigme du Système Multi-Véhicules (SMV), les VAs bénéficient de l'information issue de leur connectivité. Par conséquent, ils peuvent détecter plus précisément, traiter davantage d'informations et être contrôlés de manière plus précise. L'évaluation collaborative de la sécurité au sein du SMV permet l'établissement de stratégies avancées en matière d'évitement de collision, en particulier dans des scénarios complexes tels que les croisements au sein d'intersections et les insertions sur les entrées d'autoroute. De plus, la technologie SMV facilite la réduction des espaces entre les véhicules, améliorant ainsi la capacité et la fluidité du trafic routier. Le temps de réponse plus court du SMV permet un meilleur contrôle de la dynamique des VAs, ouvrant la voie à des stratégies énergétiques prometteuses. L'objectif principal des travaux de recherche constituant ce manuscrit de doctorat est de proposer une architecture de décision/contrôle sûre et peu énergivore pour le SMV naviguant dans des environnements dynamiques et complexes. Inspirée des architectures multi-contrôleurs, une Architecture Multi-Contrôleurs Coopérative est proposée. La première partie de l'architecture proposée concerne le niveau de prise de décision. Ce niveau, impliquant une stratégie de prise de décision à plusieurs comportements, est responsable de l'activation du comportement approprié du SMV en fonction de la métrique de sécurité. Deux comportements distincts sont proposés : (a) Le comportement nominal, conçu pour réaliser le scénario d'insertion tout en respectant les objectifs individuels des véhicules formant le SMV, est activé lorsqu'aucun risque de collision est détecté, (b) Le comportement coopératif est activé par le niveau de prise de décision lorsque l'exigence de sécurité n'est pas satisfaite par le comportement nominal. Son objectif est de résoudre le conflit lors de l'insertion en générant un ordre de passage sûr et économe énergétiquement pour les véhicules composant le SMV dans la zone d'insertion. La deuxième partie de l'architecture proposée se concentre sur le niveau de planification de trajectoire locale. Chaque comportement se voit attribuer un contrôleur dédié, conçu pour répondre à ses besoins spécifiques. Par exemple, le contrôleur du comportement coopératif est chargé de traduire l'ordre de passage des véhicules en dynamiques réalisables (trajectoire, vitesse, etc.). Cette tâche d'obtention de dynamiques réalisables est facilitée par la stratégie de reconfiguration dynamique de la formation proposée. En substance, l'approche tire parti des capacités de navigation en formation du SMV pour conceptualiser des manœuvres coopératives ayant trait d'insertion en milieu autoroutier, comme un problème de reconfiguration de formation. Une formalisation du problème de reconfiguration de la formation est présentée, utilisant une approche formelle, flexible et générique. La stratégie proposée utilise des cibles dynamiques virtuelles pour garantir une reconfiguration sécurisée de la formation, ceci, de sa forme initiale vers la configuration souhaitée par rapport à l'ordre de passage sélectionné. La performance de l'architecture globale a été évaluée par co-simulation en utilisant Matlab/Simulink et SCANeR studio.

Mots-clés : Navigation coopérative en milieu autoroutier, Système multi-véhicules, Architecture multi-contrôleurs, Stratégie multi-comportementales de prise de décision, Stratégie de reconfiguration dynamique d'une formation.

Comme tant de femme avant - et comme tant d'autres aujourd'hui encore -, elle n'a jamais eu de voyage à elle. J'aurais tellement voulu qu'elle puisse suivre la voie qu'elle aimait.

> My life on the road Gloria Steinem **A ma maman**

Peu importe qui nous sommes et où nous vivons, tout au fond, nous nous sentons tous incomplets. C'est comme avoir perdu quelque chose et éprouver la nécessité de le retrouver. Quel est ce "quelque chose" ? La plupart d'entre nous ne le découvrirons jamais. Et parmi ceux qui y parviennent, plus rares encore sont ceux qui partent à sa quête.

> Soufi, mon amour Elif Shafak **A ma chérie**

iv

Acknowledgements

Firstly, I would like to express my gratitude to my thesis advisors, Reine Talj and Lounis Adouane, for their patience, assistance, and the relevance of their guidance. A PhD thesis is a somewhat adventurous experience, with an uncertain outcome; and it is partially thanks to them that I have completed mine.

I am also thankful to Philippe Martinet and Michel Basset for the honor of reviewing this dissertation, as well as to Véronique Cherfaoui and Olivier Simonin, who graciously agreed to contribute their efforts. I must not forget Maan El Badaoui El Najjar, who chaired the committee for my defense. I thank them all for their remarks and constructive criticisms.

Of course, I am grateful to all the fellow colleagues I have encountered during this thesis, in alphabetical order: Emmanuel Alao, Joelle Al Hage, Stéphane Bonnet, Maxime Escourrou, Chan He, Remy Huet, Antoine Lima, and Lahcene Mezouari.

The work presented in this dissertation is the result of three years of effort within the Heudiasyc laboratory, in the Computer Science Department of UTC. I wish to express my gratitude to all the members who contribute to making the work more enjoyable: my laboratory mates, the secretaries, and members of the logistics team, whose assistance I have appreciated.

Lastly, I extend my appreciation to the people who share my 6 pm to 11 pm, the fellow climbers from l'Association des Grimpeurs Compiégnois, Pauline, Emeline, Gérome, and all the other climbers. I am profoundly grateful to share my adventures with Maxime Noizet, Kiryan Devigne, and all my friends; their continuous support during these experiences was a game-changer. I thank my sister and brother for their unconditional care and kindness. To my beloved partner in crime, the being of light that inspires me to always push myself further. And to my beloved mother, my ocean of kindness.

vi

Al	bstra	\mathbf{ct}			i
Ré	ésum	é			i
Ac	cknov	wledge	ments		\mathbf{v}
Li	st of	Figur	es		xi
Li	st of	Table	5		xv
Li	st of	Publi	cations		xvii
G	ener	al inti	roductio	on	1
I	Sta	te of t	the art		9
1	Fro	m indi	vidual A	utomated Vehicles to Multi-Vehicle System	11
-	1.1	Introd	uction		12
	1.2	Auton	omous ve	hicles' technology: overall context	$13^{}$
		1.2.1	Overview	w of the autonomous vehicles' developments	14
			1.2.1.1	Main past / current and coming experimental de-	
				velopments	14
			1.2.1.2	Simulation tools	16
		1.2.2	Autonor	nous vehicle's architecture	18
		1.2.3	Multi-Ve	ehicle System	19
		1.2.4	Overview	w of MVS projects around the world	20
			1.2.4.1	PATH	20
			1.2.4.2	Demo 2000	21
			1.2.4.3	Scania platooning	22
			1.2.4.4	Energy ITS	23
			1.2.4.5	SARTRE	23
			1.2.4.6	Grand Cooperative Driving Challenge	25
	1.3	Coope	rative na	vigation for MVS: scenarios overview	28
		1.3.1	Coopera	tive navigation in urban environment $\ldots \ldots \ldots$	28
		1.3.2	Coopera	tive navigation in highway environment	31
			1.3.2.1	Cooperative platooning navigation on highway	31
			1.3.2.2	Cooperative on-ramp merging on highway	33
	1.4	Conclu	usion		34
2	Coo Arc	operati hitectı	ve Motio 1re Over	on Planning and Decision-Making for MVS: view	35

	2.1	Introduction	36
	2.2	Control architecture for MVS	37
		2.2.1 Perception module	37
		2.2.2 Localization module	38
		2.2.3 Communication module	39
		2.2.4 Decision-making module	40
		2.2.5 Action module \ldots	41
	2.3	Overview of MVS control architecture	43
		2.3.1 Centralized approaches	44
		2.3.2 Decentralized approaches	45
		2.3.3 Hybrid approaches	47
	2.4	MVS navigation in formation: a review	49
		2.4.1 Leader-Follower \ldots	50
		2.4.2 Virtual Structure	52
		2.4.3 Behavior-based \ldots	53
		2.4.4 Consensus-based control \ldots	54
		2.4.5 Discussion \ldots	55
	2.5	MVS Navigation in complex environments	56
		2.5.1 Highway collaborative navigation	59
		2.5.1.1 Cooperative Adaptive Cruise Control $(CACC)$	60
		$2.5.1.1.1 \text{Optimal Control} \dots \dots \dots \dots \dots$	61
		$2.5.1.1.2 \text{String stability} \dots \dots \dots \dots \dots$	62
		2.5.1.1.3 Model Predictive Control (MPC):	63
		2.5.1.2 Evaluation of the highway collaborative naviga-	
		tion approaches	64
		2.5.1.2.1 Traffic throughput	64
		$2.5.1.2.2 \text{Energy efficiency} \dots \dots \dots \dots \dots$	65
		$2.5.1.2.3 \qquad \text{Negotiation} \dots \dots \dots \dots \dots \dots \dots \dots \dots $	66
		2.5.2 Collaborative on-ramp merging on highway	67
		2.5.2.1 Traffic throughput	68
		2.5.2.2 Energy efficiency	70
		$2.5.2.3 \text{Negotiation} \dots \dots \dots \dots \dots \dots \dots \dots \dots $	71
	2.6	Conclusion	71
II	Co	poperative Multi-Controller Architecture	73
		-	
3	The	e proposed Cooperative Multi-Controller Architecture (C-	
	MC	CA)	75
	3.1	Introduction	76
	3.2	Multi-controller architecture	76

	3.3	Overa	ll Cooperative Multi-Controller Architecture	78
		3.3.1	The C-MCA main functionalities	78
			3.3.1.1 Environment perception	78
			3.3.1.2 Communication module	80
			3.3.1.3 Global path planning	80
			3.3.1.4 Decision-making level	80
			3.3.1.5 The local trajectory planning level	82
			3.3.1.6 The control level	82
	3.4	Proble	em statement	83
		3.4.1	Scenario topology	83
		3.4.2	Control law	84
		3.4.3	Formation definition and configuration	86
			3.4.3.1 Formation modeling based on a dynamic reference	
			frame	87
			3.4.3.2 Formation modeling based on the Frenet reference	
			frame	88
	3.5	Conclu	usion	89
4	D	.	E-matin Deconfermation for On Demo Maning	01
4		Iamic .	Formation Reconfiguration for On-Ramp Merging	91
	4.1	Const	netron Antimal Papanformation Matrix (COPM)	92
	4.2	4.9.1	The inter target distance matrix	94 05
		4.2.1	Constrained inter target distance matrix	95 07
		4.2.2	4.2.2.1 Constrained antimal reconfiguration matrix	91
			4.2.2.1 Constrained optimal reconfiguration matrix	90
		193	Simulation results	102
		4.2.3	Conclusion on the COPM algorithm	100
	12	Fyton	ded Dynamia Reconfiguration Matrix (F COPM)	110
	4.0	131	Extended inter target distance matrix	110
		4.0.1	4.3.1.1 Boconfiguration gains	110
			4.3.1.2 Stability analysis w r t the convergence errors	112
		132	Trajectory segmentation strategy	112
		4.3.2	Simulation regults	115
		4.5.5	Conclusion of the E COPM algorithm	117
	4.4	4.3.4 Formo	tion Reconfiguration Approach based on an Online Control	111
	4.4	Strato	(FRA OCS)	110
			FRA OCS formalism	120
		4.4.1	Online reconfiguration going identification and velocity pro	120
		4.4.4	file generation	199
		112	Simulation results	125
		4.4.0	Conclusion of the FRA OCS algorithm	120
		4.4.4		140

	4.5	Chapter conclusion	. 129
5	Mu	lti-Behavior Decision-Making Strategy	131
	5.1	Multi-behavior selection strategy	. 132
	5.2	Nominal behavior	. 134
		5.2.1 Overview of the nominal behavior part of the C-MCA	. 134
		5.2.2 Simulation results	. 136
	5.3	Cooperative behavior	. 138
		5.3.1 Passing sequence selection	. 139
		5.3.2 Suitable passing sequence	. 141
		5.3.2.1 Global cooperation cost function	. 143
		5.3.2.2 Altruistic passing sequence	. 145
	5.4	Simulation results	. 146
	5.5	Conclusion	. 155
G	ener	al conclusion and future works	157
\mathbf{A}	pper	ndices	163
Bi	iblio	graphy	171

List of Figures

1.1	SAE autonomy levels [118]	13
1.2	SAE autonomy levels ODD's [119]	14
1.3	A brief review of the Autonomous Vehicles state developments in the	
	last two decades, adapted from [271]	17
1.4	AV's architecture [173]	18
1.5	Highway cooperative navigation part of the PATH project [70]	21
1.6	Platoon navigation under the DEMO 2000 project [93]	22
1.7	Scania platooning [image courtesy of Scania]	23
1.8	Configuration of the automated truck part of the Energy ITS project	
	[227]	24
1.9	SARTRE project demonstration [image courtesy of the Volvo car cor-	
	poration]	24
1.10	GCDC 2011 [image courtesy of IEEE]	26
1.11	GCDC 2016 [image courtesy of IEEE]	26
1.12	MVS applications: an overview	28
1.13	Cooperative intersection crossing	29
1.14	Cooperative roundabout crossing	30
1.15	Cooperative merging on highway	32
1.16	Cooperative adaptive cruise control: platooning navigation	32
1.17	Cooperative on-ramp merging on highway	33
2.1	Reference control scheme for MVS, adapted from [238]	38
2.2	Communication topologies, adapted from [192]	40
2.3	Control architecture for the navigation in formation of MVS [238]	41
2.4	General cooperative architecture composed of: communication, per-	
	ception, localization, planning and control modules [248]	44
2.5	Example of centralized architecture [238][47]	45
2.6	Example of decentralized architecture [238][181]	47
2.7	Example of centralized/decentralized architecture [238][210]	48
2.8	The leader follower formation modeling approach [238]	50
2.9	Leader-follower based on spring damper system [238]	51
2.10	Virtual structure formation modeling approach [238]	52

2 11	Behavior-based formation modeling approach [238]	54
2.12	From communication topology to algebraic graph representation for	01
2.12	consensus-based control	55
2 13	Bural freeway	57
2.14	Intersection crossing	57
2 15	Highway navigation	57
2.16	Highway on-ramp	57
2.10 2.17	Truck platooning [263]	58
2.18	On-ramp merging on highway	58
2.10	Platoon-based navigation in formation on highway	59
2.10	Cooperative Adaptive Cruise Control	60
2.20	Negotiation-based merging in highway	66
2.22	Centralized collaborative on-ramp merging on highway	69
2 23	Decentralized collaborative on-ramp merging on highway	70
2.20		••
3.1	Multi-Controller Architecture	77
3.2	Cooperative Multi-Controller Architecture	79
3.3	Flowchart of the decision-making level part of the C-MCA	81
3.4	Illustration of the scenario topology of the on-ramp merging on highway	84
3.5	Vehicle's and target's configuration in global $(X_G; Y_G)$ and local $(X_m; Y_m)$)
	reference frames, and the control variables [238]	85
3.6	Coordinates system based on the Frenet reference frame	88
4.1	The virtual structure approach used to model the formation and its	
	reconfiguration to perform the merging maneuver. (a) The initial	
	shape of the formation and its coordinates. (b) The final shape of the	0.9
4.0	formation after the merging maneuver and its desired coordinates.	93
4.2	Illustration of the merging scenario where the minimum inter-distance	06
4.9	\underline{D}_T is depicted	90
4.3	The projection of I_{d_i} w.r.t. the reference trajectory	99
4.4	for the proposed CORM (Constrained Optimal Reconfiguration Matrix)	
	now chart. (1) The optimization algorithm. (2) The projection and	100
15	Figure and the formation shape during the marging performed by the	100
4.0	COPM from encode (Simulation shape during the merging performed by the	109
16	The formation georginates and the Fuelidean in hot way distances	105
4.0	refles	105
17	The vahiales' valoaity and acceleration profiles	106
4.1 1 Q	Illustration of the on ramp morging scenario based on the read cor	100
4.0	musuation of the on-ramp merging scenario based on the road seg-	112
10	Coordinates selection based on the traveled segment	114
4.3	Coordinates selection based on the traveled segment	114

4.10	Evolution of the formation shape based on the E-CORM reconfigura-	
	tion (Simulation video: https://youtu.be/-h97Qz7K3F4)	116
4.11	The formation's coordinates	117
4.12	The Euclidean in-between distances	118
4.13	The vehicles' velocity profiles	119
4.14	Snapshots of the evolution of the formation shape during the on-ramp	
	merging scenario obtained with the help of SCANeR studio (Simula-	
	tion video: https://youtu.be/3gDSq5L3m-c)	126
4.15	The formation's coordinates	127
4.16	The Euclidean in-between distances	128
4.17	The vehicles' velocity profiles	128
51	The proposed multi-mode behavior selection level part of the overall	
0.1	C-MCA (cf. Figure 3.2)	133
52	Illustration of the on-ramp merging on highway scenario seen by the	100
0.2	nominal behavior	135
5.3	Illustration of the on-ramp merging scenario at the vehicles initial	100
0.0	state (Simulation video: https://voutu.be/8Cgw7GEFKd4)	136
5.4	Nominal behavior: progress of the Euclidean in-between distances	137
5.5	Nominal behavior: The vehicles' velocity and acceleration profiles	138
5.6	Flowchart of the prediction based on the nominal behavior	140
5.7	Image of the merging scenario at $t = CT$ (Collision Time)	140
5.8	The flowchart of the MVS passing sequence in the merging zone selec-	
	tion strategy	142
5.9	Scenario 1: Illustration of the on-ramp merging on highway scene.	
	The initial shape of the formation under the communication range of	
	the RSU is represented, along with the desired shape	147
5.10	Evolution of the formation' shape during the on-ramp merging on	
	highway	148
5.11	Euclidean in-between distances within the formation	149
5.12	The vehicles' velocity and acceleration profiles	150
5.13	Scenario 2: Illustration of the on-ramp merging on highway scene.	
	The initial shape of the formation under the communication range	
	of the RSU is represented, along with the desired shape (Simulation	
	video: https://youtu.be/u0LRJbOW94M)	151
5.14	Illustrative snapshots obtained from the SCANeR studio engine pre-	
	senting the evolution of the formation during the merging maneuver .	152
5.15	Formation coordinates during the on-ramp merging on highway	154
5.16	The Euclidean in-between distances	155
5.17	The vehicles' velocity profiles	155
.18	Road Side Unit (RSU)	165

.19	The overall simulation architecture	
.20	The overall simulation architecture	

List of Tables

1.1	Summary of the major project involving MVS
2.1	Comparison between MVS architectures [10]
4.1 4.2	The values of the inputs of the CORM algorithm $\ldots \ldots \ldots$
$5.1 \\ 5.2$	Scenario 1: Numerical results of the cooperation behavior $\ldots \ldots \ldots 148$ Scenario 2.1: Numerical results of the cooperation behavior before $t = 8s \ldots $
5.3	Scenario 2.2: Numerical results of the cooperation behavior after $t = 8s153$

List of Publications

International congress with proceedings

- [ITSC'23] Lyes Saidi, Reine Talj and Lounis Adouane, On-ramp Merging on Highway for Cooperative Automated Vehicles based on an Online Reconfigurable Formation Control Approach. IEEE International Conference on Intelligent Transportation Systems, 24th-28th September 2023, In Bilbao, Spain.
- [MMAR'23] Lyes Saidi, Lounis Adouane and Reine Talj. Altruistic Coordination Strategy for On-Ramp Merging on Highway of a Formation of Cooperative Automated Vehicles. International Conference on Methods and Models in Automation and Robotics, 22th-27th August 2023, In Międzyzdroje, Poland.
- [ITSC'22] Lyes Saidi, Lounis Adouane and Reine Talj, CORM: Constrained Optimal Reconfiguration Matrix for Safe On-Ramp Cooperative Merging of Automated Vehicles. IEEE International Conference on Intelligent Transportation Systems, from 18th September to 18th October 2022, In Macau, China.

International congress

• [VAMS'23] Lyes Saidi, Lounis Adouane and Reine Talj, *Cooperative Decision-Making for Safe On-Ramp Merging on Highway for Connected Automated Vehicles.* International Symposium on the Verification of Autonomous Mobile Systems, 9th-10th March 2023, In Paris, France.

National congress

- [CT ATT'22] Lyes Saidi, Lounis Adouane and Reine Talj, Safe and Smooth Onramp Merging on Highway Strategy for Cooperative Automated Vehicles. Journées du Comité Technique Automatique et Transport Terrestre, 5th-6th April 2022, In Valenciennes, France.
- [JJCR21] Lyes Saidi, Lounis Adouane and Reine Talj, *Toward a Robust and Safe Cooperative Highway Navigation of Multi-Vehicles Systems.* Journée des Jeunes Chercheurs en Robotique, 12th October 2021, In Paris, France.

xviii

General Introduction

1. Context of the PhD thesis

The transportation systems have provided humanity invaluable social and financial benefits, yet they are also intricately linked to adverse outcomes. The widespread use of automobile, while offering convenience, has given rise to challenges such as traffic congestion and an increase in traffic accidents. The investigation of accident causes has been a focal point of rigorous research for many years. According to [243], about 85% of accidents can be attributed to human errors. Within this percentage, over 50% are linked to decision errors, wherein drivers receive all relevant environment information but either refrain from making a decision, or choose an incorrect course of action. The remaining 35% is attributed to inattention errors, where drivers are distracted, fail to respond, or lose control of their vehicles.

A promising avenue to mitigate the majority of human errors is through the automation of the driving tasks. In recent years, the development of fully autonomous vehicles (AVs) for transportation has garnered significant attention from various researchers and industrials globally [79][197][204]. The fundamental aim of autonomous vehicles is to surpass human perceptions of safety and judgment, particularly in perilous situations, and to produce appropriate actions [116]. Driving, being a complex task, demands the autonomous vehicle's ability to make critical decisions, achieved through various intra-vehicle process (e.g., situation awareness and assessment, decision-making, etc.). Moreover, an autonomous vehicle operates within a dynamic driving environment, alongside other road users, adding complexity to the driving task.

The predominant focus of current autonomous vehicle research revolves around enabling *individual* autonomous vehicles to navigate roads safely. However, there is a growing acknowledgment that to take fully advantage from the benefits of autonomous vehicle technology, certain scenarios will require the *coordination* of relative activities and movements [199]. An intriguing area of research is autonomous navigation with a group of vehicles, known as Multi-Vehicle System (MVS). The advantages of MVS span multiple domains, including safety through accident reduction, enhanced passenger comfort promoting health, reduced transportation time by alleviating road congestion, and improved energy efficiency contributing to ecological benefits, among others. Nonetheless, some challenging problems need to be completed to ensure safe coordination between the vehicles in MVS.

2. Main objectives

The focus of the research work done in this PhD manuscript corresponds to the subject of MVS coordination ability in what is known in the literature as challenging scenarios [152]. In other terms, in scenarios such as intersection crossing or on-ramp/off-ramp merging on highway, orchestrating the vehicles' motions becomes a necessity to guarantee the on-road safety. Consequently, the main aim of this PhD work is to study the different topics related to vehicles' motion coordination (i.e. architectures, mechanisms, approaches), to propose a Safe and Energy Efficient Decision/Control Architecture for a Formation of MVS in Dynamic Environment. To accomplish this aim, three main objectives were identified.

2.1 Multi-vehicle system for challenging scenarios

Most current industrial and applied research in the area of autonomous vehicles concerns the methods and tools to enable individual AVs to hit the road safely. However, a number of situations will compulsory require coordinating the relative motions of the AVs to ensure the zero-collision requirement [152]. For instance, on-ramp merging on highway can be categorized as a challenging where coordination is needed. The highway vehicles evolve in a highly dynamic environment imposed by the highway traffic nature; and conflicting trajectories among the merging vehicles, the vehicles navigating in the secondary on-ramp road, and the highway vehicles are reportedly common [195].

With the help of the MVS cooperation ability, it is aimed to improve in one hand on-road safety while solving conflicting and unsafe trajectories, and on the other hand, to improve traffic flow by increasing on-road penetration and avoiding bottlenecks (e.g. optimizing the number of vehicles on the merging zone). Undeniably, energy efficiency and passenger comfort are a direct benefit of high flow rate, less idling zones and smooth velocity profiles.

On-road navigation is not a simple task especially when it has to deal with complex and dynamic scenarios. The first challenge is to take into account the nature of the navigation environment in the definition of the original problem of MVS control, while addressing the complexity of the driving scenario. Coordinating the motion of the vehicles part of the MVS (also known as formation control) is also one full part challenge that needs to be addressed.

2.2 Multi-vehicle system navigation in formation

Safe and reliable formation control method permits to reduce the in-between distance among the MVS, which has a direct consequence on the traffic flow since road capacity is harvested at its best. Vehicles are subject to aerodynamic drag and energy is consumed to face this counter-force. One way to improve the energy consumption is to act on the spacing policy of the formation, in fact, reducing the inter-vehicles distance reduce the aerodynamic drag, consequently energy saving can be part of the benefits of MVS navigation in formation.

Formation control is also an important part of the traffic management because of its safety and comfort related benefits. In fact, formation control permits to synchronize the motion of the vehicles in order to avoid conflicting trajectories. Long term and short synchronization can be also part of the formation control strategy with the help of formation reconfiguration. Thus, the vehicle motion can be built to impose the minimum vehicle dynamic changes w.r.t. a certain reference dynamic (e.g., minimize the acceleration and the deceleration of the vehicles to travel at a constant speed), what improves the passenger comfort.

For human driven vehicles, the creation and the navigation in formation is done effortlessly. However, for MVS, the formation control requires a safe and reliable strategy to avoid rear-end collisions, beside the question of the stability that needs to be reliably proven. The first challenge part of the MVS formation control problem is related to formation modeling. Since it is aimed to build a safe and reliable formation control approach, the formation needs to be formulated following mathematical models to facilitate the verification of the formation stability.

The second challenge behind the formation control approach is through its interactions with the decision-level. A strong connection in-between the formation control approach and the decision-making level needs to be established. The latter helps to ensure that the MVS formation is controlled to track a safe goal, and responds quickly to avoid any risk due to the dynamic nature of the highway environment.

2.3 Cooperative and altruistic multi-criterion decision-making level

In one hand, it is proposed to take advantage of the navigation in formation part of the MVS paradigm to tackle the coordination maneuver related to the on-ramp merging on highway scenario. In the other hand, MVS technology is harnessed to capitalize on its benefits, including improved safety, enhanced passenger comfort, and optimized energy efficiency.

However, effective MVS navigation requires support of a suitable decisionmaking level. Fundamentally, the primary responsibility of the decision-making level is to define the formation goal, and ensuring the safety and feasibility of this goal falls within its purview.

In the context of the targeted decision/control architecture, the decisionmaking level's task goes beyond merely selecting a safe and feasible formation goal. Since the aim is to fully exploit the MVS capabilities, the first challenge in decision-making is how to incorporate the following decision criteria: (a) safety, (b) passenger comfort, and (c) energy efficiency. These three criteria must be seamlessly integrated into the decision-making level, with their respective contributions meticulously balanced.

An MVS comprises multiple vehicles that decide to navigate together for a specific duration. In this PhD thesis, the MVS vehicles are supposed to be altruistic. In other terms, under the cooperation paradigm, the vehicles accept to participate by making reasonable efforts such as the MVS performs the merging maneuver. Consequently, while the MVS has its overarching global goal, each vehicle within the MVS has its individual goal. For example, in the case of on-ramp merging on highway performed by an MVS, the MVS goal is to execute the merging maneuver without causing accidents within the merging zone. However, the individual goals of the vehicles may differ, depending on their roles as highway or merging vehicles. Thus, the second goal of the decision-making level is to select the formation goal, for instance performing the MVS merging maneuver safely, while ensuring the MVS performance efficiency (e.g., passenger comfort, energy efficiency). In addition of taking into account the individual contributions of each vehicle within the MVS toward its global goal, and avoiding the selection of unfair goals with respect to the vehicles' participation.

3. Manuscript outlines and contributions

According to the above mentioned PhD context and objectives, the ultimate goal of this PhD thesis is to propose a safe and energy efficient decision/control architecture for a formation of MVS in dynamic environment. The dedicated endeavor

to achieve this goal is transcribed in this manuscript, which is comprehensibly divided into two parts. The outline of the PhD manuscript is depicted in Figure 1. More precisely, the first part, which contains two chapters, targets to discuss the state of the art - in particular the current research gap for autonomous vehicles cooperative motion planning and decision-making for MVS in a complex dynamic environment such as on-ramp merging on highway - as follows:

• Chapter 1 - From individual automated vehicle to multi-vehicle system

Given that the MVS is composed of multiple AVs, this chapter initiates with a concise overview of the AV technology; laying the groundwork of the subsequent discussion on MVS. It then delves into the central focus of this chapter, namely, the MVS paradigm. Following the presentation of the key concepts related to the MVS, this chapter explores projects in which MVS is actively engaged. These projects serve as examples of technological advancements and innovations aimed at enhancing the MVS capabilities when navigating in on-road environments. Furthermore, the chapter delves into the cooperative navigation of MVS, examining scenarios from a comprehensive perspective.

Chapter 2 - Cooperative motion planning and decision-making for MVS: architecture overview

This chapter aims to offer a detailed exploration of the MVS control architecture by delving into its modules and different processes. Furthermore, it provides insights into paradigms utilized in the construction of the MVS control architecture. Since formation control plays a crucial role in the aimed decision/control architecture, the chapter reviews and discusses various formation control approaches. Finally, it presents a literature overview, emphasizing MVS applications within complex environments. An evaluation through the prism of traffic throughput, energy efficiency and negotiation is proposed.

The last part of the PhD manuscript concentrates on the primary approaches and proposals addressed in this dissertation, which can be categorized into three chapters (cf. Figure 1):

• Chapter 3 - Overview of the proposed Cooperative multi-controller architecture

The central theme of this chapter revolves around the introduction and exploration of the proposed Cooperative Multi-Controller Architecture (C-MCA). It starts with a concise overview of the background of multi-controller architecture to establish the context for delving into the intricacies of the



FIGURE 1: The outline of the PhD manuscript

proposed C-MCA. The chapter extensively discuss the various modules within the C-MCA, with a particular emphasis on the decision-making and planning levels, as these are pivotal in accomplishing the goals outlines on the decision/control architecture. In addition to the foundational information related to the PhD thesis problem statement.

• Chapter 4 - Dynamic formation reconfiguration of on- ramp merging

The chapter initiates with an introduction of the Constrained Optimal Reconfiguration matrix (CORM) algorithm, a key element for formation control, accompanied by pertinent simulation results.

Than the chapter focuses on providing an in-depth exploration of the Extended - Constrained Optimal Reconfiguration Matrix (E-CORM). It initiates with a concise overview of the limitations associated with the CORM, the chapter focuses on the enhancements made to overcome these limitations through the introduction of the E-CORM. The algorithm associated with the E-CORM is elucidated, and subsequent attention is given to the presentation of the simulation results.

The latter part of the chapter is dedicated to evaluating the limitations of both the CORM and the E-CORM to devise a suitable formation reconfiguration strategy. Within this context, a novel optimization-free formation reconfiguration strategy, named Formation Reconfiguration Approach base on an Online Control Strategy (FRA-OCS), drawing inspiration from both the CORM and the E-CORM, is introduced in details. The chapter culminates with the presentation of simulation results for this newly proposed formation reconfiguration strategy.

• Chapter 5 - Multi-behavior decision-making level

This chapter is specifically focused on elucidating the intricacies of the decision-making level within the proposed C-MCA for MVS engaged on on-ramp merging on highways. The multi-behavior selection strategy, an integral part of the decision-making level, is expounded upon through the comprehensive presentation of the developed behaviors. Additionally, the chapter delves into the bi-directional link between the decision-making level and the formation control level, providing detailed insights into the selection of the passing order of the vehicles in the merging zone. The effectiveness of the proposed multi-behavior decision-making level is evaluated through the presentation of various simulation results.

General conclusions are summarized at the end of the manuscript, additionally, it outlines potential avenues for future research related to this PhD thesis.

8_____

Part I State of the art

CHAPTER

From individual Automated Vehicles to Multi-Vehicle System

Abstract

This chapter is dedicated to expounding the evolution of the Multi-Vehicle System (MVS). The central emphasis is on cooperative technologies tailored for a fleet of Autonomous Vehicles (AVs), navigating in complex and dynamic on-road environment. The chapter commences with a brief overview of AV technologies, providing a foundation for the MVS discussion. It then delves into the MVS paradigm.

Contents

1.1	Introd	uction	
1.2	Auton	omous ve	hicles' technology: overall context
	1.2.1	Overview	w of the autonomous vehicles' developments \ldots 14
		1.2.1.1	Main past / current and coming experimental de-
			velopments
		1.2.1.2	Simulation tools
	1.2.2	Autonor	nous vehicle's architecture
	1.2.3	Multi-Ve	ehicle System
	1.2.4	Overview	w of MVS projects around the world 20 $$
		1.2.4.1	PATH
		1.2.4.2	Demo 2000
		1.2.4.3	Scania platooning

		1.2.4.4	Energy ITS	23
		1.2.4.5	SARTRE	23
		1.2.4.6	Grand Cooperative Driving Challenge	25
1.3	Coope	erative na	vigation for MVS: scenarios overview	28
	1.3.1	Coopera	tive navigation in urban environment	28
	1.3.2	Coopera	tive navigation in highway environment	31
		1.3.2.1	Cooperative platooning navigation on highway	31
		1.3.2.2	Cooperative on-ramp merging on highway	33
1.4	Concl	usion		34

1.1 Introduction

With the development of the transportation system in general and the automobile industry in particular, more and more vehicles are on the roads. It is estimated that currently there are about 1.47 billion operational vehicles worldwide, including 1.06 billion passenger vehicles and 363 billion commercial vehicles [3][1]. The number is expected to reach 2 billion vehicles by 2035 [2]. As a result, a series of critical issues are becoming more serious in modern transportation systems, such as road safety, road traffic congestion, energy waste, and pollution [121]. According to a study conducted by the US National Highway Traffic Safety Administration, it was found that human error accounts for anywhere between 94% and 96% of all accidents [91]. On-road safety is a cornerstone in the development of the transportation system. Taking these alarming statistics into account, the automotive industry has itself a mission to reduce the accidents number and this by addressing the human errors.

One effective approach to significantly reduce human errors is the automation of high-risk driving tasks. Before delving into the solutions used to mitigate human errors in on-road environments, it is valuable to establish a broad framework for unmanned based technologies. In general, autonomous and Unmanned Vehicle Systems [11] are categorized according to their operating domain with the most representative being: Autonomous Underwater Vehicles, Unmanned Surface Vehicles, Unmanned Maritime Vehicles, Unmanned (or Uninhabited) Aerial Vehicles, and Unmanned Ground Vehicles (UGVs). Moreover, UGVs may move using various devices (e.g., legs, wheels). In particular, the problem and primary works in this PhD manuscript are interpreted mainly in terms of Autonomous vehicles within the multi-vehicle system, the latter are considered as UGVs with wheels.
1.2 Autonomous vehicles' technology: overall context

Indeed, one of the most important issues of UGVs research is autonomous vehicles (also known as self-driving car, driver-less car or robotic cars) for public/private transit.

Beside the improvement of the driving safety by reducing the participation of the human error consequences. Autonomous Vehicles (AVs) have the advantage of relieving the drivers from the duty of driving and paying attention, thus making it possible to exploit travel time in other activities. AVs reduce notably the number of circulating vehicles (e.g., ride-sharing applications), and thanks to route optimizations, will reduce road congestion, improve traffic flow and reduce traffic pollution [152].

The society of Automotive Engineers (SAE) created the J3016202104[118], which is a standard of six levels grading for the driving automation (cf. Figure 1.1), Ranking from no driving automation (level 0) to full driving automation (level 5). Level 3 categorizes conditional driving automation, as for level 4, it defines high driving automation. These two levels require a human driver in a



Figure 1.1. SAE autonomy levels [118]



supervisory capacity who is prepared to retake control in certain situations.

Figure 1.2. SAE autonomy levels ODD's [119]

Additionally, each level 3 and level 4 vehicle will have its own Operational Design Domain (ODD) (cf. Figure 1.2) [118][119]. Consequently, autonomous vehicles are favored to travel freely inside some limited locations, such as vast parking lots, and to execute rural or urban logistics areas.

1.2.1 Overview of the autonomous vehicles' developments

A brief overview of AVs' development during the last two decades is given in Figure 1.3. This section is dedicated to draw an overview of the main past / current and coming experimental developments related to the AVs' technology. Additionally, it presents the main simulation tools used to test and validate the AVs' developments.

1.2.1.1 Main past / current and coming experimental developments

In order to spur technology for the development of AVs, the Defense Advanced Research Projects Agency (DARPA) organized three competitions in the last two decades. The first, named DARPA Grand Challenge, was realized at the Mojave Desert, USA, in 2004, and required self-driving cars to navigate a 142 miles (228.53 km) long course throughout desert trails within a 10 h time limit. All competing cars failed within the first few miles [29].

The DARPA Grand Challenge was repeated in 2005 and required self-driving cars to navigate a 132 miles (212.43 km) long route through flats, dry lake beds, and mountain passes, including three narrow tunnels, and more than 100 sharp left and right turns. This competition had 23 finalists and 4 cars completed the route within the allowed time limit. The Stanford University's cars, Stanley [221], claimed first place, and the Carnegie Mellon University's cars, Sandstorm and H1ghlander, finished in second and third places, respectively.

The third competition known as the DARPA Urban Challenge [45], was held at the former George Air Force Base, California, USA, in 2007. It required self-driving cars to navigate a 60 miles (96.56 km) long route throughout a simulated urban environment, together with other self-driving and human driven cars, within a 6h time limit. The cars had to obey California traffic rules. This edition had 11 finalists and 6 cars completed the route within the allocatted time limit. The Carnegie Mellon University's car, Boss [233], claimed first place, the Stanford University's car, Junior [84], finished in second, and the Virginia Tech's car, Odin [220], came in third place. Even though these competitions presented challenges much simpler than those typically seen in everyday traffic, they have being hailed as milestones for the development of self-driving cars.

Since the DARPA challenges, many self-driving car competitions and trials have been performed. Relevant examples include: The European Land - Robot Trial (ELROB) [224], the Intelligent Vehicle Future Challenge [246], the Auto-Drive Challenge [117] and many more. At the same time, research on self-driving cars has accelerated in both academia and industry around the world. Notable examples of universities conducting research on self-driving cars comprise Stanford University (Stanley, winner of DARPA 2005), Carnegie Mellon University (Boss, winner of the DARPA 2007). On the other hand, technology companies and scientific research institutions have massively invested in the field. For example, Google, an internet company, which is one of the leaders in self-driving cars, based on its solid foundation in artificial intelligence. So far, Google vehicles have accumulated more than 3.2 million km of tests, becoming the closest to the actual use [129]. Alphabet's Waymo recently unveiled an SAE level 4 self-driving taxi service in Arizona. However, a certain number of situations are still to improve, where the waymo cars still need to manage [8].

Another company that has made great progress in the field of self-driving cars is Tesla. Tesla was the first to claim the industrialization of the self-driving car technology. Tesla's auto-pilot technology has made major breakthroughs in recent years. Although the Tesla's auto-pilot technology is only regarded as SAE level 2 by the national highway traffic safety administration (NHTSA) [8].

In addition to the companies mentioned above, lots of Internet companies and car companies worldwide are also focusing on the self-driving field recently. For example, the Canadian automotive supplier Magna has developed the MAX 4 technology to enable SAE level 4 capabilities in both urban and highway environments.

In Sweden, Volvo and Autoliv established a joint company-Zenuity, which is committed to the security of self-driving cars [64]. In South Korea, Samsung received approval from the South Korean government to test its driver-less cars on public roads. In China, Baidu deep learning institute led the research project of self-driving cars in 2013. In 2014, Baidu established the automotive networking business division and successively launched CarLife, My-Car and CoDriver [223]. Recently, Baidu and Volvo, announced a strategic partnership to jointly develop SAE level 4 electric vehicles that will serve the robotaxi market in china [218].

1.2.1.2 Simulation tools

Rigorous and comprehensive testing plays a key role in training self-driving cars to handle a variety of situations, where they are expected to see on public roads. The physical testing on public roads is unsafe, costly, and not always reproducible. With the craze around the self-driving technology, many simulators were developed to simplify and accelerate the training of self-driving cars [127]. For example MATLAB/Simulink self-driving vehicle team developed a toolbox that provides algorithms and tools for designing, simulating, and testing ADAS and autonomous driving systems [159]. The capability of co-simulation with other softwares like CarSim [52] makes it easier to build various vehicle models. The user personalizes the vehicle model according to its preferences in CarSim and rely on MATLAB/Simulink to control the generated model using co-simulation. MATLAB/Simulink can also be used in co-simulation with SUMO [216] to create scenarios that deal with traffic flow. Sofwars like SCANeR Studio can be used in co-simulation with MATLAB/Simulink in order to build a more established vehicle's model and complex scenarios [200].

Further, Gazebo [96] is a popular robotic simulator, and it is known for its high flexibility and its seamless integration with ROS [196]. It does include various sensors models and it allows users to create new sensors models. The creation of the simulation is manual, where the user creates a 3D models and carefully defines their physics. Simulators relying on 3D engine such as CARLA provide a high-quality simulation environments that require GPU computing unit in order to run with reasonable performance and frame rate. The user can invoke different facilities in CARLA [50] via a flexible API. According to the survey in [127], the authors advice CARLA for end-to-end functionalities testing that self-driving car offer such as perception, mapping, localization, and vehicle control.



Figure 1.3. A brief review of the Autonomous Vehicles state developments in the last two decades, adapted from [271]

1.2.2 Autonomous vehicle's architecture

The overall technical framework of self-driving cars shown in Figure 1.4 that are equipped with SAE level 3 or higher autonomy system can be divided into four parts, namely the driving environment perception system, the autonomous decision system, the control execution system and the monitoring system [173].



Figure 1.4. AV's architecture [173]

The environment **perception** system utilizes the prior knowledge of the environment to establish an environmental model including obstacles, road structures, and traffic signs through obtaining surrounding environmental information. The main function of the environment perception system is to realize functions like lane detection, traffic signal detection, and obstacles detection, by using some hardware devices such as cameras, LiDARs and radars.

The main function of the autonomous **decision** system is to make some decisions for the self-driving car, including obstacle avoidance, path planning, navigation, and so on. For example, in the path planning, the autonomous decision system plans a global path according to the current localization and the target location firstly, then reasonably plans a local path for the self-driving car by combining the global path and the local environment information provided by the environment perception system.

The **control** execution system's function is to generate the commands needed to meet the objectives given by the decision system, such as braking, steering, and accelerating to guarantee the speed control and path-following control.

The **monitor** system is responsible to check whether the car is making actual progress towards its goal and reacts with recovery actions when meeting problems like unexpected obstacles and faults.

1.2.3 Multi-Vehicle System

Most of the current industrial developments and applied researches in the area of autonomous vehicles concern the methods and tools to enable *individual* autonomous vehicles to hit the road safely. However, it is getting increasingly recognized that, to get full advantage from the autonomous vehicle technology, a number of situations will compulsory require *coordinating* the relative activities and movements of the vehicles [6] [199].

Consequently, a natural extension of an *individual* autonomous vehicle is to deal with MVS, with the help of *coordination*. MVS, which have shown immense promise as a component of future transportation systems [54] [92] [212], will be further discussed in the rest of the section. Not that the proposed MVS makes the assumption that connection is always available, either locally or globally. MVS provide appealing benefits in terms of operating in constrained environment, resolving problems concurrently, considerably reducing accidents and related costs.

Around the globe, car manufacturers and research institutes are starting to move away from the vision of building autonomous vehicles that are completely autarkic from their surroundings and towards including communication into their cars. Recent surveys cover benefits of such communication for autonomous driving [247] [26]. Vehicle connectivity is a new technology that enables Vehicle-to-Infrastructure (V2I), Vehicle-to-Vehicle (V2V) and Vehicle-to-Everything (V2X) communication [44] [150]. Vehicles equipped with interactive Advanced Driver Assistance Systems (ADAS) and Cooperative Intelligent Transportation Systems are typically considered to be connected [232]. Car connectivity may give both the regular vehicle and the autonomous vehicle new information and services. MVS are capable of receiving information from the V2X outside the field of vision and negotiating with other road users. Thus, MVS may outperform the *individual* intelligent vehicle not only in terms of safety and performance, but also in terms of traffic throughput and fuel efficiency via global route planning and cooperative driving [86].

The connected ecosystem provided by the V2X paradigm linked to the autonomous vehicle technology permits to unlock new innovative mobility solutions just to mention a few: car sharing [102], where fleets of autonomous vehicles will be available to serve our urban mobility needs; personalized public transport and ride sharing [43][215], where autonomous vehicles and buses can dynamically gather people based on their actual required routes; smart and more effective parking approaches [142], in that autonomous vehicles can search for parking slots based on criterion different from the very soon and very close, one that humans usually adopt, and exploiting additional information that they might have.

It is difficult to test these innovative mobility solutions on public roads, and

there is a lack of experimental validation in real-world conditions [105]. Therefore, several programs based on MVS have been established by governments and national organizations. The following section gives an overview of MVS projects around the world.

1.2.4 Overview of MVS projects around the world

In this section it is proposed to delve into the projects related to the MVS. These projects epitomize the technological advancements and innovations dedicated to enhancing the capabilities of MVS when navigating in on-road dynamic environments such as highways.

1.2.4.1 PATH

Partners for Advanced Transportation Technology (PATH) [164] (cf. Figure 1.5) is a research and development program of the University of California, Berkeley, has been a leader in Intelligent Transportation Systems research since its founding in 1986. In collaboration with the California Department of Transportation (Caltrans), administered by the university's Institute of Transportation Studies, PATH is a multi-disciplinary program with staff, faculty, and students from universities worldwide and cooperative projects with private industry, state and local agencies, and nonprofit institutions.

The PATH research on automated platoons was initially motivated by the need to produce a significant increase in the capacity of a highway lane, so that increases in travel demand could be accommodated with a minimum of new infrastructure construction. The PATH studies of highway capacity showed that it could be possible to increase lane capacity by factor of two to three over today's capacity if the vehicles were driven in platoons of up to ten cars [161]. The gaps between platoons would be long enough to ensure that even in the worst crash hazard condition, with maximum deceleration, a following platoon would be able to stop without hitting the last vehicle of the forward platoon. An extensive modeling and simulation study of crash safety and capacity showed the advantages of the platoon mode over individual automated vehicle [49]. The PATH studies have been based on the assumption that all vehicles would be automated, including the first vehicle on the platoon.

PATH first tested the longitudinal control of a four cars platoon at highway speeds in 1994, and then a demo based on eight cars platoon was developed for the National Automated Highway System Consortium (NAHSC) in 1997. The latter included a large set of maneuvers; car following, lane change, joining and leaving platoon to cite a few, all under completely automatic control. More recently, the PATH platooning research has focused on heavy trucks, mainly because of the potential of energy saving associated with aerodynamic drag reductions. Operating tractor-trailer trucks in close-formation automated platoons of three trucks could enable a capacity of about 1500 trucks per lane per hour [39]. The PATH experiments on truck platoons have shown the technical feasibility of driving two trucks at a gap of 3m and three trucks at a gap of 4m between trucks. These experiments have also shown direct fuel consumption savings in the range of 5% for the lead truck and 10% to 15% for the following trucks.



Figure 1.5. Highway cooperative navigation part of the PATH project [70]

1.2.4.2 Demo 2000

A demonstration on the cooperative driving with five automated vehicles was conducted in November 2000 part of the Demo 2000 cooperative driving project to show the feasibility and potential of the technologies [225]. The cooperative driving in Figure 1.6, aiming at the increase in the safety and throughput of road traffic, here means that automated vehicles are driven in a flexible platoon with a short inter-vehicle distance over a couple of lanes. The vehicles of the demonstration are equipped with RTK-GPS [229] for vehicle localization, a laser radar detection of an obstacle or a preceding vehicle, the inter-vehicle communication unit, and the onboard display for the passenger to locate the neighboring vehicles. The localization data are employed for the lateral and the longitudinal control of the automated vehicle with the precise map of the demonstration site. The data on the localization and speed of each vehicle in the platoon and the location of the obstacles are transmitted at every 100 milliseconds over the inter-vehicle communication. The protocol, nicknamed as DOLPHIN (Dedicated Omni-purpose inter-vehicle communication Linkage Protocol for Highway automatIoN) after the ecology of the dolphins, is designed for the dedicated use in the demonstration. The five automated vehicles drove at 40 - 60 km/h on a test track according to the demonstration scenario including stop and go, platooning, lane changing, merging, and obstacle detection and avoidance [230].



Figure 1.6. Platoon navigation under the DEMO 2000 project [93]

1.2.4.3 Scania platooning

Fuel is one of the largest cost for a fleet owner [39] [4]. Therefore, systems that reduce fuel consumption are high financial interest for fleet owners. Reducing the fuel consumption also reduces the environmental effect of the transport. Scania's main interest in the platooning is hence focused on heavy-duty goods vehicle platooning on highways (cf. Figure 1.7) with a highlight on fuel consumption minimization [15].

Distributed control of a heavy-duty vehicle platoon is a collaboration between Scania and the Royal Institute of Technology, and is partly funded by the Swedish government. The main focus of the project is how a single vehicle operating in a platoon should be efficiently controlled without jeopardizing safety. Longitudinal movement is autonomously controlled while lateral movement is manual. The control architecture has been developed based on a distributed control, meaning that each vehicle is responsible for its own control, based on information from onboard sensors (e.g., radar, cameras) along with, information exchange between the vehicles in the platoon via V2V communication.

It also includes real road tests of platooning. The trial takes place between two Swedish cities and it aimed to:

• Investigate the fuel saving potential in real traffic conditions,

- Evaluate the driver acceptance,
- Analyze interactions between the platoon and the surrounding traffic.



Figure 1.7. Scania platooning [image courtesy of Scania]

1.2.4.4 Energy ITS

A national Intelligent Transportation System (ITS) project by the Japanese Ministry of Economy, Trade and Industry, started in 2008, aims at energy saving and global warming prevention with ITS technologies [226][177]. The latter has two themes: an automated truck platoon and an evaluation method of effectiveness of ITS on energy saving. Another motivation for the project is mitigating the lack of skilled drivers. On a test vehicle, a platoon of three completely automated heavy trucks and one fully automated light truck traveled at 80 kilometers per hour with a distance of up to 10 meters [227]. The lateral control is based on the lane marker detection by the computer vision, and the longitudinal control is based on the gap measurement by the lidar, radar and the V2V communication (cf. Figure 1.8). According to the trial, platooning of 10m gap at 80 km/h can reduce energy by about 15% (measurement) by the aerodynamic drag reduction, and CO_2 by 2.1% (simulation) [228]. Additionally, this research included testing on four heavy-duty trucks equipped with Cooperative Adaptive Cruise Control (CACC) [227].

1.2.4.5 SARTRE

The SARTRE program (SAfe Road TRains for the Environment) [38][55] is an European Commission Co-Funded project that seeks to support a change in transport utilization. The project vision is to develop and integrate solutions that allow vehicles to drive in platoons on public roads without modification of the



Figure 1.8. Configuration of the automated truck part of the Energy ITS project [227]

infrastructure. SARTRE train defines a platoon as a collection of vehicles led by manually driven heavy-duty vehicles. The vehicles behind follow the lead vehicle automatically; both laterally and longitudinally (cf. Figure 1.9). Vehicles may join or leave the platoon dynamically (e.g., leave on arrival at the destination). SARTRE expected advantages of platooning include a reduction in fuel consumption, increased safety, traffic efficiency, increased driver convenience and comfort [56][81].



Figure 1.9. SARTRE project demonstration [image courtesy of the Volvo car corporation]

1.2.4.6 Grand Cooperative Driving Challenge

Low cost and reliable communication systems have recently renewed the interest in MVS. In the 2011 Grand Cooperative Driving Challenge (GCDC) (cf. Figure 1.10) [38][184][155][107], nine international teams joined the challenge with the purpose of supporting and accelerating the introduction of MVS in everyday traffic through a driving challenge. Two scenarios, one urban and one highway, were performed. The urban scenario featured two platoons standing one after the other with a certain distance. The task was to let the rear platoon join the foremost platoon with minimum disturbances. The highway scenario focused on the platooning performance, where a lead vehicle from the organizers guided the participating vehicles in maneuvers.

For both scenarios, evaluation criteria were based on total platoon length, platoon length variations, vehicle gap length, and string stability. The vehicles incorporated automated longitudinal control, whereas the drivers manually operated the lateral steering. The 2011 GCDC focused on forming platoons, while basic platoon operations such joining from rear on a single lane were performed, and complicated operations such as platoon merging from two parallel lanes were out of reach. The winner of the 2011 GCDC winner was Team AnnieWay; a group of researchers hosted at Karlsruhe Institute of Technology (Karlsruhe, Germany) [98].

Many challenges surfaced during the 2011 GCDC, ranging from different interpretation of standards (e.g., time, position) to issues with radar reflections. One of the main challenges that occurred was related to wireless communication, where all participants should intercept the transmitted information in the same way.

While 2011 GCDC focused on basic platooning such as forming and maintaining a platoon, the 2016 GCDC [183] aimed to demonstrate how cooperative automated vehicles can perform complicated platooning operations with close to reality traffic scenarios (cf. Figure 1.11). In addition to the Heudiasyc team, nine other teams were part of the 2nd GCDC edition and were challenged through two typical environments; highway and urban intersection [85].

Advanced platoon operation was investigated in the 2016 GCDC. The cooperative platoon merge scenario involved two platoons driving on a highway on different lanes that must merge into one platoon. A competition zone was defined within which the vehicles' operations were judged. Highway scenario included common events on road traffic (e.g., road works, lane closure, traffic merge).

Intersections are one of the most critical and challenging traffic environments that are attracting significant works [61]. In order to showcase MVS abilities, a cooperative urban non-controlled intersection was considered. To be more realistic, the latter involved non-communicating vehicles. In addition, a third scenario was considered, including an emergency vehicle that required passage in a congested traffic situation. Team Halmstad from the Halmstad University (Halmstad, Sweden) won the challenge [19].

In Table 1.1 a summary of the above discussed project is given.



Figure 1.10. GCDC 2011 [image courtesy of IEEE]



Figure 1.11. GCDC 2016 [image courtesy of IEEE]

	37.1.1.4	$\alpha + 1$	TC	m m : , , ;	G	<u> </u>
	Venicle type	Control	Infrastructure requirement	Traffic integration	Sensors	Goals
PATH (1986 - 1997)	Cars or heady	Longitudinal	Reference markers	Dedicated lane	Mixed	- Energy savings
	duty trucks	and lateral	in road surface	(highway focused)		- Increase traffic throughput
		Longitudinal				- Demonstrate platooning efficiency
Demo 2000 (2000)	Cars	and lateral	None	Mixed	Mixed	- Build a dolphin inspired
		and lateral				communication protocol
Seenia platooning				Mirrod		- Fuel savings
(2001- now)	heavy-duty trucks	Longitudinal	None	(highway focused)	Mixed	- Demonstrate complex
						maneuvers using platooning
Energy ITS	honry duty trucks	Longitudinal	Lano markors	Dedicated lane	Mixed	- Mitigate lack of skilled drivers
(2008 - 2012)	neavy-duty trucks	and lateral	Lane markers	Dedicated lane	MIXCU	- Energy saving
						- Increase safety and passengers'
SARTRE	Cars and heavy	Longitudinal	Nepe	Mixed	Production	comfort
(2009 - 2012)	duty trucks	and lateral	none	(highway focused)	sensors	- Reduce road congestion,
						along with energy saving
1st GCDC (2011)	Cars mainly	Longitudinal	None	Mixed	Mixed	- Accelerate MVS deployment
				(highway focused)		- Demonstrate platooning feasibility
						- Outperform complex platooning
2nd GCDC (2016)	Cars	Longitudinal and lateral	None	Mixed	Mixed	maneuvers
						- Demonstrate cooperative
						intersection crossing feasibility

 Table 1.1. Summary of the major project involving MVS

1.3 Cooperative navigation for MVS: scenarios overview

The challenge of **coordinating** the actions and interactions of MVS within onroad environments in a vast multifaceted one. This challenge encompasses an array of highly diverse scenarios, ranging from scenarios involving cooperative merging on highway on ramps, as well as, employing tools such as Cooperative Adaptive Cruise Control (CACC) for tasks like platooning. To urban settings involving cooperative navigation through signalized and non-signalized intersections. Despite this apparent diversity, these varied scenarios can be effectively unified and conceptualized within a coherent framework (cf. Figure 1.12).

With **coordination** we refer to the decision-making and planning (involving vehicles themselves and possibly some additional infrastructure process) aimed at orchestrating vehicle' actions as to achieve a goal which cannot be achieved (or not optimally) by each vehicle in isolation. The coordination goal may belong either to an individual vehicle (e.g., intersection crossing), or to a group of vehicles (e.g., platoon navigation), or even the traffic system as a whole (e.g., improving the city traffic flow) (cf. Figure 1.12).

In the following section, it is proposed an overview of the topics related to the problem of coordinating the motion of MVS from the scenario perspective.



Figure 1.12. MVS applications: an overview

1.3.1 Cooperative navigation in urban environment

In urban core areas, MVS navigation becomes highly difficult under a variety of urban situations involving numerous road users. Several MVS applications have been explored to benefit safety, mobility and the environment with connected vehicles [126][97][71]. Among them, MVS cooperation technology for intersection

or roundabout applications attract representative and significant efforts. More precisely, research on MVS cooperative driving is conducted at both signalized and unsignalized intersection/roundabout.

Intersection crossing (cf. Figure 1.13) and roundabout (cf. Figure 1.14) navigation in urban areas is the problem of coordinating vehicles' motions while concurrently navigating through the intersection/roundabout. According to [152], the problem of intersection/roundabout is a competitive resource-oriented problem, where the vehicles are self-interested agents willing to obtain the right-ofway as soon as possible across the shared resource represented by the intersection/roundabout.



Figure 1.13. Cooperative intersection crossing

Today, intersections management is realized either by a central controller, the traffic light, or by imposing to vehicles (i.e., the drivers) pre-defined coordination rules to be obeyed (e.g., stop at sign or give the right-of-way to vehicles coming from the right), as well as for roundabout [195]. This put the responsibility for safety fully in charge of humans, and not promote efficiency.

To tackle the problem of intersection/roundabout navigation, it is possible (i) to act on the supply side, increasing the number of roads or lanes in a network, (ii) to reduce the demand, restricting the access to urban areas at specific hours or to specific vehicles, or (iii) to improve the efficiency of the existing network, by means of a widespread use of MVS.

In fact, with the help of V2V communication and/or V2I communication, it is possible to conceive a variety of innovative solutions, safer and more efficient. Roughly, V2I communication is often preferred at signalized intersections in order to get Signal Phase and Time (SPaT) information and prevent unnecessary speed changes or complete stops [255].



Figure 1.14. Cooperative roundabout crossing

Within an intelligent transportation system, V2V and V2I communications are often used at intersection/roundabout [105][248], eventually making traffic light and stop signs obsolete. With the help of V2V and/or V2I communication, the possibility of sharing future intentions among vehicles is feasible, helping to prevent conflicting trajectories. Thus, following the planning and scheduling algorithm, MVS can be assigned specific sequences to cross the intersection/roundabout. Alongside, communication offers opportunity to optimize the passing order, also known as passing sequence, among the vehicles within a certain communication range, this means that the navigation task through the intersection/roundabout can be seen as a global task related to the participating vehicles, where each vehicle has its own passing order provided by a multicriterion function. This cooperative behavior permits to translate the problem of intersection/roundabout from a competitive resource-oriented task to a cooperative altruistic-task oriented, where the global aim tackles the traffic throughput, idling time at intersection/roundabout and energy efficiency, in addition to risk metrics.

In fact, the research work related to MVS navigation coupled with V2V/V2I communication in intersection/roundabout has gained considerable attention during the last decades. Comprehensive surveys about this problem are analyzed in [61][83][106][171][189].

1.3.2 Cooperative navigation in highway environment

Among the maneuvers performed by the vehicle on a highway environment one can cite on/off-ramp merging [195][270], platoon navigation [128], lane change [166]. Some of these maneuvers can be challenging since they involve numerous road users performing at high maneuvers [72].

With the significant developments achieved during the last decade with the help of intelligent transportation systems, addressing mobility issues on highway become possible. The integration of MVS on highway environment appears to be the next ITS move and the natural extension to the aforementioned mobility technology. Mainly because, in one hand, under the MVS paradigm, collaborative navigation on a highway addresses the problem of coordinating the vehicle's maneuvers, so they travel together without causing any collision risk [34][88][165]. In the other hand, connectivity integrated in MVS, make them capable to not only drive by themselves with on-board sensors, but also to communicate with each other with the help of the V2V communication [110][32]. Consequently, it improves road safety where future actions can be well anticipated where a better understanding of the navigation scene can be built. Beside the safety benefits, MVS technology associated with communication permits to increase traffic throughput by reducing safety distance between MVS, and to improve energy efficiency by avoiding unnecessary speed changes [30].

Over the past decade, the impact of MVS technology on highways has captured significant attention. This innovative technology is regarded in the literature as promising, not only for addressing traffic congestion on highways but also for enhancing safety and mobility by addressing the challenges associated with on/off ramp merging. An overview addressing the utilization of the MVS technology in the context of highway navigation in platoon is provided in Section 1.3.2.1. Section 1.3.2.2 specifies MVS utilization in the context of on-ramp merging on highway.

1.3.2.1 Cooperative platooning navigation on highway

On the highway, human driven vehicles have a natural tendency to follow one another closely and operate within a phenomenon known as longitudinal formation or platooning (cf. Figure 1.15). For a human-operated vehicle, a platoon-based formation can be considered as basic behavior, and it can be easily distinguished in everyday life. However, for MVS, they must stay in the same lane and follow nearby vehicles by maintaining safe distance and velocity, to form the desired longitudinal shape. The aim of the platoon formation control is to confirm that all the vehicles in a platoon move at the same speed while maintaining the desired geometry, which is stated by a desired inter-vehicle spacing strategy [115]. Therefore, for MVS, platoon formation requires specific algorithms, controllers and strategies.



Figure 1.15. Cooperative merging on highway

It was under the PATH (cf. Section 1.2.4.1) research project that the current CACC implementation in production vehicles was mainly developed as the extension of the commercially available ACC system [163] (cf. Figure 2.20). Therefore, most CACC vehicles are also equipped with sensors installed on ACC vehicles, such as odometers, radar and/or LiDAR, along with communication. Given the fact that the communication bandwidth might become insufficient when the number of MVS increases in a CACC system, short ranged wireless technologies are more accepted for V2V communication [103]. By sharing vehicle information such as acceleration, speed and position, MVS in the communication range can cooperate to increase highway capacity [192].



Figure 1.16. Cooperative adaptive cruise control: platooning navigation

In CACC systems, MVS share their own data with other MVS in the network by V2V communications, which is realized in a distributed manner without central management [252].

CACC takes advantages of V2V communications to allow MVS to form platoons and be driven at harmonized speeds with short-time headway between them. MVS in a certain communication range can cooperate with others to obtain the following benefits: (i) driving safety is increased since actuation time is shortened compared to manually driven, and downstream traffic can be broadcasted to following vehicles in advance; (ii) roadway capacity is increased due to the reduction of time/distance headway between vehicles; (iii) energy consumption and pollutant emissions are reduced due to the reduction of unnecessary velocity changes and aerodynamic drag on following vehicles [157].

1.3.2.2 Cooperative on-ramp merging on highway

Traffic merging at highway on-ramps (cf. Figure 1.17) is a major conflict that generates safety and mobility concerns. The difficulty arises for the AV along the on-ramp where it has to discern whether to accelerate or decelerate to enter the main line safely and may not have a clear line of sight. Meanwhile, the mainline highway users may have to modify their speeds to permit the entrance of merging AV, thus affecting traffic flow which may result in road congestion. To address these issues, cooperative MVS have was studied and applied to the highway on-ramp merge case, where different control algorithms have been proposed and implemented to allow MVS to merge with each other in a cooperative manner. Existing related works have been reviewed in [195].



Figure 1.17. Cooperative on-ramp merging on highway

1.4 Conclusion

In this chapter, a comparison between single autonomous vehicle systems and multi-vehicle systems is explored. Since MVS comprises several individual AVs in a cooperative manner, this chapter provides a summary of the mechanisms employed by the AVs within their control architecture, paving the pay for the presentation of the MVS paradigm. An extensive overview of prominent projects involving MVS is presented. Additionally, the chapter delves into contexts where MVS finds utility in dynamic and challenging on-road environments. It examines relevant works related to both urban cooperative navigation and highway platooning navigation, in addition to the on-ramp merging on highway. Finally, the chapter highlights the significant advantages of utilizing MVS in various scenarios, with a particular focus emphasis on highway context.

CHAPTER 2

Cooperative Motion Planning and Decision-Making for MVS: Architecture Overview

Abstract

This chapter is devoted to providing a comprehensive overview of MVS control architecture. It delves into an exploration of pertinent literature, examining the various applications of MVS across different domains. The chapter sheds light on the paradigms employed in constructing MVS control architecture. Furthermore, it offers a detailed literature review of MVS applications within complex and dynamic environments.

Contents

2.1	Introd	uction
2.2	Contro	ol architecture for MVS
	2.2.1	Perception module
	2.2.2	Localization module
	2.2.3	Communication module
	2.2.4	Decision-making module
	2.2.5	Action module
2.3	Overv	iew of MVS control architecture
	2.3.1	Centralized approaches 44

	2.3.2 Decentralized approaches
	2.3.3 Hybrid approaches
2.4	MVS navigation in formation: a review 49
	2.4.1 Leader-Follower
	2.4.2 Virtual Structure
	2.4.3 Behavior-based
	2.4.4 Consensus-based control $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 54$
	2.4.5 Discussion
2.5	MVS Navigation in complex environments
	2.5.1 Highway collaborative navigation
	2.5.1.1 Cooperative Adaptive Cruise Control (CACC) 60
	$2.5.1.1.1 \text{Optimal Control} \dots \dots \dots \dots 61$
	$2.5.1.1.2 \text{String stability} \dots \dots \dots \dots \dots \dots \dots \dots \dots $
	2.5.1.1.3 Model Predictive Control (MPC): 63
	2.5.1.2 Evaluation of the highway collaborative naviga-
	tion approaches $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 64$
	$2.5.1.2.1 \text{Traffic throughput} \dots \dots \dots 64$
	$2.5.1.2.2 \text{Energy efficiency} \dots \dots \dots \dots 65$
	$2.5.1.2.3 \qquad \text{Negotiation} \dots \dots \dots \dots \dots \dots 66$
	2.5.2 Collaborative on-ramp merging on highway 67
	$2.5.2.1 \text{Traffic throughput} \dots \dots \dots \dots \dots \dots \dots 68$
	$2.5.2.2 \text{Energy efficiency} \dots \dots \dots \dots \dots \dots \dots 70$
	$2.5.2.3 \text{Negotiation} \dots \dots \dots \dots \dots \dots \dots \dots \dots $
2.6	Conclusion

2.1 Introduction

In the recent years, the automotive research and industry has undergone a profound transformation, driven by rapid advancements in technology and a growing emphasis on safety, sustainability and efficiency. At the forefront of this evolution, one concept is pointed to revolutionize the way we think about transportation: Multi-Vehicle System (MVS). These cutting edge technologies represent the fusion of several years of development of the automated vehicles, connectivity and automotive engineering, promising to reshape not only the way we move and travel but also our urban environment and our roads.

The primary objective of this chapter is to provide an extensive overview of the MVS control architecture, along with the examination of the relevant literature concerning the utilization of MVS across diverse applications. Section 2.2 outlines the various modules and process integrated in the MVS control architecture. In Section 2.3, an insight into the paradigms employed in constructing the MVS control architecture is presented. Formation control stands as an important aspect of the MVS paradigm, and thus, in Section 2.4, a comprehensive review of various formation control approaches is conducted. Lastly, Section 2.5 provides a literature review highlighting MVS applications within complex environments.

2.2 Control architecture for MVS

Automated Vehicles (AVs), known as self-driving cars or autonomous vehicles, have captured the imagination of the public, scientists and industry alike during three decades. These vehicles leverage sophisticated sensors, complex decisionmaking processes and real-time data analysis to navigate the roads with minimal to absence of human intervention. The promise of improved safety, enhanced energy management, reduced congestion, and increased accessibility has fueled intense research and development. One consensus started to emerge between the contributors to the AVs developments; to take full advantages of the AV technology, AVs must collaborate with the road users.

On the other hand, MVS takes the concept of driving automation to the next level by introducing a collaboration level between vehicles, infrastructures, and even pedestrians. This inter-connectivity has the potential to induce numerous benefits such as enhancing safety, more efficient energy consumption and realtime traffic optimization (cf. Section 1.3).

The autonomous management of a fundamental MVS essentially encompasses five core process as depicted in Figure 2.1, which are detailed as follows:

2.2.1 Perception module

The perception module plays a pivotal role in gathering precise data for both the vehicles part of the MVS and the surrounding vehicles. It relies on a combination of sensors designed for distinct purposes:

- Proprioceptive sensors: These sensors focus on capturing information related to the current state of the individual vehicle within the MVS. This includes data such as position, orientation, velocity, battery level, etc. Common examples of proprioceptive sensors encompass odometers, gyroscopes, and Inertial Measurement Units (IMUs).
- Exteroceptive sensors: Exteroceptive sensors, on the other hand, are designed to collect information from the external environment. They include sensors like cameras and range sensors as LIDAR, sonar and infrared devices (as illustrated in Figure 2.4).



Figure 2.1. Reference control scheme for MVS, adapted from [238]

One of the critical tasks of the perception module involves the detection of various obstacles, static or dynamic, within the MVS environment. This task holds paramount importance, as both localization and decision-making modules (refer to Section 2.2.2 and Section 2.2.4) rely heavily on its outputs. Through the utilization of data fusion algorithms, the vehicle's cooperative perception layer effectively combines its own proprioceptive and exteroceptive data with the perception data shared by the vehicles/infrastructure through the communication module (cf. Section 2.2.3). The objective is to deliver a dependable estimation of obstacle positions and a coherent representation of the surrounding environment.

Common approaches to cooperative perception predominantly employ range sensors and cameras [190] [136].

2.2.2 Localization module

The localization module is responsible for establishing the precise poses of the MVS w.r.t. its navigation environment. This is a pivotal process since it serves as a foundation upon which many other processes rely.

Typically, localization requires data fusion from the aforementioned sensors, combined with data shared via communication module. This fusion is crucial in achieving a precise estimation of the poses of the MVS vehicles. Various research endeavors have explored data fusion techniques like Extended Kalman filter (EKF) [114], Markov localization [31], particle filters [191], and robust filters designed to handle uncertainty [76]. It is worth noting that this thesis primarily concentrates on developments within the decision-making and control levels part of the MVS architecture, assuming the presence of an accurate localization system to facilitate all simulation results.

2.2.3 Communication module

The MVS paradigm has witnessed extensive research efforts from scientific institutions, industry players, and various organizations aiming to address communication capabilities [108]. These capabilities encompass various forms of communication, including Vehicle-to-Vehicle (V2V), Vehicle-to-Infrastructure (V2I), Vehicle-to-Pedestrian (V2P), and Vehicle-to-Network (V2N) communication, collectively known as Vehicle-to-Everything (V2X) communication. V2X communication facilitates diverse use cases by enabling the exchange of messages among infrastructure elements, vehicles, and pedestrians, employing various wireless communication technologies such as Dedicated Short-Range Communication (DSRC) [130] [242] and cellular network technologies like 5G [169].

V2X communication holds the promise of substantially enhancing road safety by enabling anticipatory actions through a more comprehensive understanding of the navigation environment, augmenting traffic throughput by reducing intervehicle safety distances within the same formation, and improving fuel efficiency by minimizing unnecessary maneuvers.

The information flow topology defines the way a vehicle obtains information from other vehicles part of the MVS. Some typical types of information flow typologies include (a) predecessor-following (PF), (b) predecessor-leader following (PLF), (c) bidirectional (BD), and (d) bidirectional-leader follower (BDL), which are illustrated in Figure 2.2 [266][249]. It is important to note that the majority of scenarios involving MVS are inherently dynamic, introducing complexities such as interactions between multiple MVS or potential communication disruptions within existing typologies. Consequently, the information flow within the MVS is dynamic, meaning that communication patterns car vary as MVS navigate through different typologies during their operations.

It is worth noting that main aim of this thesis is on the decision-making and the control level. Therefore, the communication module is said to be reliable (i.e., communication losses and delays were not considered). This assumption helps to focus on the considered levels part of the MVS control architecture and to facilitate thus the different analysis.



Figure 2.2. Communication topologies, adapted from [192]

2.2.4 Decision-making module

The decision-making module derives its outcomes from a myriad of elements, including the assigned task, information regarding other vehicles within the MVS, acquired through the communication module (as discussed in Section 2.2.3), the environment's representation from the perception module (cf. Section 2.2.1), and the vehicle's position obtained from the localization module (cf. Section 2.2.2).

To illustrate, when a vehicle is navigating towards its designated goal and detects an obstacle in its path, the decision-making process comes into play. It must decide whether the vehicle should brake and come to a halt or maneuver to evade the obstacle, based on the vehicle's specific task and the overarching MVS objectives.

Various architectural frameworks have been devised to address these decisionmaking elements. One such framework is the LAAS architecture (An Architecture for Autonomy) [16], organized into three hierarchical levels: the decision level, the execution level, and the functional level. The decision level oversees processes requiring anticipation and a degree of global knowledge within the context of task execution. It encompasses planning and decision-making capacities. The execution level translates tasks into procedures composed of elementary robot actions and supervises their execution while reacting to the environment. The functional level comprises elementary robot functions implementing control loops. Another architectural approach, especially suited for high-complexity navigation tasks, is the multi-controller (or behaviors) based decision architecture [11]. This approach is founded on the concept that a robot can accomplish a complex global task through the coordination of several elementary behaviors/controllers, such as trajectory tracking, obstacle avoidance, and MVS coordination (e.g., splitting and joining), among others, to better govern the overall vehicle behavior [11]. An example showcasing the use of this architecture approach is the hybrid multicontroller architecture proposed in [238] (cf. Figure 2.3) for a group of indoor robots. This architecture subdivides the overarching MVS navigation task into a set of accurate and dependable elementary controllers, such as obstacle avoidance, trajectory tracking, target reaching, and navigation in formation.



Figure 2.3. Control architecture for the navigation in formation of MVS [238]

2.2.5 Action module

In the action module, the commands generated by the decision-making module (refer to Section 2.2.4) are applied to the vehicle. A control law formulates these commands, instructing the actuators (motors) based on sensor information, the decision-making module (cf. Section 2.2.4), and the vehicle's model. This control law's complexity can be tailored to align the system's modeling intricacies and the specific task at hand.

For example, in navigation applications, the design of the control law often focuses on guiding the vehicle along a predetermined trajectory, as demonstrated in [239][58]. Additionally, the control law can account uncertainties associated with the sensors and actuators, ensuring robust and dependable system operation [258].

Discussion: MVS' technology future challenges

Despite decades of research and development dedicated to MVS systems, there remains a significant untapped potential for technology to enhance the experience of road users. Nevertheless, MVS continue to grapple with a multitude of challenges, as outlined below [86][105][151][238]:

Accurate perception/localization data: Perception errors may be exacerbated in MVS. A data association mechanism that is efficient for a variety of vehicles and sensor designs is required [136]. The performance of the cooperative perception is highly dependent on the accuracy of relative localization. However, relative localization accuracy may be limited in particular circumstances [147].

Real-time coordination: In scenarios featuring dynamic environments with multiple agents, ensuring system safety often hinges on cooperative motion planning for new maneuvers. This kind of planning relies heavily on the algorithms integrated on board and is particularly sensitive to the presence of nearby traffic. Consequently, achieving real-time coordination becomes imperative for the onboard controller [75][186].

Communication: There is a notable absence of unified communication topologies and protocols tailored for the development of MVS [109]. To address this challenge, a dependable decision/control strategy is needed to effectively manage various communication issues such as delays, packet losses, error messages, and interactions with uncooperative agents [66][217]. Furthermore, the efficiency of cooperative perception can be adversely affected by transmission latency and bandwidth limitations. The trade-off between communication bandwidth and maintaining optimal real-time closed-loop performance of the MVS has been a relatively under-explored aspect in this context [14][198][69].

Accurate forecasts: The lack of a unified communication protocol and a normalized decision/control architecture for MVS that facilitate access to more accurate predictions has led the majority of current methods to depend solely on static route information, like road curvature and speed limits. However, the potential of MVS often hinges on advancements in data forecasting, which is linked with central technological challenges.

Real-world verification: The large majority of MVS related projects (e.g., CACC) were tested in a controlled environments (i.e., mainly due to local laws, on-road scenarios complexity, drivers acceptance, etc.)[21][68][80]. Thus, the MVS

efficiency in dynamic and open on-road environments is difficult to know. Current verification issues for MVS' advanced algorithms include real-world testing and application to non-highway scenarios.

Certainly, MVS have the potential to eliminate or greatly reduce unforeseen human errors by leveraging perception and localization sensors, as well as communication systems to enhance their awareness of the environment. These technologies offer substantial advantages in various scenarios, as discussed in the Section 1.3 pertaining to CAV applications. However, realizing the full potential of MVS on public roads necessitates further exploration and utilization of these technologies. The decision and control mechanisms governing CAV systems should not solely rely on individual vehicle perceptions or objectives but should also consider the overall system state. A central challenge in the field of MVS navigation is the formulation of a robust control strategy. This strategy is a cornerstone in ensuring that all vehicles within the CAV system can seamlessly configure themselves in a synchronized and efficient manner to achieve their designated objectives. Additionally, by incorporating V2X communication, MVS can engage in the exchange of crucial data, as perception and localization information, as well as real-time status updates like intended acceleration, actual acceleration, or velocity, with fellow road users [105]. An overview of MVS-related decision/control architecture will be discussed in the next section.

2.3 Overview of MVS control architecture

As depicted in Figure 2.4, the MVS comprises multiple modules deployed within each AV part of the global ecosystem. Specifically, in this context, we are focusing on the MVS control architecture, encompassing system responsible for communication, perception, localization, decision-making/planning and control modules (cf. Figure 2.4). The literature contains a plethora of descriptions regarding control architectures for MVS [12][113][146][240][24][23][25][248]. It is worth noting that this PhD thesis does not aim to provide an exhaustive classification of MVS structures and architectures. Nevertheless, a pivotal consideration that must be addressed before embarking on the development of the MVS decision/control architecture is the choice between a centralized and decentralized management scheme within the architectural framework [195] [194][11]. This section is dedicated to evaluating the advantages and disadvantages of centralized vs. decentralized management within the MVS context.



Figure 2.4. General cooperative architecture composed of: communication, perception, localization, planning and control modules [248]

2.3.1 Centralized approaches

An architectural configuration earns the designation of *centralized*, when a portion or the entirety of the sensory and/or decision-making processes of individual robotic entities is detached from their physical structures and overseen by a central control unit [74][268], commonly referred as a supervisor, or central planner [11]. A centralized architecture is often synonymous of a Top-Down approach, which entails envisioning a conductor (the supervisor) orchestrating a group of mobile robots, akin to orchestra.

The principal advantage of this architectural approach lies in the fact that a central unit, whether termed a controller or supervisor, can base its decisions on a comprehensive understanding of the entire system, which often surpasses the decision-making capacity of an individual component within the MVS [135]. This paradigm is usually adopted in small-scale systems, where the required computational capacity is moderate. Nevertheless, it is important to note that these architectures necessitate a complete awareness of every element within the system, demanding substantial computational power and a significant flow of information through extensive communication [135][261]. Additionally, they lack for robustness (due notably to its dependency on a single controller/ supervisor).

Regarding the implementation, centralized architectures are designed to have the capacity to manage data originating from communication modules, perception/localization modules, and motion modules. This with the help of information flow topology [265], sensor fusion approaches [104], and cooperative planning [192]. In scenarios where shared topological spaces are involved, such as intersection crossing, centralized approaches often emerge as viable solution when a limited number of intersecting vehicles is involved, in order to tackle the collision problem. In [77], the authors propose a reservation scheme as a means to oversee the control of a single intersection where two roads intersect. Thus, each vehicle is treated as a driver agent responsible for submitting a request to reserve specific space-time cells, allowing them to cross the intersection within a particular time interval. Upon receiving these reservation requests, the centralized reservation system takes decisions; accepting the request if there are no conflicts with previous accepted reservations, otherwise, the request is declined. With the development of the communication technologies such as 5G, centralized architectures gain a major popularity. However, its longer end-to-end latency prohibits its use in vehicle safety [150].

An example of centralized architecture is proposed in [238][47]. This architecture named RoboSketon (cf. Figure 2.5) is based on three-layer architecture and it contains an agent that manages the other agents (cf. Figure 2.5). If the controlling agent (coachAgent) fails, the hole system could readily fail [238].



Figure 2.5. Example of centralized architecture [238][47]

2.3.2 Decentralized approaches

A decentralized architecture is a system in which parts have local authorities by themselves [22]. In contrast to the centralized architecture, in a decentralized architecture, each vehicle within the MVS maintains its own independent processes for perception/localization and decision-making/planning. Such architectures entail a reduction in the volume of communicated signals and data. Notably, in this decentralized setup, each vehicle within the MVS does not require a comprehensive understanding of the overall environment prior to taking actions within its immediate surroundings.

One of the key advantages of this decentralized design is that the AVs within the MVS paradigm can individually control their operations without necessitating external commands, which impacts resilience if the system faces defects and failures. Moreover, decentralized control facilitates parallel computation, enhancing system responsiveness and thereby ensuring the reliability and scalability of the implementation [48][89][192]. The primary drawback of decentralized management lies in its heavy reliance on extensive coordination. This is primarily because the specific tasks allocated to each vehicle are embedded within their local control systems. Consequently, if there is a need to alter the assigned tasks, achieving a global reconfiguration of the MVS without the presence of a supervisor can be a challenging endeavor.

The decentralization paradigm can also be applied to various aspects of the MVS, including communication technologies [5][100], perception techniques involving decentralized fusion [251], and cooperative motion planning [195]. Additionally, in the context of road safety, decentralized systems like dedicated short-range communication have demonstrated their advantages when compared to cellular network like 5G [124][111]. The decentralization paradigm when well mastered is more flexible to deal with MVS having a large number of entities and is generally a synonym of Bottom-Up approach [10].

A typical example is the ALLIANCE/L-ALLIANCE architecture developed in [179][180][181] (cf. Figure 2.6) for heterogeneous robots. Robots have information about their actions and the actions of the group through the top-board communication topology. This architecture uses behavior-based controllers which depend on the assigned robot's task. L-Alliance [180] is an extension of the Alliance architecture [179] which uses reinforcement learning to adjust the activation parameters of the behaviors [238].



Figure 2.6. Example of decentralized architecture [238][181]

2.3.3 Hybrid approaches

Other comprehensive studies with similar discussion suggesting some variations in control approach/architectures might also be discovered in [59][87][122][182][260] [145][267]. It is proposed to highlight several types of notable control architectures as follows:

• Hierarchical architecture: This type of control architecture exhibits a form of local centralization, as discussed in [192][13][22], while maintaining decentralization in a global context. To clarify further, each vehicle within this framework independently manages its assigned responsibilities but provides updates to a central planner part of the vehicle's architecture regarding the status of its activities. Based on this definition, it becomes evident that the vast majority of MVS are structured hierarchically, as elucidated in [101][125]. This hierarchical structuring is particularly well-suited for large-scale designs, where the overarching task is subdivided into a series of smaller, more manageable sub-problems, each addressed within distinct layers [236].

An illustrative example of this architectural structuring is found in the UM-PRS (University of Michigan Procedural Reasoning System) [139][138], where a multi-agent system is employed for environmental recognition.

• Hybrid Centralized/Decentralized architecture: This control framework combines a high-level central controller with localized decentralized controllers within individual vehicles [22][175][209][238]. Consequently, this design combines both of the noteworthy advantages of the centralized architecture (cf. Section 2.3.1) and the decentralized architecture (cf. Section 2.3.2):

- 1. Centralized planner: The centralized planner assumes a pivotal role as a high-level controller overseeing the vehicles within the MVS.
- 2. Robustness: The decentralized controllers within the vehicles part of the MVS exhibit fault tolerance, enhancing the system's robustness.
- 3. Flexibility: The hybrid architecture provides the flexibility to adjust the global task or control strategy based on inputs from both the centralized and decentralized control components.

One compelling illustration of this architecture's application is its use in MVS navigation in formation, with a primary focus on safety and flexibility. The centralized approach allows to the central entity (leader of the formation) to manage the configuration desired formation even for formation reconfiguration. The decentralized approach allows each robot (follower) locally generate its commands to track its assigned target [238].

One example of the centralized/decentralized architecture is proposed in [210] (cf. Figure 2.7). It is based on three-layer architecture, and it allows coordination between the robots (communication layer) and autonomy in action (executive and behavioral layer) [238][210].



Figure 2.7. Example of centralized/decentralized architecture [238][210]
The central aim of this Ph.D. thesis revolves around the creation of a Decision/Control architecture that mitigates in-between the advantages of both the centralized and the decentralized framework (cf. Table 2.1). Designed for cooperative based maneuvers (cf. Section 1.3), the MVS targeted architecture aims to seamlessly align with the individual control architectures of the vehicles part of the MVS, while the Decision/Control architecture for MVS consistently leans toward a bottom-up strategy, where decentralized modules are in charge of performing local tasks. There are specific scenarios or tasks where leveraging global knowledge from a central planner can enhance MVS performance.

	Perception	Localization	Decision	Action	
Centralized	A central en from all the strand the r compute the	tity uses the in deported sense obots' environn e commands to	Each robot executes the command sent by the central entity.		
Decentralized	Each robot uses local information from its sensors to understand its environment and to generate and execute its commands to accomplish the assigned task.				

 Table 2.1. Comparison between MVS architectures [10]

2.4 MVS navigation in formation: a review

Numerous research laboratories, including the current thesis work, have directed their focus toward applications involving navigation in formation. These applications find utility across a spectrum of domains, spanning transportation, agriculture, and military applications [238][254][160].

Formation control, at its core, pertains to the capacity to maintain the relative positions of a group of vehicles/robots with respect to each others or to a specific reference. In essence, this entails ensuring that the group of vehicles or robots adheres to a predefined geometric formation [11].

Navigating MVS in formation presents a set of intricate challenges, as highlighted in [238]. These challenges encompass fundamental questions:

- **Defining the desired formation:** What configuration should the MVS aims to achieve?
- **Determining actual positions:** How do the individual vehicles ascertain their current positions within the formation?

- **Maintaining formation:** What strategies do the vehicles employ to ensure the formation remains intact?
- Adapting to Road Conditions and Obstacles: How can the MVS adapt to varying road topologies and the presence of obstacles?
- **Evaluating formation performance:** What criteria are employed to assess the performance of the MVS formation?

To address these questions, various approaches have been developed, including Leader-Follower, Virtual Structure, Behavior-based, and other Optimizationbased techniques [62][170], each offering insights into the complexities of formation navigation within MVS.

2.4.1 Leader-Follower

The concept of leader-follower formation approach involves a hierarchical structure, comprising a single vehicle designated as the leader (UGV_L) , while the rest, referred as the followers $(UGV_i \text{ and } UGV_j)$, determine their positions relative to the leader pose or another reference pose (cf. Figure 2.8). They achieve this by utilizing their representation of the environment from their own positions [238][162][82].



Figure 2.8. The leader follower formation modeling approach [238]

In the realm of transportation, a prevalent application of this formation strategy is the one-dimensional leader-follower, often referred to as platooning-based formation or column formation. In this context, a leader assumes a pivotal role within the MVS formation, tasked with following a reference trajectory while simultaneously serving as the guiding reference for the followers. The followers, in turn, closely track the leader's or another guide's position and adapt their control inputs accordingly. This leader-follower can be characterized by either constant spacing or constant headway, as it can be seen in Cooperative Adaptive Cruise Control (CACC) (cf. Section 2.5.1.1), among other leader-follower models found in the literature, such as the spring-damper system [65]. The spring-damper model introduces one or more interaction models, enabling control over the intervehicle distances within the MVS formation (cf. Figure 2.9).

One notable advantage of the leader-follower approach is its compatibility with the decentralized architecture, wherein each vehicle autonomously makes decisions based on local information in relation to a common reference, thereby maintaining the formation's shape. Control techniques rooted in optimal control or model predictive control can be leveraged to uphold the formation shape. In [140], an exhaustive survey of methods employed to ensure the effectiveness of the leader-follower approach is given.

However, this approach has its limitations. It heavily relies on the leader, rendering it susceptible to faults in the leader vehicle. Any issues with the leader can disrupt the entire navigation task or even lead to collisions within the MVS formation. Furthermore, when dealing with large MVS formations, where the leader is responsible of supervising the followers, significant computational costs arise, and communication issues, such as package losses and delays, can become prominent concerns.



Figure 2.9. Leader-follower based on spring damper system [238]

2.4.2 Virtual Structure

A vehicle formation materializes when multiple vehicles collaborate across multiple lanes while adhering to a pre-defined configuration. To govern such formations, the virtual structure has been adopted, effectively dividing the formation control into a high-level virtual structure and a low-level control problem [185].

This approach can be summarized through the following steps [259]:

- 1. **Defining the virtual rigid body:** The dimension, shape, and dynamics of a virtual rigid body are defined based on the number of vehicles and the road geometry (cf. Figure 2.10).
- 2. Vehicle tracking: Each vehicle within the MVS closely tracks its designated virtual agent or target (cf. Figure 2.10).
- 3. Virtual agent motion: The motions of the virtual agent or target are computed to align with the desired maneuver.



Figure 2.10. Virtual structure formation modeling approach [238]

The advantages of the virtual structure approach primarily stem from its simplicity compared to other formation control methods. It also offers flexibility in shaping different types of formation control [57][35][37]. Some research combines the leader-follower approach with a deformable virtual structure to tackle formation reconfiguration [241] (cf. Figure 2.3). Formal safety guaranteed for the entire fleet is provided by employing a reconfiguration matrix that considers inter-vehicle distances to prevent collisions [241] [238].

However, there are drawbacks associated with the virtual structure approach, mainly linked to its centralized nature. The central unit is responsible of assigning virtual agents/nodes and overseeing formation maintenance. Additionally, factors like road shape and vehicle dimensions must be taken into account.

2.4.3 Behavior-based

Formation control using a behavior-based approach involves the task of decomposing the main objective into several fundamental sub-tasks (cf. Figure 2.11), with the ultimate aim of generating the desired emergent behavior [20]. The selection of these sub-tasks is carried through two primary methods:

- **Competitive selection:** In this method, only one behavior is chosen at any given time. The design of such a selection system follows a hierarchical structure, with the desired behavior holding the highest priority compared to others (cf. Figure 2.11). It is worth noting that architectures employing hierarchical selection may encounter challenges in stability analysis during the switching phase [11].
- **Cooperative selection:** In contrast to the competitive selection, the cooperative selection in Figure 2.11 allows for the simultaneous selection of two or more behaviors, which are then fused to create an emergent complex behavior. While this approach offers the potential to achieve sophisticated behaviors [11], it introduces complexities during the transition phase what can be a challenging task [205].

The primary advantage of the behavior-based approach is its adaptability to handle more complex tasks and its capacity to generate emergent behaviors. Some research work has successfully integrated graph theory into behavior-based formation control to address issues associated with time-varying communication [187]. Another proposed formation control architecture based on selection of behaviors, combining both self-centered maneuvers through competitive selection and more altruistic maneuvers with the assistance of cooperative selection is presented in [33].



Figure 2.11. Behavior-based formation modeling approach [238]

2.4.4 Consensus-based control

In the realm of MVS, consensus-based control represents an important and fundamental part of the literature related to formation control [192]. Thus, consensusbased control is part of the MVS control architecture paradigm (cf. Section 1.2.3), especially, decentralized MVS control architecture (cf. Section 2.3.2). The essence of consensus-based control lies in the convergence of individual entities toward a common objective through interactions facilitated by sensing and communication networks (cf. Figure 2.12), as elaborated in Section 2.2.3.

The formation modeling used part of the consensus-based control theory relies on the algebraic graph topology as in Figure 2.12, where the vehicles are represented by the nodes and in-between node links are the vehicle's in-between communication. The primary objective of consensus-based control is to stabilize the relative distances or headway between vehicles in the MVS to achieve predefined task. Formation control, in essence, plays an important role in realizing consensus-based control, as it governs the spacial configuration and relative positions of the MVS vehicle, contributing to the fulfillment of a shared goal [192][149][94].

In [213], a first order consensus-based protocol was implemented. This entailed the utilization of a decentralized control framework built upon the leaderfollower formation approach, ensuring the formation maintenance through information exchange among neighboring vehicles. Additionally, in [73], a control algorithm rooted in consensus-based control was proposed, accounting for communication delays to govern a platoon-based formation. The stability of this formation was mathematically proven using the Lyapunov formalism [132][131]. In this approach, the leader disseminates its state information, while each follower solely measures its relative position w.r.t. the preceding vehicle part of the MVS. This localized approach to formation maintenance was integrated with consensus-based control, even accommodating network and sensor time delays as well as variable leader velocities. Conditions ensuring both internal and string stability of the platoon were established.

Consensus-based control enables the formal modeling of formations by integrating the control of vehicles and the MVS communication topology. Nevertheless, consensus-based control does entail certain drawbacks. For instance, it often simplifies the dynamics of the MVS vehicles by treating them solely as integrators [192]. Furthermore, consensus-based control theory relies on the assumption of continuous connectivity among the vehicles to establish a strongly connected graph, which might not always hold in practice [245].



Figure 2.12. From communication topology to algebraic graph representation for consensus-based control

2.4.5 Discussion

It is evident that the primary focus of the literature on MVS formation control has predominantly revolved around platooning-based formations. Approaches such as leader-follower paradigm (cf. Section 2.4.1) have been extensively employed to tackle various challenges, including platooning modeling [141], string stability (ensuring the stability of the platoon)[90], and information flow [192], among others.

However, the leader-follower approach exhibits certain limitations, particularly in flexibility and adaptability, when dealing with what we refer to as *challenging scenarios*, such as intersection crossings or on-ramp merging (as discussed in Section 2.4.1). To mitigate these limitations, exploring the virtual structure approach (cf. Section 2.4.2) emerges as a potential solution, with a clear imperative for tailored adaptations to suit on-road contexts.

On the other hand, behavior-based formation modeling and control may find its application in scenarios where the main task can be easily divided, and a common goal is distinctly identified by the formation's agents. This approach typically relies on a central entity to define task decomposition and distribution among agents. However, to circumvent the drawbacks of centralization, our work endeavors to promote decentralized approaches.

Consensus-based control offers the advantage of formally modeling formation structures. However, its dependence on the assumption of a strongly connected graph makes it less fault-tolerant, particularly in dynamic environments such as highway scenarios.

Clearly, our literature review has indicated that on-road formation-based approaches have not significantly explored the applications of the virtual structure approach. While this approach may initially seems conceptual, we intend to delve into this topic further in Chapter 3 to explore its feasibility and applications.

2.5 MVS Navigation in complex environments

As previously discussed in Section 1.3, and as evident from the literature review, the coordination challenges within MVS extend to both rural (cf. Figure 2.13) and urban settings. These challenges encompass a range of scenarios, including highway speed coordination (cf. Figure 2.15), merging onto on-ramps and off-ramps (cf. Figure 2.16), and coordination intersections (cf. Figure 2.14) [40][106][248][195]. Additionally, MVS planning technology, known for its substantial impact on road capacity [17][264], has gained significant attention over the past decade. Furthermore, a majority of fundamental research efforts have concentrated on MVS highway speed harmonization [112][178][148][219], as it holds the potential to reduce overall times and average energy consumption [234].

In this PhD manuscript, our primary focus lies on exploring the manifold potential of MVS in the context of highway scenarios:

• MVS highway navigation in formation: Autonomous vehicle formation control, akin to broader concept of highway speed harmonization, exerts an important influence on traffic management, yielding numerous benefits such as heightened road safety, optimized traffic flow, and decreased energy consumption. While human-driven vehicles effortlessly navigate in formation, achieving the same behavior for MVS requires a robust and safe strategy to prevent collisions between the vehicles inside and outside the



Figure 2.13. Rural freeway



Figure 2.15. Highway navigation



Figure 2.14. Intersection crossing



Figure 2.16. Highway on-ramp

MVS formation. Additionally, the question of MVS formation stability demands a formal 1 systematic response.

Employing a safe and reliable formation control method allows for reduced inter-vehicle spacing within the MVS paradigm, thus optimizing the road capacity utilization. Since vehicles contend with aerodynamic drag that consumes energy, optimizing inter-vehicle distances can mitigate this drag effort, resulting in energy savings as one of the benefits of MVS navigation in formation (cf. Figure 2.17) [263][211].

• Cooperative on-ramp merging on highway: Traffic merging at highway on-ramps presents significant safety and mobility challenges. The complexity emerges when the vehicle on the on-ramp must make split-second decisions regarding acceleration or deceleration to merge safely into the desired highway lane, often without a clear line of sight (cf. Figure 2.18).

¹Formal: by formal, we refer to the existence of an analytical mathematical model.



Figure 2.17. Truck platooning [263]

Simultaneously, highway road users must adjust their speeds to accommodate the merging vehicle, potentially disrupting traffic flow and leading to congestion (cf. Figure 2.18).



Figure 2.18. On-ramp merging on highway

To address these challenges, application of cooperative MVS to highway on-ramp merging scenarios can be explored. Various control architectures can be evaluated to their suitability in managing this specific scenario and its associated complexities [195].

As a result, Section 2.5.1 shows a comprehensive review of the highway cooperative navigation relative application through the Cooperative Adaptive Cruise Control (CACC) (cf. Section 2.5.1.1). For the problem of cooperative on-ramp merging on highway, Section 2.5.2 offers a thorough review of relevant approaches. The MVS reviewed application performance are evaluated through the prism of traffic throughput, energy efficiency, and negotiation.

2.5.1 Highway collaborative navigation

Collaborative navigation along a highway entails the intricate task of orchestrating the maneuvers of a fleet of vehicles, collectively referred to as MVS in formation navigation in highway, to function as a cohesive unit known as a formation (cf. Section 2.4). Advances in communication technology have ushered in the era of V2V communication (cf. Section 2.2.3), enabling the exchange of information among vehicles within the same formation.

Traditionally, vehicles on the road tend to follow each other, creating platoonbased formations (cf. Figure 2.19). For human-operated vehicles, this platoon behavior is a natural occurrence, easily observable in everyday driving scenarios. However, for MVS, adhering to lanes and maintaining safe distances and velocities when following nearby vehicles become more complex endeavor. This complexity arises primarily from the need to effectively manage the dynamics of MVS vehicles, underscoring the requirement for cooperative control over these dynamics.

The primary objective of platoon formation control is to ensure that all vehicles within a platoon travel at the same velocity, adhering to the desired one-dimensional spacial configuration, which is stipulated by a predefined inter-



Figure 2.19. Platoon-based navigation in formation on highway

vehicle spacing strategy [120]. Consequently, the formation of a platoon-based shape necessitates the development of specialized algorithms, controllers and strategies [152].

In the forthcoming section, we will introduce the concept of Cooperative Cruise Control (CACC) as an important module for executing cooperative navigation maneuvers on the highway. It is proposed in what follows to delve into collaborative navigation within formation and explore its ramifications on highway throughput, energy efficiency and its synergy with negotiation.

2.5.1.1 Cooperative Adaptive Cruise Control (CACC)

The core of the CACC concept lies on the fusion of Adaptive Cruise Control (ACC), a subset of automated longitudinal speed control systems, with a cooperative module empowered by the V2V communication and/or V2I communication (cf. Section 2.2.3) [208].

Over the past decade, considerable advancements in MVS technologies have been achieved. MVS now possess the capability not only to autonomously navigate using onboard sensors but also to communicate with other vehicles via V2V communication. The terms of CACC have been used with varying interpretations, leading to different perceptions of their functions and capabilities in the context of CACC and platooning-based solutions. In CACC systems, MVS exchange their parameters via the V2V communication without a central management unit [253]. Consequently, CACC leverages V2V communication (cf. Figure 2.19) to enable the vehicles part of the MVS to form platoons and travel at synchronized speed [248]. This synchronization is achieved by minimizing the error d_e between the desired distance d_d (depending on the spacing policy and the current dynamic of the MVS) and the current inter-vehicle distance $d_{(.)(.)}$ (cf. Figure 2.20).



Figure 2.20. Cooperative Adaptive Cruise Control

Through the sharing of vehicle information, including acceleration, speed, and position, MVS with a certain communication range of the roadside unit collaborate to obtain several benefits:

- Enhanced driving safety is achieved through reduced actuation times compared to manual driving, along with improved anticipation of the future actions of the MVS.
- Increased highway capacity results from reduced time/distance headway between the MVS.
- Reduced energy consumption and pollutant emissions stem from minimized unnecessary velocity changes and decreased aerodynamic drag on following vehicles.

The vehicle control strategy plays also a role in CACC systems as it determines the vehicle's dynamics. Specifically, the longitudinal control strategy has been extensively studied, [249][72][176], as vehicles operating in CACC mode must maintain the same longitudinal speed as other vehicles in the platoon while sustaining a constant longitudinal distance/headway relative to their preceding vehicle. Various longitudinal controllers have been proposed to address different objectives, such as platoon formation, optimizing fuel consumption, or ensuring system stability [167][174].

The literature related to the CACC includes several coordination approaches (e.g., consensus-based, optimization-based, etc.). A comprehensive review of these approaches can be found in [249][72][144].

One of the objective of this PhD work is take advantage of the MVS advantages (e.g. safety, passenger comfort, energy efficiency). From this perspective, the optimization techniques are well-suited to include these several criteria. The following sections will undertake an in-depth analysis of various longitudinal controllers founded, based on optimization techniques.

2.5.1.1.1 Optimal Control Optimal control has been employed to optimize energy consumption or travel time in [158]. While many consensus-based control approaches (cf. Section 2.4.4) focus solely on vehicle speed and position, simplifying the longitudinal control to a single or double integrator model under the assumption of linearity, optimal control techniques often consider non-linearity and constraints [249][72][207]. These constraints may involve vehicle power-train dynamics and aerodynamics [231].

In numerous cases where optimal control is applied to CACC systems, the objective function minimizes the total energy consumed by the vehicle traveling within a designated area. For instance, in the Eco-CACC system proposed in [257],

it computes the fuel-optimum vehicle trajectory through a signalized intersection, the optimal control problem is defined as:

$$\min_{a_{-},a_{+}} \int_{t_{0}}^{t_{0}+T} F(v(t),v'(t))dt$$
(2.1)

subject to

$$\int_{t_0}^{t_0+T} v(t)dt = d+l$$
 (2.2)

$$0 \le a_{-} \le a_{-}^s \tag{2.3}$$

$$0 \le a_+ \le a_+^s \tag{2.4}$$

Here, $F(\cdot, \cdot)$ represents a nonlinear function of speed v(t) and acceleration v'(t), estimating the energy consumption rate based on vehicular speed and acceleration levels. a_{-} and a_{+} are the upstream deceleration and the downstream acceleration levels, respectively. d and l are the length of the control zone before and in the signalized intersection, respectively.

2.5.1.1.2 String stability String stability is a fundamental requirement to ensure the safety of the CACC system. It aims to attenuate distance error, velocity, or acceleration along the upstream direction in a platoon, as discussed in [188]. The problem of string stability can be formulated as:

$$|y|_{\infty} \le |u|_{\infty} \tag{2.5}$$

Here, y represents the scalar output of distance error, velocity, or acceleration of the following vehicle i + 1, and u represents the scalar output of distance error, velocity, or acceleration of the preceding vehicle i. String stability is guaranteed if:

$$\left|\frac{Y(s)}{U(s)}\right|_{\infty} \le 1 \tag{2.6}$$

where Y(s) and U(s) are the Laplace transforms of y and u, respectively.

Numerous works have analyzed the string stability of CACC systems [249][176] [133], and some conclusions have been proposed to ensure string stability:

• If a constant distance spacing policy is adopted for vehicle spacing, the predecessor-follower (cf. Section 2.2.3) information flow may not guarantee string stability. Extending the information by broadcasting the leader's

information to the following vehicles, using a predecessor-follower-leader information flow (cf. Section 2.2.3), for example, can ensure string stability.

• To relax the formation rigidity imposed by the constant spacing policy, a constant time headway spacing policy can be employed, ensuring string stability by allowing the inter-vehicle distance to depend on the vehicle's velocity [249].

2.5.1.1.3 Model Predictive Control (MPC): Traditionally, Model Predictive Control (MPC) is framed within the state space framework for single vehicle, where controlled system is described by a linear model as expressed in [214]:

$$\dot{x}(t) = Ax(t) + Bu(t), \quad x(0) = x_0$$
(2.7)

Here, $x(k) \in \mathbb{R}^n$ represents the state input, and $u(k) \in \mathbb{R}^p$ represents the control input. With n and p are the the state variables and inputs numbers respectively. A is the state matrix, with $dim[A] = n \times n$, and B is the input matrix, with $dim[B] = n \times p$. x_0 is the initial state vector. Typically, a receding horizon implementation is formulated as an optimization problem, such as:

$$J_{u(t)}(x_0) = \min_{a(t)} \int_0^T [q_f(v(t), a(t)) + \gamma] dt$$
(2.8)

Subject to:

$$\Delta \dot{x}(t) = v_p(t) - v(t) \tag{2.9}$$

$$\dot{v}(t) = a(t) \tag{2.10}$$

$$\Delta x(t) \ge \Delta x_{\min,0} + h_{\min}v(t) \tag{2.11}$$

$$\Delta x(t) \le \Delta x_{\max,0} + h_{\max} v(t) + \gamma r \tag{2.12}$$

$$a_{\min} \le a(t) \le a_{\max}(v(t)) \tag{2.13}$$

$$v_{\min} \le v \le v_{\max} \tag{2.14}$$

In this context, q_f represents the current fuel consumption depending on v(t)and a(t), where v(t) is the ego-vehicle's velocity and a(t) is its acceleration. $v_p(t)$ represents the preceding vehicle's velocity. $\Delta x(t)$ denotes the actual inter-vehicle distance, with $\Delta x_{\min,0}$ and $\Delta x_{\max,0}$ being the minimum and maximum distances when stationary. h_{\min} and h_{\max} correspond to the minimum and maximum time headway, associated with the minimum and maximum inter-vehicle distances. The relaxation aspect is introduced through the slack variable γ and relaxation parameter r. Constraints are also applied to acceleration $(a_{\min} \text{ and } a_{\max})$ and velocity $(v_{\min} \text{ and } v_{\max})$.

Typically, centralized MPC systems assume knowledge of all the states for computing control inputs. However, in practical applications, especially in dynamic scenarios like highway traffic, gathering information from all the vehicles to compute a large-scale optimization problem is not always feasible. This limitation has led to the development of Distributed Model Predictive Control (DMPC) [53][172], and stochastic MPC (SMPC) to include the uncertainty of the system [168].

2.5.1.2 Evaluation of the highway collaborative navigation approaches

As discussed in Section 1.3, MVS collaborative navigation is linked with several advantages. In this section, an extensive examination of MVS formation navigation in highway is presented, with a focus on aspects such as traffic throughput, energy efficiency, and negotiation.

2.5.1.2.1 Traffic throughput The primary goal of the CACC systems is to establish and regulate platoon-based formations. Consequently, the advantages of CACC systems can be considered as benefits for formation-based navigation on highways.

The impact of platoon enabled by CACC on highway throughput was investigated in [235]. In this research, a stochastic microscopic-traffic simulation model was developed to assess various of traffic flow performance, including safety, exhaust-gas emissions, and noise emissions. The proposed model utilizes real traffic measurements, such as instances, lanes speeds, and vehicle lengths, to generate traffic scenarios at the beginning of simulations runs. The simulations demonstrated that the presence of more CACC-equipped vehicles in traffic leads to higher average velocities. Regarding traffic throughput, scenarios with 100 % CACC-equipped vehicles showed the best performance compared to other mixing penetration scenarios.

In [143], a CACC modeling framework was proposed, incorporating interactions between CACC-equipped vehicles and manually driven vehicles in mixed traffic. This framework included lane-changing rules and automated speed control to ensure realistic CACC performance. Several simulations were conducted on a 4-lanes freeway segment with an on-ramp and off-ramp lanes. The case study compared basic CACC-operated vehicles with Multi-Lanes CACC (ML-CACC)- operated vehicles for different market penetrations. Results indicated that traffic flows with CACC market were consistently lower that those with ML-CACC.

The study in [123] shows the effect of platooning on mixed traffic and its impact on improving highway throughput. To address this scenario, a new fluid model of mixed-autonomy traffic flow was introduced, which was then used to analyze and design platoon coordination strategies. Simulations were conducted to study the effect of platoon penetration on highway throughput. The results showed that traffic throughput increased with higher platoon fractions on the highway. The study also explored the impact of platoon size, revealing the presence of an optimal platoon size.

2.5.1.2.2 Energy efficiency Vehicle platooning offers a significant opportunity to enhance energy efficiency by substantially reducing the distance between vehicles, thereby decreasing the aerodynamic drag coefficient [244]. The extent of drag reduction in platooning depends on factors such as the vehicle shapes with the platoon, their arrangement, and the distance between them. Savings are most pronounced for vehicles positioned in the middle of the platoon, and the overall savings increase with the number of vehicles in the platoon. For instance, when two vehicles maintain 1-meter gap between them, the average gap reduction is estimated to be around 10 % [269]. In cases where the platoon consists of a mix of vehicle types, drag reduction has been estimated at 20% [201], and even up to 40 % [78]. In scenarios involving long platoons of vans (five or more vehicles) spaced 0.5 to 1-meters apart, drag reduction of up to 44 % to 55% has been reported in [201].

Platoon-based formations facilitated by CACC systems also lead to energy efficiency improvements by minimizing unnecessary changes in velocity. In [250], an approach was proposed to minimize energy consumption and pollutant emissions within platoons during various stages, including sequence determination, gap closing and opening, platoon cruising with gap regulation, and platoon joining and splitting. Compared to consensus-based (cf. Section 2.4.4) CACC system, the results demonstrated that an ECO-CACC (cf. Section 2.5.1.1) could reduce global energy consumption by 1.5 % during platoon formation and 2 % during platoon reconfiguration phases [250].

In [42], an energy consumption model was developed for various types of vehicles, including internal combustion light-duty vehicles, electric vehicles, hybrid electric vehicles, buses and trucks. This model aimed to quantify the effects of platooning on fleet fuel consumption. The findings indicated that energy consumption reductions of up to 3%, 3.5%, 4.5%, 10%, and 15%, respectively.

It becomes evident that vehicles platooning is especially well-suited for heavyduty vehicles, particularly on highways where travel is at high speeds, leading to substantial aerodynamic drag. Trucks, which often cover long distances on highways, can benefits from joining neighboring trucks to form platoons, even if they have different starting points and destinations. For an in-depth analysis of fuel economy in truck platooning, refer to [262].

2.5.1.2.3 Negotiation To date, the literature lacks of substantial exploration of negotiation- and agreement-based approaches concerning navigation in formation, applied to the case of highways and MVS formation reconfiguration in highway (cf. Figure 2.21) [152]. This absence may stem from several factors, including unsuitability of such approaches or simply insufficient exploration of these possibilities. One key reason for scarcity of negotiation in this context could be the limited scope for negotiation or discussion with a platoon during cruising. For instance, while one might argue that leader selection could be a negotiation topic, leaders are typically chosen strategically and functionally, often as the vehicle positioned at the front of the formation. Similarly, negotiation speed profiles may not be a significant consideration since the primary goal of a platoon is to travel safely at the maximum allowable speed [152].



Figure 2.21. Negotiation-based merging in highway

However, a potential avenue for exploration is negotiation platoon reconfiguration. Platoon-based formations operate with dynamic environments, such as highways, and are subject to various maneuvers, including vehicles joining, splitting, or changing lanes (individually or as a formation). These maneuvers need to be executed with a global perspective to ensure safety and efficiency, while also being fair in terms of the efforts required from each vehicle. For instance, consider the splitting maneuver depicted in Figure 2.21. Platoon members could engage in discussions where each vehicle proposes its solution to create the desired gap. A negotiation mechanism could then be employed to select the best proposition, taking into account safety and efficiency.

In practice, scenarios like these are often managed by the platoon leader [17]. In this work, the authors introduced a platoon management protocol supporting three fundamental maneuvers: merging, splitting, and lane changes. The reconfiguration protocol relies on centralized platoon coordination, where all maneuvers are decided and planned by the platoon leader, and followers follow orders and send requests to and from the leader. However, this approach has limitations associated with centralization (cf. Section 2.3.1), such as restricted flexibility (i.e., typically limited to the communication range) and a lack of formal methods for analyzing the stability of the proposed solutions.

In addition to the above cooperative highway navigation approaches, on-ramp merging on highway is also one scenario subject to MVS cooperative navigation. The following section presents an overview of the approaches related to MVS on-ramp merging on highway.

2.5.2 Collaborative on-ramp merging on highway

The merging of traffic at highway on-ramp presents a significant challenge, leading to safety and traffic flow concerns. This challenge becomes particularly pronounced for the merging vehicles, as they must make real-time decisions regarding acceleration or deceleration to safely integrate into the mainlines traffic, often without a clear line of sight. Simultaneously, the drivers on the highway may need to adjust their speeds to facilitate the smooth entry of merging vehicles, potentially, disrupting traffic flow and can induce congestion. In response to these complex issues, researchers have explored the concept of cooperative merging, specifically involving MVS, to address the merging problem at highway on-ramp.

One can note that the corner stone idea behind collaborative on-ramp merging on highway is to minimize the global effort provided by each vehicle while performing the merging scenario. Naturally, the idea of Prioritizing the stability of the already established vehicle groups (platoons) arises as an evidence. The works in [203][95][256] propose several strategies for vehicle merging in an already established platoon with low collaborative efforts.

Various control algorithms have been proposed and implemented to enable the vehicles part of the MVS to merge with one another in a cooperative manner. An extensive review of existing research can be found in [195][28][270][41]. In this section, it is proposed to evaluate the existing literature related to cooperative on-ramp merging on-highway through the prism of traffic throughput (cf. Section 2.5.2.1), energy efficiency (cf. Section 2.5.2.2), and its implication with negotiation (cf. Section 2.5.2.3).

2.5.2.1 Traffic throughput

Enhancing highway traffic flow throughput seamless merging at highway on ramps has been a widely explored challenge in the literature. Initially, centralized approaches (cf. Figure 2.22) were introduced, utilizing a merging coordinator with support of V2I communication. For instance, in [202], a two-layer centralized controller was proposed based on heuristic rules derived from empirical observations of system behavior. The first layer established the merging sequence by estimating each vehicle's merging time, assuming constant-speed travel. The second layer determined the necessary constant acceleration to resolve conflicts identified during the merging sequence. However, heuristic-based approaches are limited in their adaptability to dynamic environments, and their optimality is not rigorously proven due to the absence of formal optimization algorithms approach.

To address the optimality concerns in passing sequences, several centralized methods have been developed in the literature, focusing on optimizing travel time to increase highway throughput. These methods use the communication coordinator's responsibility to optimize the passing sequence order based on merging times. An example of optimizing problem formulation is developed below [195], the particularity of this approach resides on reliance on the merging zone crossing time, what makes it less road geometry dependent and more generic:

$$\min_{u} \frac{1}{2} \sum_{j=1}^{m} \sum_{i=1}^{n} \left[t_{j,i}^{\text{out}} - t_{j,i}^{\text{in}} \right]^2$$
(2.15)

Subject to:

$$\dot{x}_{j,i} = v_{j,i} \tag{2.16}$$

$$\dot{v}_{j,i} = u_{j,i} \tag{2.17}$$

$$t_{j,i}^{\text{out}} - t_{j,i}^{\text{in}} \ge \Delta T_a \tag{2.18}$$

$$o < v_{j,i}(t,u) \le v^{\max} \tag{2.19}$$

$$x_{j,i}(t) \le x_{j,i+1}(t) + \delta \quad \forall t \tag{2.20}$$

$$x_{j,i}(t) \le x_{p,q}(t) + S \quad \forall t, j \ne p, i \ne q$$

$$(2.21)$$

where $t_{j,i}^{\text{in}}$ and $t_{j,i}^{\text{out}}$ are the times that the vehicle *i*, on the road *j*, enters and exits the merging zone, ΔT_a is the minimum allowed time to cross the intersection at the maximum speed v^{\max} , and δ is the desired safe distance between vehicles on the same road. *S* is the length of the shared zone.



Figure 2.22. Centralized collaborative on-ramp merging on highway

Another approach, as presented in [195], offered an optimization framework with an analytical closed-form solution for online vehicle coordination at merging zones. Simulations demonstrated significant reductions in travel times and improvements in traffic flow. In addition, in [63], the authors introduced an approach that virtually positioned vehicles onto highway's main lane before actual merging, ensuring smooth and safe merging maneuvers. The results indicated substantial improvements in traffic flow when a high percentage of MVS vehicles were involved.

Discussion While centralized methods have shown promise in addressing highway on-ramp merging, they come with certain limitations. These approaches heavily rely on communication, exacerbating any communication-related issues that may arise. Moreover, their critical flaw lies in fault tolerance; if the central coordinator fails, the entire system may fail.

To mitigate these issues, decentralized approaches (cf. Figure 2.23) can be considered. These approaches draw inspiration from centralized methods, such as the concept of virtual vehicles/platooning.

In the virtual vehicle/platooning approach, each merging vehicle communicates its position and velocity within a certain range. Based on this data, each vehicle constructs its view of the merging scenario, replacing real-vehicles with virtual ones. Passing orders are then generated individually by each vehicle part of the MVS, aided by the virtualization. Conflict checks are performed to prevent merging collisions, with adjustments made if conflicts arise, depending on the merging strategy. An advanced variation of this approach introduced



Figure 2.23. Decentralized collaborative on-ramp merging on highway

the concept of slots², allowing the vehicles part of the MVS to navigate within virtual slots. Conflict checking is conducted across these slots, accommodating localization uncertainties and ensuring minimal safety distances [153]. In comparison with centralized methods, this cooperative merging method demonstrated superior performance in terms of traffic throughput and average on-ramp vehicle delay.

2.5.2.2 Energy efficiency

Merging onto a dense highway from an on-ramp presents a complex challenge, characterized by its combinatorial nature and the limited information available regarding average speed in the different lanes [206]. Additionally, ramp merging often leads to congestion and the formation of phantom traffic jams³ [234]. V2V communication (cf. Section 2.2.3) offers a solution by enabling the anticipation of neighboring vehicles' intentions. This advanced knowledge of lane speeds can enhance the eco-driving control algorithms of MVS [234], resulting in more informed and smoother merging maneuvers. According to [193] [195], these improvements can lead to an increased energy efficiency. Moreover, the benefits may extend beyond individual vehicles; by reducing the occurrence of phantom jams, the overall energy efficiency of traffic can be enhanced. However, it is worth noting

 $^{^2{\}rm A}$ slot is a geometrical space in the highway mainline, defined by its coordinates w.r.t. a common reference frame within the MVS, usually used by the vehicle for negotiation purposes to avoid collision

³Phantom traffic jams: dense traffic crawls to a halt for no apparent reason.

that very few works have explored this particular advantage of MVS coordination on merging [234].

2.5.2.3 Negotiation

In the absence of road coordinator and predefined heuristic rules, a vehicle seeking to merge into a lane from an on-ramp vehicle must ensure two conditions: (a) avoiding collisions with vehicles already in the targeted lane, and (b) executing a smooth merge without abruptly slowing down the vehicles on the target lane [152].

The approaches proposed in [17] and [18] rely on V2V communication (cf. Section 2.2.3) through the Vehicular Ad-Hoc network VANET protocol to facilitate negotiation. The merging vehicle initiates action proposals to vehicles already on the target lane, with the latter having the right to decline and propose alternative actions. In congested traffic conditions, such approaches are susceptible to issues of resource allocation, as vehicles on the highway may be unwilling or unable to accept proposals of cooperation from on-ramp vehicles. In other terms, the allocation of resources (space and time) for vehicles willing to merge from on-ramps to highway can be problematic due to reluctance or inability of vehicles already on the highway to accept this merging proposals.

To address the challenge of resource allocation, negotiation approaches based on incentive mechanisms can be conceptualized. For instance, merging vehicles could employ auctions [7][9][67], where they offer compensation to enter the target lane. However, as of now, no relevant examples of such mechanisms have been found in the literature.

2.6 Conclusion

This chapter introduced various paradigms employed in the construction of the multi-vehicles system (MVS) control architecture. Additionally, it delved into the concept of formation control through a comprehensive review of the diverse approaches utilized for this purpose. Furthermore, this chapter explored the relevant literature concerning the versatile application of MVS, where the main challenges related to the different applications were highlighted.

Addressing the challenges of MVS in dynamic and complex on-road environment, such as on-ramp merging and navigation in formation in highway, requires a multifaceted control architecture. The key takeaways from this chapter are listed as follows:

1. Formal modeling of the MVS formation: The formation composed by the vehicles part of the MVS should be done following a formal approach.

Formal modeling helps to prove the stability of the formation motion, provides safety proofs, and enhances the cooperation efficiency.

- 2. **Distributed solution:** The proposed solution should aim for a high degree of decentralization to minimize dependencies on central units.
- 3. **Promoting cooperation:** To prevent selfish behaviors, negotiation mechanisms should be an integral part of the solution to encourage cooperative interactions. Additionally, the notion of altruism coupled with MVS cooperation can be interesting to mitigate this issue.
- 4. **Performance demonstration:** The effectiveness of the proposed solution should be demonstrated in through the prism of its effects on safety, traffic flow, passenger comfort, and energy efficiency.
- 5. **Robustness:** The decision-making layer must prioritize robustness to ensure safe operations, and risk management methods should be integrated to maintain collision-free interactions.
- 6. Uncertainty assessment: Recognizing that uncertainties are inherent to the problem, the proposed solution should incorporate mechanisms to address these uncertainties effectively.

Part II

Cooperative Multi-Controller Architecture

CHAPTER **3** The proposed Cooperative Multi-Controller Architecture (C-MCA)

Abstract

In this Chapter an global overview of the proposed Cooperative Multi-Controller Architecture (C-MCA). Drawing inspiration from the foundational multi-controller architecture, a brief summary of the latter is presented. Subsequently, the chapter delves into the key functionalities of the C-MCA, with a particular focus on decision-making and planning at the system level. The chapter concludes with a dedicated section addressing the problem statement related to the this PhD work.

Contents

3.1	Introd	uction	
3.2	Multi-	controller	architecture
3.3	Overa	ll Coopera	ative Multi-Controller Architecture
	3.3.1	The C-N	ACA main functionalities
		3.3.1.1	Environment perception
		3.3.1.2	Communication module
		3.3.1.3	Global path planning 80
		3.3.1.4	Decision-making level
		3.3.1.5	The local trajectory planning level
		3.3.1.6	The control level
3.4	Proble	em staten	nent

	3.4.1	Scenario	$ topology \dots \dots \dots \dots \dots \dots \dots \dots \dots $	3
	3.4.2	Control	law	4
3.4.3		Formation definition and configuration 8		
		3.4.3.1	Formation modeling based on a dynamic reference	
		3.4.3.2	frame	7
			frame	8
3.5	Conclu	usion .		9

3.1 Introduction

The overarching goal of this PhD work centers on building a decision/control architecture that takes advantage from the MVS coordination ability to overcome the challenges related to on-ramp merging on highway. Consequently, two main sub-objectives related to the decision/control were identified: (1) Developing a suitable approach for MVS navigation in formation for on-ramp merging on highway, and (2) Build a decision-making level designed to mitigate the on-ramp merging challenges.

The examination of the MVS paradigm and its associated control architectures in Chapter 1 and Chapter 2 have led to the realization that classical control architectures come with inherent limitations. Keeping the idea of building a decision/control architecture suitable for MVS on-ramp merging on highway, while mitigating the classical control architecture limitations. This chapter delves into the proposed Cooperative Multi-Controller Architecture (C-MCA).

The C-MCA is based on the foundational multi-controller architecture, consequently, Section 3.2 elaborates the multi-controller architecture background relevant to this PhD work. Section 3.3 presents an overview of the proposed Cooperative Multi-Controller Architecture (C-MCA). Lastly, section 3.4 outlines the problem statement tackled in this work.

3.2 Multi-controller architecture

The principal aim of the Multi-Controller Architecture (MCA) is to systematically decompose the fundamental functions of the MVS, in order to break down the complexity of the tackled scenario. In essence, it endeavors to disassemble the overarching task into a sequence of sub-tasks. Consequently, implementing the MCA necessitates the development of reliable elementary controllers and specialized mechanisms, often referred to as behaviors, to proficiently manage their interactions.

The MCA, depicted in Figure 3.1, serves as the foundational framework for our research. Its purpose is to oversee interactions among elementary behaviors while upholding the overall control system's stability, as discussed in [116]. The MCA, featured in Figure 3.1, primarily encompasses three elementary behaviors: lane keeping, adaptive cruise control and lane change. It is pertinent to highlight that the underlying operational principals of three two behaviors have been extensively expounded upon in the literature (cf. Chapter 2). Our particular interest lies in the adaptation of the behavioral concept for MVS on-ramp merging on highway navigation, as well as, the selection process part of the decision-making of the MCA in Figure 3.1.

During each sampling interval, one of these three behaviors is activated via a the behavior selection mechanism (cf. Figure 3.1) part of the decision-making level based on inputs from the perception, localization and communication modules. Each vehicle's controller consists of a dedicated and uniform set of waypoints, defined by pose $X_T = [x_T, y_T, \theta_T]$ and the desired velocity v_T . It is crucial to underscore that, once these target points are established, the controllers must rely on robust control laws to consistently reach and track these designated target points.



Figure 3.1. Multi-Controller Architecture

Building upon the MCA paradigm outlined in this section, the subsequent

section provides an overview of the main functionalities of the proposed C-MCA tailored for merging onto highway on-ramps in the context of MVS. Special attention is given to the decision-making and planning levels.

3.3 Overall Cooperative Multi-Controller Architecture

3.3.1 The C-MCA main functionalities

The primary focus of this work revolves around adapting the MCA detailed in Section 3.2, to suit the requirements of cooperative navigation within the MVS paradigm in complex and dynamic environments. More precisely, it is proposed to employ the MCA as the foundational framework in the development of a Cooperative Multi-Controller Architecture (C-MCA) (cf. Figure 3.2) designed to overcome the challenges related to MVS navigation on on-ramp merging and cooperative highway navigation.

The proposed C-MCA is depicted in Figure 3.2. It has been designed around various interconnected modules that facilitate planning, control, access, and management of on-ramp merging and cooperative highway navigation within MVS. This section aims to establish a global focus of the main functionalities part of the C-MCA, whith an emphasis on the decision-making level (cf. Figure 3.2 (1)) and the local trajectory planning (cf. Figure 3.2 (2)).

3.3.1.1 Environment perception

As discussed in both Section 2.2.1 and Section 2.2.2, the roles of the perception and localization modules involves furnishing the essential environmental characteristics indispensable for MVS navigation. These characteristics encompass critical information such as the vehicle's precise localization, the number of lanes, lane markings, distances to road boundaries, and the pose of detected obstacles. To achieve this, these modules operate based on a vector map, serving as a detailed representation of the environment. Fusion algorithms, combining data from perception and localization, are employed to generate a reliable representation of the environment.

Assumption 1: It is essential to underscore that this PhD manuscript concentrates on the decision-making and planning/control modules with the C-MCA. Consequently, the perception, localization, and vector map modules are assumed to be reliable and are not the primary focus of this research.



Figure 3.2. Cooperative Multi-Controller Architecture

3.3.1.2 Communication module

This module operates in collaboration with the Road Side Unit (RSU) positioned within the merging zone (cf. Figure 3.4). The RSU acts as a conduit for the exchange of crucial information between the merging vehicle and the highway vehicles, thus enhancing of the MVS's motion coordination capabilities. In essence, when referring to the communication aspect of this research, it pertains to the exchange of information between the vehicle components of the MVS and the RSU, all within the communication range defined in Figure 3.4. Further details related to the RSU can be found in Appendix A).

Assumption 2: It is important to note that the communication module, both within the vehicles and integrated into the RSU, functions as tools aimed at enhancing the overall abilities of the MVS. However, it is crucial to acknowledge that the complexities inherent in the levels addressed in this research preclude the comprehensive investigation of communication-related aspects such as communication delays, packet losses, and dynamic communication topologies.

3.3.1.3 Global path planning

The global path planner module gives a selected sequence of way-points through the road network. Additionally, it has the responsibility of setting the goal of the vehicle (e.g., its destination, navigating lane, etc.). It uses the inputs of the perception modules to cope with the traffic road rules (e.g., speed limits, traffic signs, etc.).

3.3.1.4 Decision-making level

After defining the environmental perception and the vehicle's global path, it becomes imperative to adopt an appropriate decision-making strategy for the vehicle component within the C-MCA. This strategy must consider various factors, including the vehicle's objectives, the MVS overarching objectives, traffic regulations, etc.

The core requirement of the decision-making level is to ensure the safety of the MVS during the on-ramp merging maneuver. Consequently, given the shared nature of on-ramp merging scenario and the high dynamic relative to the highway, the decision-making must establish a strategy to solve the conflicting scenarios. Before delving into the presentation of the proposed decision-making level part of the C-MCA, a short overview of the literature approaches used to solve the conflicts part of the shared zones is proposed.

As with intersection crossing, the merging conflict is mainly justified because of the shared nature of the merging zone [152]. Thus, solving conflicting scenarios requires an appropriate strategy designed to take into account explicitly these conflicting configurations. Several strategy can be found in the literature and they can be mainly classified in two categories: (1) Centralized approaches and Decentralized approaches. In Centralized approaches, a Central controller defines the passing sequence of the vehicle in the shared zone, and the vehicles have no words concerning the selection policy. In contrast, in decentralized approaches, negotiation-based strategy can be used to establish the passing sequence. Some contributions part of the intersection crossing and on-ramp merging literature focused on the competitive nature of the scenario [152]. Thus, the use of negotiation can be performed with auction-based mechanism. While approaching the shared zone, each vehicle can contact the RSU and place a bid; place an offer to buy a portion of the shared zone for a certain period. The value of the bid expresses the urgency of the vehicle. The RSU collects the bids and accord the conflict portion to the vehicle that placed the highest bid [51][46][237]. The main limitation of the auction-based strategy is the liveness of the method, i.e., as explained in [152] liveness stands for starvation of the vehicles; in some cases, the constant bidding strategy of the vehicle can prevent other vehicles from wining an auction, with the risk for them an indefinitely waiting time.



Figure 3.3. Flowchart of the decision-making level part of the C-MCA

In this PhD work, instead of relying on the competitive nature of the onramp merging scenario, it is proposed to rely on its cooperative nature. Consequently, the conflicting is solved while ensuring the respect of the MVS overarching goal and vehicle's individual goals. The decision-making level within the C-MCA module is founded upon a multi-behaviors decision-making system. This decision-making module (cf. Figure 3.2 (1)) is responsible for selecting one of the two operational behaviors within the local trajectory level (cf. Figure 3.2 (2) and Figure 3.3), with respect to the safety criterion. In simpler terms, the decision-making level prompts the nominal behavior (responsible for promoting the vehicle's individual goals) to make predictions about the vehicle's expected nominal motion (cf. Figure 3.3 and Figure 3.2 (2)). Based on these predictions, the safety criterion is assessed, if the nominal behavior is safe then the latter is activated. However, when a conflicting merging is detected based on the nominal behavior, the decision-making level and the cooperative behavior works together to solve the conflict (cf. Figure 3.3 and Figure 3.2 (1) (2)), while respecting both the MVS overarching goal and the vehicle's individual goals. Further details about the multi-behavior decision-making level part of the C-MCA can be found in Chapter 5.

3.3.1.5 The local trajectory planning level

The local trajectory planning level part of the C-MCA in Figure 3.2 is composed of two primary behaviors:

- 1. The nominal behavior: It takes into account data from the environment perception, and global path planning modules to predict the expected behavior of each vehicle within the MVS. The nominal behavior is designed to achieve the MVS vehicle's individual goals, thus it is optimized to enhance the performance of the involved vehicle. The forecaster behavior is subsequently evaluated using a safety metric by the decision level within the C-MCA, further details can be found in Section 5.2.
- 2. The cooperative behavior: In cases where the nominal behavior fails to adhere to the safety criterion, the C-MCA decision-making level activates the cooperative behavior. Consequently, the cooperative behavior receives additional inputs, including safety evaluations related to the nominal behavior, in addition to the selected passing sequence. Based on these inputs, the cooperative behavior has the responsibility of generating the vehicles' dynamics corresponding to the passing sequence. The translation of the passing sequence into the MVS dynamics is ensured by the formation control strategy part of C-MCA. Further details about the formation control strategy are given in Chapter 4.

3.3.1.6 The control level

The control level is tasked with the precise tracking of the set-points generated by the local planning level. These set-points serve as a reference for the control to follow. The local trajectory planning level creates a velocity profile and yaw rate that are well-suited for the vehicle.

The used control law is the one detailed in Section 3.4.2. The control law is synthesized for a tricycle model-based vehicle (cf. Section 3.4.2) using Lyapunov synthesis and is detailed in Section 3.4.2.

3.4 Problem statement

As previously discussed in both Chapter 2 and Chapter 3, and as highlighted by the comprehensive literature reviews in [40] [106] [248] [195], the coordination challenges confronting MVS extend their influence over a wide array of environments, encompassing both rural and urban settings, including highways. These challenges manifest in a multitude of scenarios, including the imperative for coordination on highways to maintain uniform speeds, as well as the intricacies of merging onto and exiting from on-ramps. As elucidated in Section 1.3, the cooperative navigation of MVS on highways ushers in an array of driving benefits, ranging from enhancing safety to improved traffic flow and optimized energy efficiency (cf. Section 1.3).

In this section, we aim to provide an overview of the background that underlies this PhD thesis. Section 3.4.1 will present a representation of the highway scenario involving on-ramp, thereby establishing the overarching road topology that serves as the foundational scenario of this work. Subsequently, the control law used in this PhD work is detailed. Lastly, the formalism definition and modeling of the formation are discussed.

3.4.1 Scenario topology

This research endeavors to leverage the capabilities of MVS to confront the intricacies associated with cooperative highway navigation. More precisely, it sets out to address the challenging scenario of cooperative merging at highway on-ramps, a scenario associated with significant safety and mobility complexities. The specific use at the heart of this research is illustrated in Figure 3.4, encompassing a multi-lane highway environment incorporating an on-ramp as the entry point to the highway section. Additionally, this scenario accounts for a designated merging zone, characterized by a fixed length, where merging maneuvers are authorized. Furthermore, the prescribed speed limits governing both the highway and the on-ramp segments are integrated as predetermined parameters with this scenario.

The distinctive feature of the on-ramp merging scenario is the complex motion synchronization hurdles encountered by the vehicles. These challenges material-



Figure 3.4. Illustration of the scenario topology of the on-ramp merging on highway

ized as vehicles on the on-ramp must make decisions about whether to accelerate or decelerate. All this, while ensuring a safe and continuous merge onto the desired highway lane, often under conditions where a clear line of sight is not guaranteed. Simultaneously, highway users must adapt their speeds to accommodate to the behavior of the merging vehicle, which can potentially disrupt the flow of traffic and result in congestion. One way to help synchronization ability of the vehicles participating in the merging scenario is with the help of communication. In this work, we take the assumption of a on-ramp merging scenario equipped with a Road RSU (cf. Appendix A).

3.4.2 Control law

Before presenting the used control law, it is important to know the vehicle's model.

Assuming that the vehicle evolves in asphalt road and in cluttered urban environment with relatively low speed, the following model is based on tricycle model [238]. The two front wheels are replaced by a single virtual wheel located at the center of both front wheels.

$$\begin{aligned} \dot{x} &= v \cos(\theta) \\ \dot{y} &= v \sin(\theta) \\ \dot{\theta} &= v \tan(\gamma) / l_b \end{aligned} \tag{3.1}$$

 (x, y, θ) is the vehicle's pose on the global reference frame X_G, Y_G . v is the vehicle's linear velocity and γ is the steering angle of the vehicle. l_b is the vehicle's wheelbase. v and γ are the control inputs of the vehicle (cf. eq 3.1).


Figure 3.5. Vehicle's and target's configuration in global $(X_G; Y_G)$ and local $(X_m; Y_m)$ reference frames, and the control variables [238]

According to Figure 3.5, w_b corresponds to the track width of the vehicle and I_{cc} the instantaneous center of curvature of the vehicle trajectory. The radius of curvature r_c is given by:

$$r_c = \frac{l_b}{tan(\gamma)} \tag{3.2}$$

and $cc = 1/r_c$ is the curvature of the vehicle trajectory.

The used control law [239] aims to drive the vehicle toward specific targets (static or dynamic) in the environment. At each sample time the tracked target is defined by a posture (x_T, y_T, θ_T) and velocity V_T (this velocity could be equal to 0 if the target is static). Using a Lyapunov formulation, the used control law [239] is presented below.

The adopted Lyapunov function V is given by eq. 3.3.

$$V = \frac{1}{2}K_d d^2 + \frac{1}{2}K_l d_l^2 + K_o \left[1 - \cos\left(e_{\theta}\right)\right]$$

= $\frac{1}{2}K_d d^2 + \frac{1}{2}K_l d^2 \sin^2\left(e_{RT}\right) + K_o \left[1 - \cos\left(e_{\theta}\right)\right]$ (3.3)

where the initial values of e_{RT} and e_{θ} must satisfy the following initial conditions:

$$e_{RT} \in]-\pi/2, \pi/2[\text{ and } e_{\theta} \in]-\pi/2, \pi/2[$$
 (3.4)

The Lyapunov function (cf. eq. 3.3) is therefore a function of three parameters which depend on: the distance d between the target and vehicle's position; the distance d_l from the vehicle to the target line (line that passes through the target position with an orientation equal to the target orientation), this term is related to the Line of Sight and Flight of the target; and the orientation error e_{θ} between the vehicle and the target.

The desired linear velocity v and the front wheel steering γ of the vehicle which allow to asymptotically stabilize the error vector $(e_x, e_y, e_\theta, (v - v_T))$ toward zero (permitting therefore to have $\dot{V} < 0$) are given by:

$$v = v_T \cos(e_\theta) + v_b$$

$$\gamma = \arctan(l_b c_c)$$
(3.5)

where v_b and c_c are defined by:

$$v_b = K_x \left[K_d e_x + K_l d \sin\left(e_{RT}\right) \sin\left(e_\theta\right) + K_o \sin\left(e_\theta\right) c_c \right]$$
(3.6)

with:

$$c_{c} = \frac{1}{r_{c_{T}}\cos(e_{\theta})} + \frac{d^{2}K_{l}\sin(e_{RT})\cos(e_{RT})}{r_{c_{T}}K_{o}\sin(e_{\theta})\cos(e_{\theta})} + K_{\theta}\tan(e_{\theta}) + \frac{K_{d}e_{y} - K_{l}d\sin(e_{RT})\cos(e_{\theta})}{K_{o}\cos(e_{\theta})} + \frac{K_{RT}\sin^{2}(e_{RT})}{\sin(e_{\theta})\cos(e_{\theta})}$$
(3.7)

 $K = (K_d, K_l, K_o, K_x, K_{\theta}, K_{RT})$ is a vector of positive constants defined by the designer. Accurate analysis of this stable and efficient control law is given in [239][238].

3.4.3 Formation definition and configuration

Based on the MVS paradigm (cf. Section 1.2.2) and the communication range of the RSU (cf. Figure 3.4), a set of N vehicles involved in creating the formation is identified. In other words, the RSU establishes connections with all the vehicles within its communication range. Consequently, the RSU assigns a unique communication identifier, in addition to the definition of the reference vehicle used

as a reference frame by the formation coordinate system. This strategy is defined as follows:

- Reference vehicle: The RSU has the responsibility of designating the reference vehicle V_R (cf. Figure 3.4). The pose of the latter is used as a mobile reference frame by the formation coordinates system. The RSU can select V_R either from the vehicles part of the MVS or by considering a virtual vehicle. In this thesis work, the reference vehicle V_R is selected from the MVS vehicles traveling in the lane containing the merging zone, and based on the shortest distance w.r.t. merging zone (cf. Figure 3.4). Its position is denoted by x_R, y_R , and its velocity by \mathcal{V}_R .
- Formation modeling approach: The formation of MVS is structured based on the virtual structure approach (cf. Section 2.4.2). Figure 3.6 presents the formation coordinates $F = f_i, i = 1, ..., N$, where f_i are the i th vehicle coordinates w.r.t. the reference frame of the formation.

3.4.3.1 Formation modeling based on a dynamic reference frame

In Figure 3.4, the RSU broadcast V_R pose to the vehicles part of the formation in order to compute their coordinates w.r.t. V_R reference frame. According to [241], two suitable frames for formation definition can be used:

- Cartesian reference frame: also called rigid formation, it aims to maintain the formation's shape, the positions and orientations of the vehicles within the formation, computed relative to V_R , which is also called the leader vehicle, with respect to the Cartesian reference frame (the local frame of the reference vehicle defined as X_m and Y_m). This approach aims to reduce vehicle's dependence on a global reference frame. Applying a straightforward transformation allows the determination of the vehicles part of the formation coordinates within a local reference frame attached to the reference vehicle [154][36].
- Frenet reference frame: also known as flexible formation, this approach is utilized when it is more critical to track the reference vehicle's movements than to maintain a fixed formation shape during navigation. The Frenet reference frame is employed to adapt the formation to the reference vehicle's trajectory. This trajectory is used to define the formation's longitudinal (curvilinear) and lateral (perpendicular to the trajectory) coordinates [241][27][137] (cf. Figure 3.6).

In the Cartesian formation, the reference vehicle path is not taken into account such as in the Frenet formation, only its current Cartesian pose and dynamics have to be known by the vehicles part of the formation. Since the reference vehicle trajectory is part of the Frenet formation modeling, the latter offers a safe formation navigation in static environment. In contrary, according to [241], the Cartesian formation allows a safe stable geometric formation shape.

During the on-ramp merging scenario, the formation shape is reconfigured from one initial shape to another, while guaranteeing the safety of the reconfiguration. Based on the advantages of both of the formation modeling approaches, the Frenet modeling approach is the one that satisfies the scenario requirements. In fact, one of the particularities of the on-road environment is its high dynamic, especially for highway environment. Additionally, the Frenet modeling approach offers more flexibility, an important requirement for navigation in dynamic environments.

3.4.3.2 Formation modeling based on the Frenet reference frame

Each vehicle part of the formation is characterized by its coordinates $X_i = [x_i, y_i]^T$ within the global reference frame $[X_G, Y_G]$ (cf. Figure 3.6). In order to locate X_i w.r.t. the reference vehicle V_R (cf. Figure 3.6), it is proposed to use a Frenet reference frame (cf. Section 3.4.3.1) centered on V_R (cf. Figure 3.6).



Figure 3.6. Coordinates system based on the Frenet reference frame

According to V_R pose X_R and its reference trajectory (cf. Figure 3.6), the coordinates system is described as follows:

- The *i*-vehicle coordinates part of the formation are defined as $f_i = [h_i, l_i]^T, i \in N$ (cf. Figure 3.6).
- $-h_i$ and l_i are the longitudinal and lateral vehicle's coordinates w.r.t. the Frenet reference frame, respectively. The longitudinal coordinate

 h_i represents the curvilinear distance between V_R and X_i according to the tangent to V_R trajectory, while the lateral coordinate l_i is computed based on the perpendicular line between $X_R(h_i)$ and its reference trajectory, and that passes through X_i , $l_i = PerpDistX_R(h_i), X_i$ (cf. Figure 3.6).

The kinematic model in eq. 3.1 is written in the global reference frame X_G, Y_G , thus a transformation from the mobile reference frame to the global reference frame is obtained with the following equations:

$$\begin{bmatrix} x_{T_i} \\ y_{T_i} \end{bmatrix} = \begin{bmatrix} x_R(h_i) \\ y_R(h_i) \end{bmatrix} + \begin{bmatrix} -l_i \sin(\theta_R(h_i)) \\ l_i \cos(\theta_R(h_i)) \end{bmatrix}$$
(3.8)

with $[x_{T_i}, y_{T_i}]^T$ is the i-th vehicle virtual target w.r.t. V_R 's mobile reference frame. $[x_R, y_R]^T$ and θ_R are the reference vehicle V_R pose and orientation.

3.5 Conclusion

This chapter presented the design of the Cooperative Multi-Controller Architecture (C-MCA) for safe and energy efficient on-ramp merging on highway, in addition to the details given on the main modules, as well as, their interactions. Indeed, the C-MCA aims to satisfy both the MVS overarching goal and the vehicle's individual goals. With the help of the multi-behavior decision-making level part of the architecture, the nominal behavior is activated when it is safe to satisfy the vehicle's individual goals. Satisfying the vehicle's goals is not always possible, especially given the shared nature of the merging zone. Consequently, the C-MCA has a dedicated cooperative behavior that works along the negotiation protocol part of the decisionmaking level to solve conflicting merging scenarios. In fact, a safe passing sequence is selected by the decision-making level, and the latter is translated into the vehicle's dynamics by the formation control approach part of the cooperative behavior. The following chapter is dedicated to present the details related to the proposed formation control strategies.

CHAPTER **4** Dynamic Formation Reconfiguration for On-Ramp Merging

Abstract

In scenarios involving conflicting mergings, the decision-making component of the C-MCA is tasked with determining a secure passing sequence for the MVS within the merging zone. The passing sequence is subsequently translated into the MVS dynamics in the planning level through the application of the formation reconfiguration mechanism. Within this chapter, several dynamic formation reconfiguration approaches for on-ramp merging will be discussed. Each proposed formation reconfiguration will be outlined, and its performance will be assessed to identify any limitations. These identified limitations will then guide the refinement of the formation reconfiguration mechanism, iteratively improving it until the planning objectives of the C-MCA are satisfactorily met¹.

Contents

4.1	Introd	uction	92		
4.2	Constrained Optimal Reconfiguration Matrix (CORM)				
	4.2.1	The inter-target distance matrix	95		
	4.2.2	Constrained inter-target distance matrix	97		
		4.2.2.1 Constrained optimal reconfiguration matrix	98		
		4.2.2.2 Safe and feasible local trajectory planning .	99		
	4.2.3	Simulation results	103		

¹The detailed contributions in this chapter have been the subject of the following publications ITSC'22, VAMS'23, MMAR'23 and ITSC'23.

	4.2.4	Conclusion on the CORM algorithm	109				
4.3	Exten	Extended Dynamic Reconfiguration Matrix (E-CORM)					
	4.3.1	Extended inter-target distance matrix	110				
		4.3.1.1 Reconfiguration gains	112				
		4.3.1.2 Stability analysis w.r.t. the convergence error	s112				
	4.3.2	Trajectory segmentation strategy	113				
	4.3.3	Simulation results	115				
	4.3.4	Conclusion of the E-CORM algorithm	117				
4.4	Forma	ation Reconfiguration Approach based on an Online					
	Contr	ol Strategy (FRA-OCS)	119				
	4.4.1	FRA-OCS formalism	120				
	4.4.2	Online reconfiguration gains identification and veloc-					
		ity profile generation	122				
	4.4.3	Simulation results	125				
	4.4.4	Conclusion of the FRA-OCS algorithm	128				
4.5	Chapt	er conclusion	129				

4.1 Introduction

The key concept behind the proposed dynamic formation reconfiguration approach is to take advantage of the MVS coordination ability to achieve safely and smoothly the merging scenario. In other terms, dynamic formation reconfiguration is used to control the vehicles motion during the merging phase.

Once the $N \in \mathbb{N}$ vehicles part of the formation are identified, the Frenet frame (cf. Section 3.4.3.1) is used to model the initial shape F^{init} . The formation desired shape F^{end} (cf. Figure 4.1) is obtained through the translation of the passing sequence sq selected by the decision-making level. Lastly, in order to perform the merging maneuver, the formation reconfiguration process is launched.

Formation reconfiguration is the process that reshapes the formation from its initial shape F^{init} toward the final one F^{end} by generating intermediary formation coordinates F(t) (cf. Figure 4.1). The intermediary coordinates are transformed from the Frenet frame to the global Cartesian frame to generate the vehicles virtual targets T_d . The respect of the merging safety is ensured with the help of the inter-vehicle target distances. In other terms, each vehicle part of the MVS has its own virtual target T_d generated by the formation reconfiguration process. The distance between two targets is



Figure 4.1. The virtual structure approach used to model the formation and its reconfiguration to perform the merging maneuver. (a) The initial shape of the formation and its coordinates. (b) The final shape of the formation after the merging maneuver and its desired coordinates.

the inter-vehicle target distances. Consequently, the MVS safety during the merging is ensured by the generation of safe virtual targets. The system of equations that describes the formation reconfiguration are given below:

$$\begin{cases}
F^{init} = [f_1^{init^T}, ..., f_N^{init^T}]^T, \\
F^{end} = [f_1^{end^T}, ..., f_N^{end^T}]^T, \\
F(t) = [f_1(t)^T, ..., f_N(t)^T]^T,
\end{cases}$$
(4.1)

 $f_i^{init}, f_i^{end}, i \in N$ are the coordinates of V_i in the initial and final formation, while $f_i(t), i \in N$ are its instantaneous coordinates.

 $e_{f_i} = [e_{h_i}, e_{l_i}]^T$ is the convergence error between the desired coordinates of V_i in the formation and the actual ones, it can be defined as:

$$\begin{cases} e_{f_i} = f_i^{end} - f_i(t), \\ f_i(t) = [h_i(t), l_i(t)]^T, \\ f_i^{end} = [h_i^{end}, l_i^{end}]^T, \end{cases}$$
(4.2)

The global error for a formation composed of N vehicles can be written as:

$$e_F = F^{end} - F(t) \tag{4.3}$$

The derivative expression of the error can be written as:

$$\dot{e}_F = g(e_{f_1}, \dots, e_{f_N}) \tag{4.4}$$

Equation 4.4 describes the formation reconfiguration problem as a control problem, where the goal is to control the convergence error from its initial value (error with the initial formation shape) toward the final one (error with the desired shape, so zero), while ensuring the respect of the vehicles' safety.

Part of this research work, several formation reconfiguration approaches were proposed. The following sections aim to describe these approaches. In Section 4.2, the Constrained Optimal Reconfiguration Matrix (CORM) is described along with its advantages and limitations. Based on the CORM's limitations, the Extended Constrained Optimal Reconfiguration Matrix (E-CORM) was proposed (cf. Section 4.3). Lastly, in order to overcome both the CORM's and the E-CORM's limitations, the Formation Reconfiguration Approach based on an Online Control Strategy (FRA-OCS) is proposed and presented in Section 4.4. The performance of each of the proposed formation reconfiguration approaches is discussed with the help of several simulation results.

4.2 Constrained Optimal Reconfiguration Matrix (CORM)

The proposed Constrained Optimal Reconfiguration Matrix (CORM) involves an analysis of the derivative of e_F (cf. eq. 4.4). This analysis aims to ensure convergence to the new formation shape and generates smooth trajectories using the virtual targets during the reconfiguration process. Additionally, it helps to manage the minimum inter-target distance to prevent collisions between vehicles.

The CORM is originally inspired by the inter-target distance matrix, thus, a concise overview of the latter is presented in the following section. Subsequently, the section focuses on the Constrained inter-target distance matrix proposed to explicitly take into account the road constraints. Lastly, this section delves into the safe and feasible local trajectory planning strategy proposed to ensure the feasibility of the formation reconfiguration during the merging.

4.2.1 The inter-target distance matrix

The proposed function \dot{e}_F for the entire formation is adapted from [238][241], and it is designed as a linear system:

$$\dot{e}_F = A e_F \tag{4.5}$$

where $e_F = [e_{f_1}^T, ..., e_{f_N}^T]^T$ and $A^{2N \times 2N}$ are the state vector and the intertarget distance matrix corresponding to a formation of N vehicles, respectively.

$$A = \begin{bmatrix} a_1 I^{2 \times 2} & a_{12} I^{2 \times 2} & \cdots & a_{1N} I^{2 \times 2} \\ -a_{12} I^{2 \times 2} & a_2 I^{2 \times 2} & \cdots & a_{2N} I^{2 \times 2} \\ \vdots & \vdots & \ddots & \vdots \\ -a_{1N} I^{2 \times 2} & -a_{2N} I^{2 \times 2} & \cdots & a_N I^{2 \times 2} \end{bmatrix}$$
(4.6)

The gains a_i on the diagonal with $i \in N$ control the convergence rate of the error, while a_{ij} with $i \neq j | \forall i, j \in \{N \times N\}$ are related to the inter-target distance between T_{d_i} and T_{d_j} , to ensure the convergence of the formation toward its desired shape.

The stability of the formation error system can be straightforward proved using Lyapunov analysis with the Lyapunov candidate function:

$$V = \frac{1}{2}e_F^T e_F \tag{4.7}$$

V is a positive-definite function. To guarantee the stability of the system, \dot{V} must be negative-definite. By taking the derivative of eq. 4.7 and using eq. 4.5, \dot{V} can be written as:

$$\dot{V} = e_F^T \dot{e}_F = e_F^T A e_F \tag{4.8}$$

Since A is negative-definite, then $\dot{V} < 0$ and the formation error system has an asymptotic convergence.

Safety insurance

It is important to highlight that to avoid the collisions between the N vehicles part of the formation, the condition relating the minimum interdistances \underline{D}_T and the gains of the matrix A must be satisfied. Indeed, the distance between the defined target set-points must be greater than a certain constant minimum distance, called \underline{D}_T . The latter is chosen to take into account the dimensions of the vehicle in addition to a certain offset (cf. Figure 4.2).



Figure 4.2. Illustration of the merging scenario where the minimum inter-distance \underline{D}_T is depicted

For example, the case of two targets T_{d_2} and T_{d_3} is analyzed. The intertarget distance can be computed as:

$$d_T^2 = e_{f_{23}}^T e_{f_{23}} \tag{4.9}$$

Taking the derivative of the last equation to obtain the minimum value:

$$2e_{f_{23}}^T \dot{e}_{f_{23}} = 0 \tag{4.10}$$

After several developments (the details can be found in Appendix B), it is obtained finally in order to respect the minimum inter-target \underline{D}_T [241]:

$$e_{f_{23}}^{T} \left[\frac{a_2 - a_3 + a_{23}}{a_3 - a_{23}} e_{f_2} + e_{f_{23}}^{end} \right] \ge \underline{D}_T^2 \tag{4.11}$$

Thus, the values of a_2 , a_3 , and a_{23} must be chosen to satisfy eq. 4.11.

The inter-target distance matrix proposed in [241] is designed for an openworld environment, where a reactive collision avoidance approach was used only against the other moving robots present in the environment. Nevertheless, in the proposed CORM, it is targeted to deal with structured on-road environment, with road borders (cf. Figure 4.1). It is thus important that the proposed formation reconfiguration takes into account these constraints. In order to take explicitly into account these road constraints while reconfigurating the MVS formation during the merging maneuver the constrained inter-target distance matrix is proposed. The details of the proposed approach are given in the following section.

4.2.2 Constrained inter-target distance matrix

In on-road environment, the vehicles travel in a constrained environment, imposed by the road borders and the road geometry. As stated in Section 4.2.1, the inter-target distance matrix proposed in [241] do not takes environments with such constraints into account. In order to mitigate these constraints, it is proposed a two-steps reconfiguration matrix computation.

First, through an optimization algorithm (cf. Section 4.2.2.1), it is proposed to compute a constrained inter-target distance matrix. Consequently, the virtual targets T_d of the N vehicles part of the formation can be generated. This ensures the reconfiguration of the formation from the initial shape to the desired one, through the merging maneuver (cf. Figure 4.1), while guaranteeing the respect of the safety distance between the vehicles. The geometry of the road (i.e., the road borders and the road center-line) is taken into account at this level with the help of an optimization process based on the objective function given by eq. 4.12. In Section 4.2.2.1, the details of this first step are presented.

The constrained optimization in the first step allows to generate \mathcal{M} (cf. Figure 4.3); an approximation of the global reference trajectory w.r.t. the objective function given in eq. 4.12. However, the targets T_d are not guaranteed to be onto the reference path (cf. Figure 4.3), thus, in the second step of the proposed formation reconfiguration approach, it is proposed to project these generated virtual targets w.r.t. the reference path. The projection uses a Frenet reference frame w.r.t. the reference path to obtain the projected targets T_p . The dynamic of the projected targets T_p is similar to the one of T_d , which means that if the vehicles follow precisely T_p , the vehicles stay on their reference path. However, their in-between distance profile will not be the same as if they follow T_d . Thus, it is proposed to impose a new dynamic to T_p to obtain \overline{T}_p . The latter makes sure that the vehicles are at the same safety distance as the one obtained with the constrained inter-target matrix when they enter the conflicting zone (cf. Figure 4.3). Section 4.2.2.2 gives the details of the computation of the projected target and its dynamic.

4.2.2.1 Constrained optimal reconfiguration matrix

The objective of the optimization algorithm presented in what follows is to compute the optimal constrained inter-distance matrix w.r.t. the objective function in eq. 4.12.

$$J_{\forall k \in N}^{a_k} = \sum_{k=0}^{T} \left[w_i \left[\frac{PerpDist\{T_{d_i}(k), T_{p_i}(k)\}}{PerpDist\{T_{p_i}(k), Border\}} \right]^2 + w_j \left[\frac{PerpDist\{T_{d_j}(k), T_{p_j}(k))}{PerpDist\{T_{p_j}(k), Border\}} \right]^2 \right]$$
(4.12)

Equation 4.12 is composed of two terms: The first term takes into account the i-th vehicle, it aims to minimize distance between V_i 's reference path and T_{d_i} (cf. Figure 4.3), while the second term related to the j-th vehicle, considers the minimization of distance between T_{d_j} and its reference path (cf. Figure 4.3). w_i and w_j with $w \in \mathbb{R}^+$, correspond to the optimization weights balance between the two sub-criteria, the weight related to the merging vehicle is higher in order to give the latter a sufficient flexibility w.r.t. to the vehicles already on the main line. For $i \neq j, \forall i, j \in N, T_{d_{i,j}}$ and $T_{p_{i,j}}$ are the targets obtained with constrained inter-target distance matrix and their projected points respectively (cf. Figure 4.3). $PerpDist\{T_{d_{i,j}}(k), T_{p_{i,j}}(k)\}$ are the perpendicular distances between T_d and its projected target T_p w.r.t. to the reference path, while $PerpDist\{T_{p_{i,j}}(k), Border\}$ are the perpendicular distances between the projected target and the road border, used to normalize the objective function (cf. Figure 4.3).

In Figure 4.4, the details of the optimization process are illustrated. Following eq. 4.12, through the optimization algorithm based on gradient descent, it is aimed to compute the diagonal gains of the reconfiguration matrix A. The objective function is non-linear, the optimization algorithm (cf. Algorithm 1, inputs) needs the optimization boundaries $[a_{i,j_{min}}, a_{i,j_{max}}]^T$ and the starting point $[a_{i_0}, a_{j_0}]^T$ with $i \neq j, \forall i, j \in N$. The anti-diagonal gains of A, in charge of the inter-target distances, are computed while using eq. 4.11.

The full merging scenario is launched with a constant reconfiguration matrix A. The data related to the merging scenario are used to compute the objective function in eq. 4.12 (cf. Algorithm 1). The latter is used to judge on the stop criterion; when the minimum of the cost function is reached. At last, the optimization algorithm returns the optimal values of the gains



Figure 4.3. The projection of T_{d_i} w.r.t. the reference trajectory

 a_i^* and a_j^* , $i \neq j$, $\forall i, j \in N$, the gains $a_{ij}, i \neq j, i, j \in N$ are computed with the help of eq. 4.11.

Algorithm 1: Optimization Algorithm					
I	nput : GlobalInput Scenario data				
	$[a_{i_{min}}, a_{i_{max}}]^T$ boundaries of the gain a_i				
	$[a_{j_{min}}, a_{j_{max}}]^T$ boundaries of the gain a_j				
	$[a_{i_0}, a_{j_0}]^T$ start point of the optimization				
	ObjectiveFunction the objective function in eq. 4.12				
C	\mathbf{Putput} : $[a_i^*, a_j^*]^T$ optimal values of the gains a_i and a_j				
1 A	$IergingData \leftarrow [PerpDist\{T_{d_i}, T_{p_i}\}, PerpDist\{T_{d_j}, T_{p_j}\},$				
	$PerpDist\{T_{p_i}, Border\}, PerpDist\{T_{p_i}, Border\}\}^T$				
2 W	2 while $(StopCriterion \neq True)$ do				
3	$[a_i, a_j]^T \leftarrow Fixe([a_{i_0}, a_{j_0}]^T, [a_{imin}, a_{imax}]^T, [a_{jmin}, a_{jmax}]^T)$				
4	$MergingData \leftarrow MergingScenario(GlobalInput, a_i, a_j)$				
5	$T \leftarrow length(MergingData)$				
6	forall $t \in [0,T]$ do				
7	$ Cost(t) \leftarrow ObjectiveFunction(MergingData(t)) $				
8	$_GlobalCost \leftarrow Sum(Cost)$				

4.2.2.2 Safe and feasible local trajectory planning

In this section it is proposed to delve into the projection approach based on a Frenet reference frame that ensures the respect of the global reference path imposed by the global reference planner. In addition, this section



Figure 4.4. The proposed CORM (Constrained Optimal Reconfiguration Matrix) flowchart. ① The optimization algorithm. ② The projection and mean velocity computation

presents the computation of the velocity profile imposed to the projected target T_p to ensure the safety, feasibility, and smoothness of the merging

maneuver.

1. Global reference aware target: The constrained optimal intertarget distance matrix allows to respect the road constraints while ensuring the vehicle's safety. However, the generated targets T_d are not guaranteed to be onto the reference path (cf. Figure 4.3 ①). In order to ensure this requirement, it is proposed to use the reference path as a guiding system for the vehicle and compute its effective target T_p w.r.t. this latter (cf. Figure 4.3 ①).

Each target T_d of the merging vehicle is projected w.r.t. the reference merging path using a Frenet reference frame $[X_f, Y_f]$ (cf. Figure 4.3) to obtain T_p . The lines 10 and 11 in Algorithm 2 and eq. 3.8 details the transformation from the mobile reference frame centered on V_R to the global reference frame $[X_G, Y_G]$, in addition to the projection function that uses the reference merging path and $T_d \in \mathcal{M}$ to obtain T_p (cf. Figure 4.3 (D)

2. Safe and feasible velocity profile: The projected target T_p has a similar dynamic as T_d . However, in order to draw full advantage from the constrained optimal inter-target distance matrix in terms of safety formal insurance, it is proposed to use the latter to compute the necessary mean velocity that must be imposed to the vehicles, such that they enter the conflicting zone in Figure 4.3 ② at the same moment as if they have followed T_d .

Before the presentation of the details related to the imposed dynamics, for the clarity and the understanding of the reader, it is proposed to define the conflicting zone. The conflicting zone in Figure 4.3 (2) defines the area where a collision between the merging vehicle and the vehicle on the highway vehicle may occur. $\mathcal{P}_{merging}$ defines the position of the merging vehicle where the surrounding circles of V_i and V_j may overlap, resulting in a collision. The points A, B, C, and D define the limits of the conflicting zone. A is the pose of the merging vehicle where a collision may occur, B is related to the limits of the merging zone, D defines the end of the merging zone, while C is its projection w.r.t. the highway center-line (cf. Figure 4.3 (2)).

Based on the definition of the conflicting zone, it is proposed to compute the mean velocity $\bar{\mathcal{V}}_{i,j}$ where i, j are the indices of the vehicles V_i and V_j respectively (cf. Figure 4.3 (2)), such that the in-between distance respects the following:

$$EucDist\{\bar{T}_{p_j}(t_{end}), \bar{T}_{p_i}(t_{end})\} = EucDist\{T_{d_j}(t_{end}), T_{d_i}(t_{end})\}$$
(4.13)

Algorithm 2: Projections and computation of the imposed dynamic

Input : GlobalInput Scenario data a_i^*, a_j^*, a_{ij}^* optimal gains of the reconfiguration matrix (cf. Algorithm 1) Output : $\mathcal{V}_{i,j}$ the mean velocity of the vehicles V_i and V_j 1 $k \leftarrow 1$ 2 $\mathcal{F}(k) \leftarrow F^{init}$ **3** $\varepsilon(k) \leftarrow \mathcal{F}^{end} - \mathcal{F}(k)$ 4 $Buffer_R \leftarrow ReferenceTrajectory(V_R)$ 5 $Buffer_{i,j} \leftarrow ReferenceTrajectory(V_i, V_j)$ 6 while $(\varepsilon(k) \neq 0)$ do $k \leftarrow k+1$ 7 $\mathcal{F}(k) \leftarrow DynamicReconfiguration(\mathcal{F}^{end}, \mathcal{F}(k-1))$ 8 $\varepsilon(k) \leftarrow \mathcal{F}^{end} - \mathcal{F}(k)$ 9 $T_{d_{i,i}}(k) \leftarrow Transform(Buffer_R, \mathcal{F}(k), X_{V_R})$ 10 $T_{p_{i,j}}(k) \leftarrow Projection(Buffer_{i,j}, T_{d_{i,j}})$ 11 $X_{V_{i,j}}(k) \leftarrow Control(X_{V_{i,j}}(k-1), T_{p_{i,j}}(k))$ 12 13 $\mathcal{V}_{i,j} = \frac{CurviDist\{V_{i,j}(1,k)\}}{t_{end}-t_{init}}$

> where t_{end} is the time when the pose $\mathcal{P}_{merging}$ (cf. Figure 4.3) is reached by the merging vehicle. $EucDist\{T_{d_j}(t_{end}), T_{d_i}(t_{end})\}$ represents the Euclidean distance between the targets T_{d_i} and T_{d_j} generated during the first step of the proposed CORM for the vehicles V_i and V_j , while $EucDist\{\bar{T}_{p_j}(t_{end}), \bar{T}_{p_i}(t_{end})\}$ (Figure 4.3 ②) is the in-between Euclidean distance between the targets with the imposed dynamic $\bar{\mathcal{V}}_i$ and $\bar{\mathcal{V}}_j$ for the vehicles V_i and V_j , respectively.

> The mean velocity $\bar{\mathcal{V}}_{i,j}$ where i, j are the indices of the vehicles V_i and V_j (Figure 4.3 (2)) is computed based on the line 13 in Algorithm 2, where t_{init} is the corresponding time when the reconfiguration was launched. A curvilinear distance formula is used to get the traveled distance by each vehicle between t_{init} and t_{end} .

> In order to have the smoothest possible behavior w.r.t. the vehicles' dynamics, it is proposed to use a sigmoid function to shape the velocity profile. This latter is used to create a velocity profile that goes from the vehicle initial velocity toward the mean velocity, and goes to the reference vehicle V_R velocity when the vehicle enters the main line. This choice is motivated by the sigmoid ability to smoothly control the convergence rate from an initial velocity to the final desired one. In other terms, using the sigmoid function permits us to impose a

feasible and comfortable acceleration and deceleration profile to the vehicles.

4.2.3 Simulation results

In order to evaluate the efficiency of the CORM in terms of its ability to guarantee the safety requirement as well as the smoothness of the vehicle's dynamics, while performing the on-ramp merging maneuver, a first scenario is proposed. It aims to perform the merging maneuver with a formation of three vehicles. The simulation video can be found in https://youtu.be/UM2cLt74pVM. Then, a comprehensive summary of several conducted simulations is presented in Table 4.2.

1. On-ramp merging in formation:



Figure 4.5. Evolution of the formation shape during the merging performed by the CORM framework (Simulation video: https://youtu.be/UM2cLt74pVM)

In the following simulation, it is aimed to perform a merging maneuver with a formation of three vehicles. The considered merging scenario is an on-ramp merging with an incidence angle $\mu = 10^{\circ}$, V_1 (i.e., the reference vehicle V_R) is placed in the main line, so as V_2 . The third vehicle V_3 initially placed in the secondary on-ramp merging road (cf. Figure 4.5). The initial formation shape is triangular (cf. Figure 4.5), consequently at the end of the reconfiguration phase the aim is to put the vehicles part of the formation in a linear shape to form a platoon (cf. Figure 4.5). Table 4.1 resumes the scenario inputs.

T	V-l
inputs	values
Initial formation goordinates [m]	(0 - 40 - 50)
mitial formation coordinates $[m]$	$\begin{pmatrix} 0 & 0 & 9.2 \end{pmatrix}$
Final formation coordinates $[m]$	(0 - 80 - 40)
Final formation coordinates [m]	
$\mathcal{V}_{1,2,3}[m/s]$	$\left[19.4, 19.4, 19.4 ight]$
$[a_{2min}, a_{2max}]^T$	[-0.4, 0.1]
$[a_{3min}, a_{3max}]^T$	[-0.4, -0.05]
$[a_{2_0}, a_{3_0}]^T$	[-0.25, -0.25]
$[w_2, w_3]^T$	[1, 4]
$[a_2^*, a_3^*]^T$	[-0.1228, -0.365]
$\underline{D}_T[m]$	12

Table 4.1. The values of the inputs of the CORM algorithm

The minimum inter-target distance \underline{D}_T (cf. Figure 4.2) is computed using the following model:

$$\underline{D}_T = (R_i + R_j) + offset \tag{4.14}$$

where R_i and R_j are the radius of the circles that surround the vehicles V_i and V_j , and offset is the safety distance to avoid rear-end collisions. The reconfiguration of the triangular shape toward the desired linear one is illustrated in Figure 4.5. The formation defined by its initial coordinates passes through the reconfiguration phase, where V_3 is in front of V_2 as desired w.r.t. the formation's final coordinates, consequently, the $EucDist(V_1, V_2)$ needs to increase. The intermediate formation showcases the pose of the three vehicles in the conflict zone. As expected $EucDist(V_1, V_2)$ has increased to make space for V_3 in the convoy. In the second phase, after the convergence of the formation toward its desired shape, the vehicles form a linear shape, where the desired in-between distances are respected.

To evaluate the CORM capability in terms of safety and convergence errors, it is proposed to study the evolution of the formation coordinates and the in-between distances profiles illustrated in Figure 4.6. A first order asymptotic convergence from the initial formation shape toward the desired one can be noticed in the formation coordinates plots. As for the minimum distances, the in-between distances profiles are always greater than \underline{D}_T . The in-between distances when the vehicles are in the conflicting zone is greater than 20m. The final inbetween distances meet the safety requirement for a convoy formation; $EucDist(V_i, V_j) = 2[s] \times \mathcal{V}_{i,j}[m/s] = 40m.$

In Figure 4.7, the linear velocity, the longitudinal, and the lateral accelerations are presented for each of the three vehicles part of the



Figure 4.6. The formation coordinates and the Euclidean in-between distances profiles

formation. As expected, the velocity of the vehicle V_2 decreases at the beginning of the scenario to make space for V_3 , before it increases to make sure that V_2 meets the velocity requirement in the platoon. As for V_3 , its velocity increases at the beginning to enter the highway center-line while respecting the safety distances. A decreasing can be noticed around 4s to follow the convoy velocity (fixed to 19.4m/s - 70km/h). Figure 4.7 confirms that the longitudinal and the lateral accelerations respect the maximum and minimum authorized accelerations (i.e., $-4m/s^2$ for deceleration and $3m/s^2$ for acceleration). Thus, with the help of the results in Figure 4.7, the vehicles' dynamics during the formation reconfiguration while the merging is performed are smooth and comfortable.

2. Influence of the projection phase on the CORM efficiency:

This section is dedicated to validate more intensively the proposed approach, while emphasizing mainly the ability of the projection phase for different environments structure (several values of μ).



Figure 4.7. The vehicles' velocity and acceleration profiles

For a range of incidence angles between $\mu = 10^{\circ}$ and $\mu = 30^{\circ}$ with $\Delta \mu = 10^{\circ}$ step each time. It is proposed to study the performance of the CORM for velocity between 5m/s and 15m/s with an increase of 5m/s each time. For all the simulations, the inputs of the CORM algorithm are the same as in Table 4.1. The conducted simulations are summarized in Table 4.2.

According to the performed simulations, it is important to emphasize that the safety of the vehicles is always ensured; the minimum intervehicle distance in the conflicting zone \underline{D}_T is always greater than 20m. The metric $Error_{max}$ is the maximum distance between T_d and \overline{T}_p . This latter is lower than 1.5m (distance between the center-line and the road border), in other terms, T_d is never out of the road borders. To evaluate the smoothness of the merging, it is proposed to discuss the obtained velocity and acceleration profiles w.r.t. variable velocity of the reference velocity and different incidence angles. The velocity profiles are similar to the one in Figure 4.7, with a sigmoid shape that makes \mathcal{V}_2 decrease toward $\underline{\mathcal{V}}_2$ to make space for V_3 , and \mathcal{V}_3 increase toward $\overline{\mathcal{V}}_3$ in the beginning. Then, \mathcal{V}_2 and \mathcal{V}_3 increases and decreases respectively toward \mathcal{V}_R to form the final desired platoon. As for the acceleration, the lateral and the longitudinal behavior respect the limits of feasibility and comfort (i.e., $-4m/s^2$ for the deceleration and $3m/s^2$ for the acceleration).

$\mu[\text{deg}]$	10			20			30		
$\mathcal{V}_R[m/s]$	5	10	15	5	10	15	5	10	15
$[\underline{\mathcal{V}}_2, \overline{\mathcal{V}}_2][m/s]$	[4.36, 5]	[8.73, 10]	[13.45, 15]	[4.21, 5]	[8.49,10]	[13.26, 15]	[4.13, 5]	[8.35,10]	[13.06, 15]
$[\underline{\mathcal{V}}_3, \overline{\mathcal{V}}_3][m/s]$	[5,5.59]	[10, 11.125]	[15, 16.80]	[5, 5.824]	[10,11.6316]	[15,17.43]	[5, 6.10]	[10,12.10]	[15, 18.14]
$[\underline{a}_2, \overline{a}_2]_{long}[m/s^2]$	[-0.29, 0.15]	[-0.60, 0.29]	[-0.70, 0.35]]	[-0.35, 0.18]	[-0.68, 0.34]	[-0.78, 0.53]	[-0.38, 0.19]	[0.70, 0.37]	[-0.87, 0.44]
$[\underline{a}_2, \overline{a}_2]_{lat}[m/s^2]$	[0,0]	[0,0]	[0,0]	[0,0]	[0,0]	[0,0]	[0,0]	[0,0]	[0,0]
$[\underline{a}_3, \overline{a}_3]_{long}[m/s^2]$	[-0.76, 1.52]	[-1.66, 2.02]	[-1.14, 2.27]	[-0.570, 1.07]	[-1.66, 2.02]	[-1.54, 2.92]	[-0.65, 1.39]	[-1.31, 2.55]	[-2.00, 2.75]
$[\underline{a}_3, \overline{a}_3]_{lat}[m/s^2]$	[-0.29, 0.74]	[-0.39, 0.80]	[-0,74,1.38]	[-0.84, 0.75]	[-1.02,0.96]	[-1.60, 1.59]	[-1.56, 1.55]	[-2.23, 1.94]	[-3.63, 2.53]
$\underline{D}[m]$	21.66	21.80	23.30	24.75	24.01	23.30	28.70	27.30	25.725
$Error_{max}[m]$	0.52	0.73	0.78	0.85	0.88	0.92	1.10	1.13	1.14

Table 4.2. The summary results of the conducted simulation for a variable incidence angle μ and vehicles' velocity \mathcal{V}

4.2.4 Conclusion on the CORM algorithm

This section extensively explored the topic of cooperative formation reconfiguration, with the help of the proposed Constrained Optimal Reconfiguration Matrix (CORM) framework. The CORM framework can be succinctly described as a two-step process:

- 1. First, it employs an optimization algorithm that explicitly accounts for environmental constraint. This step focuses on the computation of the convergence rate necessary for the constrained inter-target distance matrix A. This matrix is responsible for reshaping the virtual structure from its initial configuration to the desired shape required for executing the on-ramp merging maneuver.
- 2. The second step is based on a projection approach with safe and appropriate dynamics. It ensures that each vehicle within the formation, participating in the merging scenario, effectively tracks the dynamic target while aligning with the global reference frame. In this step, the targets determined in step (1) are projected onto a Frenet reference frame linked to the global reference path. This ensures alignment with the global path and a smooth velocity profile is generated using a sigmoid shape, guaranteeing a smooth merging maneuver.

The evaluation of the CORM algorithm was carried out in a simulation environment. It showcased the algorithm's ability to adhere to safety criterion, even in challenging scenarios involving high-speed merging. The algorithm's suitability for varying incidence angles μ of the merging road and diverse formation dynamics was demonstrated through extensive simulations.

However, it is crucial to acknowledge the limitations of the CORM algorithm. One of its key limitations is that it assumes the same dynamics for both longitudinal and lateral reconfiguration. In other words, it employs identical control gains for both longitudinal and lateral coordinates reconfiguration during the merging. Additionally, the generated trajectory with the constrained inter-target distance matrix does not seamlessly align with the imposed reference trajectory, particularly when reconfiguring the position of the merging vehicle from a secondary road location to the highway center-line.

Taking these limitations into account and in the optic of mitigating them, the following section presents the Extended Dynamic Reconfiguration Matrix (E-CORM).

4.3 Extended Dynamic Reconfiguration Matrix (E-CORM)

As stated in Section 4.2.4, the proposed CORM framework uses a restricted motion convergence approach which limits its flexibility. Taking this limitation into account, the Extended Constrained Optimal Reconfiguration Matrix (E-CORM) proposes to decorrelate the dynamic of the longitudinal and the lateral coordinates, allowing more flexibility; in addition to a road segmentation approach that permits to respect the global path imposed by the global trajectory planning level.

Before the presentation of the details of the proposed E-CORM for formation reconfiguration, and for the clarity of this PhD manuscript and the understanding of the reader, it is important to note that the formation definition and configuration formalism used for the E-CORM is the same as the one used for the CORM (cf. Section 3.4.3).

The following section is dedicated to the presentation of the details of the proposed E-CORM. Thus, in Section 4.3.1, the extended inter-target distance matrix is presented. Section 4.3.2 details the toad segmentation approach used part of the E-CORM. In Section 4.3.3, the simulations results related to the E-CORM performance are discussed. Lastly, Section 4.3.4 concludes on E-CORM's advantages and limitations.

4.3.1 Extended inter-target distance matrix

In order to characterize the evolution of the reconfiguration from the initial shape toward the desired one, while ensuring the respect of the safe intertarget distance \underline{D}_T between the N vehicles part of the formation, the E-CORM proposes an intermediate state vector S given as the following:

$$S = \dot{e} + \lambda e + \gamma \int e \, dt \tag{4.15}$$

To overcome the CORM lack of flexibility, the E-CORM uses in addition to the convergence error e (cf. eq. 4.2), the convergence rate \dot{e} of the error and the sum of the errors through the integrative term of the latter. $\lambda, \gamma \in \mathbb{R}^+$ are the convergence gains and permit to offer more flexibility to the E-CORM algorithm.

The convergence of S follows a first order convergence model detailed in eq. 4.16.

$$\dot{S} = A \times S = A\dot{e} + A\lambda e + A\gamma \int e \, dt \tag{4.16}$$

where $A^{2N \times 2N}$ is a negative-definite convergence matrix.

Using eq. (4.2) in eq. (4.16) permits to write the extended system of equations representing the studied system.

$$\dot{S}_{h_{1}} = a_{h_{1}}\dot{e}_{h_{1}} + a_{h_{1}}\lambda_{h_{1}}e_{h_{1}} + a_{h_{1}}\gamma_{h_{1}}\int e_{h_{1}}dt$$

$$\dot{S}_{l_{1}} = a_{l_{1}}\dot{e}_{l_{1}} + a_{l_{1}}\lambda_{l_{1}}e_{l_{1}} + a_{l_{1}}\gamma_{l_{1}}\int e_{l_{1}}dt$$

$$\vdots
\dot{S}_{h_{N}} = a_{h_{N}}\dot{e}_{h_{N}} + a_{h_{N}}\lambda_{h_{N}}e_{h_{N}} + a_{h_{N}}\gamma_{h_{N}}\int e_{h_{N}}dt$$

$$\dot{S}_{l_{N}} = a_{l_{N}}\dot{e}_{l_{N}} + a_{l_{N}}\lambda_{l_{N}}e_{l_{N}} + a_{l_{N}}\gamma_{l_{N}}\int e_{l_{N}}dt$$

where h_i and l_i , representing the longitudinal and the lateral coordinates of the formation, converge toward the target with different convergence rates a_{h_i} and a_{l_i} .

The system in eq. (4.18) presents the matrix form of the studied system.

$$\begin{bmatrix} \dot{S}_{h_1} \\ \dot{S}_{l_1} \\ \dot{S}_{h_2} \\ \dot{S}_{l_2} \\ \vdots \\ \dot{S}_{h_N} \\ \dot{S}_{l_N} \end{bmatrix} = \Omega_1 \begin{bmatrix} \dot{e}_{h_1} \\ \dot{e}_{l_1} \\ \dot{e}_{h_2} \\ \dot{e}_{l_2} \\ \vdots \\ \dot{e}_{h_N} \\ \dot{e}_{l_N} \end{bmatrix} + \Omega_2 \begin{bmatrix} e_{h_1} \\ e_{l_1} \\ e_{h_2} \\ e_{l_2} \\ \vdots \\ e_{h_N} \\ e_{l_N} \end{bmatrix} + \Omega_3 \begin{bmatrix} \int e_{h_1} dt \\ \int e_{l_1} dt \\ \int e_{h_2} dt \\ \vdots \\ \int e_{h_N} dt \\ \int e_{h_N} dt \end{bmatrix}$$
(4.18)

with Ω_1 , Ω_2 and Ω_3 are given in eq. (4.19).

$$\begin{cases} \Omega_{1} = diag[a_{h_{1}}, a_{l_{1}}, \cdots, a_{h_{N}}, a_{l_{N}}], \\ \Omega_{2} = diag[a_{h_{1}}\lambda_{h_{1}}, a_{l_{1}}\lambda_{l_{1}}, \cdots, a_{h_{N}}\lambda_{h_{N}}, a_{l_{N}}\lambda_{l_{N}}], \\ \Omega_{3} = diag[a_{h_{1}}\gamma_{h_{1}}, a_{l_{1}}\gamma_{l_{1}}, \cdots, a_{h_{N}}\gamma_{h_{N}}, a_{l_{N}}\gamma_{l_{N}}], \end{cases}$$
(4.19)

with $\Omega_1^{2N\times 2N},\,\Omega_2^{2N\times 2N}$ and $\Omega_3^{2N\times 2N}$ are the extended reconfiguration matrices.

4.3.1.1 Reconfiguration gains

The convergence matrix used in the CORM algorithm (cf. Section 4.2) assumes that both the longitudinal and the lateral motions converge according to the same convergence rate, which reduces the CORM flexibility. The E-CORM relies on an augmented constrained inter-target matrix presented in eq. 4.19 to overcome the CORM limitations; namely considering geometrical limitation encountered with the CORM to align on the reference path (such as the path given by the global path planning level (cf. Figure 4.11)). In fact, separating the longitudinal convergence from the lateral one adds more flexibility to the E-CORM. The gains a_{h_i} and a_{l_i} are computed using the same non-linear optimization algorithm as the one used for the CORM framework in Section 4.2.

4.3.1.2 Stability analysis w.r.t. the convergence errors

The fundamental stability analysis ensured under the E-CORM is given in this section. The stability of the system in eq. 4.19 is proved in two steps; 1) the stability analysis on the state S in the system in eq. 4.15 using a Lyapunov analysis and; 2) the stability of the convergence error e using the system given in eq. 4.17.

First, it is proposed to define the Lyapunov candidate function:

$$V = \frac{1}{2}S^T S \tag{4.20}$$

V is a positive-definite function. To guarantee the stability of the system, \dot{V} must be negative-definite. By taking the derivative of eq. 4.20 and using eq. 4.16, \dot{V} can be written:

$$\dot{V} = \dot{S}^T S = S^T A^T S \tag{4.21}$$

Since A^T is a diagonal negative-definite matrix, then $\dot{V} < 0$ and the state S converges asymptotically to zero.

The second step of the stability analysis is to prove the convergence of the formation error system given in eq. 4.15. The stability analysis of eq. 4.21 permits to write around the equilibrium point of the state S:

$$S = \dot{S} = 0 \tag{4.22}$$

Taking the systems given in eq. 4.17, eq. 4.22 and the Laplace transformation, the system can be written as in eq. 4.23:

$$E_{h_{1}}p^{2} + \lambda_{h_{1}}E_{h_{1}}p + \gamma_{h_{1}}E_{h_{1}} = 0$$

$$E_{l_{1}}p^{2} + \lambda_{l_{1}}E_{l_{1}}p + \gamma_{l_{1}}E_{l_{1}} = 0$$

$$\vdots$$

$$E_{h_{N}}p^{2} + \lambda_{h_{N}}E_{h_{N}}p + \gamma_{h_{N}}E_{h_{N}} = 0$$

$$E_{l_{N}}p^{2} + \lambda_{l_{N}}E_{l_{N}}p + \gamma_{l_{N}}E_{l_{N}} = 0$$

where p is the Laplace operator. The stability of the system given in eq. 4.23 can be studied as a 2^{nd} order polynomial. According to the Routh-Hurwitz criterion, the system given in eq. 4.23 is stable if all the terms composing the characteristic polynomial have the same sign. Consequently, given that the gains λ_{h,l_i} and γ_{h,l_i} are positive, thus according to Routh-Hurwitz criterion the system of error e given in eq. 4.15 for S = 0 is always stable, and ensures the convergence of e to zero.

4.3.2 Trajectory segmentation strategy



Figure 4.8. Illustration of the on-ramp merging scenario based on the road segmentation strategy

The proposed MVS control architecture relies on the existence of a global reference path that vehicles need to follow to perform their driving maneuver. Consequently, the aim of the optimization algorithm part of the CORM framework was to approximate the reference path when generating the virtual targets of each vehicle of the formation. As discussed in Section 4.2.4, the CORM encounters some difficulties to generate an estimate that aligns with the global reference path. In order to take into account the vehicles reference path and overcome the CORM limitation related to the latter, in this section a segmentation strategy that takes explicitly the reference path is proposed part of the E-CORM algorithm.

Taking advantage from the E-CORM flexibility, it is proposed to segment the merging scenario w.r.t. the vehicle's motion on each segment. Figure 4.8 gives an illustration of the proposed segmentation strategy. Segment Arepresents the portion of the scenario where the vehicles need to track their reference path, thus only the longitudinal reconfiguration of the formation is required. The segment B lies on both the longitudinal and the lateral formation reconfiguration.



Figure 4.9. Coordinates selection based on the traveled segment

Furthermore, the optimization procedure to compute the extended constrained inter-target distance matrix (cf. Section 4.3.1.1) can be expensive in terms of the computation time. Thus, the segmentation of the road topology permits to reduce the computation time by optimizing only the necessary gains part of reconfiguration matrix A. In other terms, when the vehicle travels on the segment A, only the longitudinal reconfiguration is necessary. Consequently, only the gains related to the longitudinal behavior are optimized (cf. Figure 4.9). The lateral coordinates l_i are given according to the road center-line. On the other hand, when both of the longitudinal and lateral behaviors are necessary (i.e., the vehicle travels in segment B), both of the gains a_{h_i} and a_{l_i} are optimized. This selection mode is possible thanks to the capacity of the algorithm to guarantee the continuity of the dynamic targets T_d ensured by the formalism using the state S (cf. equation 4.15).

4.3.3 Simulation results

To assess the effectiveness of the E-CORM algorithm in terms of ensuring the safety and the smoothness of the vehicles' dynamic during the on-ramp merging maneuver, a simulation scenario similar to the one in Section 4.2.3 is proposed. The video of the simulation can be found in https://youtu.be/-h97Qz7K3F4.

In the following simulation, it is aimed to perform a merging maneuver with a formation of three vehicles. The considered merging scenario is an on-ramp merging where V_1 (i.e., the reference vehicle V_R) is placed in the main line, so as V_2 . The third vehicle V_3 (the merging vehicle) is initially placed in the secondary on-ramp merging road (cf. Figure 4.10). The initial formation shape is triangular (cf. Figure 4.10 (a)), consequently at the end of the reconfiguration phase, the aim is to put the vehicles part of the formation in a linear shape to form a platoon (cf. Figure 4.10(c)). The minimum inter-target distance \underline{D}_T is computed using eq. 4.14.

The reconfiguration of the triangular configuration into the desired linear one is visually represented in Figure 4.10. In this process, the initial formation, defined by the original coordinates, undergoes a reconfiguration phase. During this phase, both vehicles V_2 and V_3 are positioned at approximately equi-longitudinal distance relative to V_1 (V_R). The intermediate formation illustrates the positions of all three vehicles within the conflict zone. As anticipated, the gap between V_3 and V_1 decreases, resulting in V_3 merging between V_1 and V_2 (cf. Figure 4.10 (b)). The final formation shape, as depicted in Figure 4.10 (c), shows the three vehicles within the formation now forming a platoon, aligning with the intended configuration.

To evaluate the E-CORM capability in terms of safety and convergence errors, it is proposed to study the evolution of the formation coordinates illustrated in Figure 4.11 and the in-between distances profiles illustrated in Figure 4.12. In Segment A, only the E-CORM's longitudinal reconfiguration is activated, consequently, the lateral coordinates of V_3 are computed with the help of the reference path. In Segment B both the longitudinal and the lateral reconfiguration are activated. As it can be seen in Figure 4.11, the formation reconfiguration is almost complete by the end of Segment B. The Euclidean distances in-between the vehicles are greater than



Figure 4.10. Evolution of the formation shape based on the E-CORM reconfiguration (Simulation video: https://youtu.be/-h97Qz7K3F4)

 \underline{D}_T (cf. Figure 4.12), thus the merging is considered to be safe. When the three vehicles travel in highway, only the longitudinal reconfiguration is activated, thus, the formation continues its convergence toward the desired shape.

In Figure 4.13, the linear velocity is presented for each of the three vehicles part of the formation. As expected, the velocity of the vehicle V_3 increases in-order to reduce the gap between V_1 and V_3 , before it decreases in order to respect the velocity requirement in the platoon. As for V_2 , its velocity increases to converge toward its desired formation longitudinal coordinate, before it increases so that V_2 can travel part of the platoon with constant in-between distance w.r.t. V_3 .



Figure 4.11. The formation's coordinates

4.3.4 Conclusion of the E-CORM algorithm

This section delved extensively into the realm of cooperative formation reconfiguration, employing the proposed Extended Constrained Optimal Reconfiguration Matrix (E-CORM) framework. The primary goal behind introducing E-CORM was to address the CORM's lack of flexibility.

E-CORM leverages an extended inter-target distance matrix, allowing for the reconfiguration of the formation during the merging, while introducing an enhanced level of motion flexibility. Unlike the restrictive nature of CORM, which obstruct vehicles from precisely following their global reference trajectory, and by the addition of the proposed extended inter-target distance matrix, the E-CORM overcomes this limitation by explicitly incorporating the global reference trajectory using the proposed road segmentation strategy.

The road segmentation strategy plays a pivotal role in activating only the necessary vehicle behaviors concerning their position in the environment. Consequently, when vehicles are expected to remain on their reference path, only longitudinal reconfiguration is required. In contrast, when the merging vehicle enters the merging zone, both longitudinal and lateral reconfig-



Figure 4.12. The Euclidean in-between distances

uration are engaged to facilitate the merging process. Importantly, since E-CORM depends on an optimization process to compute the extended inter-target distance matrix gains, these computations are tailored to the activated behaviors, reducing the overall optimization time cost.

The evaluation of the E-CORM algorithm was carried out in a simulation environment. It showcased the algorithm's ability to adhere to safety criterion and its enhanced flexibility even in challenging scenarios involving high-speed merging.

However, it's worth noting that E-CORM's reliance on a non-linear optimization algorithm to compute the extended inter-target distance matrix remains a notable limitation. This holds significance, especially given that one of the aims of this PhD work is to propose an online control architecture. The time-intensive nature of optimization algorithms does not align with the scope of this specific objective. In order to mitigate both of the CORM's and the E-CORM's limitations, in addition of achieving the objective of the decision/control architecture, in the following section the Formation Reconfiguration Approach based on an Online Control Strategies (FRA-OCS) is proposed.



Figure 4.13. The vehicles' velocity profiles

4.4 Formation Reconfiguration Approach based on an Online Control Strategy (FRA-OCS)

As discussed in Section 4.2, the CORM algorithm's lacks of motion flexibility constitute one of its main limitations. Consequently, in Section 4.3, the E-CORM framework was proposed to overcome the CORM limitation. Through the study of the E-CORM performance, it was pointed out that its reliance on an optimization process constitutes a limitation w.r.t. the objective of this work, which is to propose an online control architecture.

To overcome both the CORM's and the E-CORM's limitations, in this section an optimization-free formation reconfiguration algorithm named Formation Reconfiguration Approach based on an Online Control Strategies (FRA-OCS) is presented.

Before the presentation of the details of the proposed FRA-OCS for formation reconfiguration, and for the clarity of this PhD manuscript and the understanding of the reader, it is important to note that the formation definition and configuration formalism used part of the FRA-OCS is the same as the one used for the CORM (cf. Section 3.4.3).

Section 4.4.1 delves into the proposed FRA-OCS formalism, additionally Section 4.4.2 provides the details related to dynamic generator part of the proposed FRA-OCS. The FRA-OCS performance is evaluated through a simulation scenario in Section 4.4.3. Lastly, it is proposed to draw conclusion about the FRA-OCS in Section 4.4.4.

4.4.1 FRA-OCS formalism

Prior to delving into the specifics of the proposed Formation Reconfiguration Approach based on an Online Control Strategy (FRA-OCS), for the clarity and the understanding of the reader, it is important to note that the fundamentals related to the virtual structure formalization used to represent the formation of vehicles can be found in Section 3.4.3.

To address the limited flexibility of both the CORM (cf. Section 4.2) and the E-CORM (cf. Section 4.3) algorithms, the FRA-OCS introduces an intermediate state, S, as an essential component for characterizing the evolution of the reconfiguration process from the initial shape of the formation to its desired shape. By employing this intermediate state vector, the FRA-OCS enables a smooth and controlled transition of the formation towards its desired shape. Eq. 4.24 provides the explicit expression of the intermediate state vector S.

$$S = \dot{e} + \lambda e \tag{4.24}$$

The FRA-OCS utilizes in addition to the convergence error vector e (cf. Section 3.4.3), the convergence rate \dot{e} . By introducing the gain $\lambda \in \mathbb{R}^+$, the FRA-OCS offers greater flexibility in achieving the desired formation reconfiguration.

The use of an optimization approach allows the computation of reconfiguration gains within the CORM algorithm (cf. Section 4.2). By combining the longitudinal and lateral motions, the computation time required for optimization is reduced. However, one drawback of this motion coupling is its limited flexibility. The state S used in the E-CORM (cf. eq. 4.15) is composed of a proportional, a derivative, and an integrative terms. However, in the final form in the FRA-OCS, the state S was chosen to use only the proportional and the derivative terms. This choice is mainly motivated by the need of a simplified mathematical representation to facilitate the online identification of the different gains involved. As a result, in addition to
the state vector S used by the FRA-OCS, the proposed approach suggests to use the road segmentation strategies proposed in Section 4.3.2, decoupling the longitudinal convergence from the lateral convergence. Figure 4.8 illustrates the segmentation approach based on the road geometry used to define the available vehicle motion according to its position.

The convergence of ${\cal S}$ follows a first order convergence model detailed in eq. 4.25.

$$\dot{S} = A \times S = A\dot{e} + A\lambda e \tag{4.25}$$

where $A^{2N \times 2N}$ is a negative-definite convergence matrix.

Using eq. 4.2 in eq. 4.25 permits to write the extended system of equations representing the formation reconfiguration studied system.

$$S_{h_{1}} = a_{h_{1}}\dot{e}_{h_{1}} + a_{h_{1}}\lambda_{h_{1}}e_{h_{1}}$$

$$\dot{S}_{l_{1}} = a_{l_{1}}\dot{e}_{l_{1}} + a_{l_{1}}\lambda_{l_{1}}e_{l_{1}}$$

$$\vdots$$

$$\dot{S}_{h_{N}} = a_{h_{N}}\dot{e}_{h_{N}} + a_{h_{N}}\lambda_{h_{N}}e_{h_{N}}$$

$$\dot{S}_{l_{N}} = a_{l_{N}}\dot{e}_{l_{N}} + a_{l_{N}}\lambda_{l_{N}}e_{l_{N}}$$
(4.26)

where h_i and l_i , representing the longitudinal and the lateral coordinates of the formation, converge toward the target with different convergence rates a_{h_i} and a_{l_i} .

The system in eq. 4.27 presents the matrix form of the studied system.

$$\begin{bmatrix} S_{h_1} \\ \dot{S}_{l_1} \\ \vdots \\ \dot{S}_{h_N} \\ \dot{S}_{l_N} \end{bmatrix} = \Omega_1 \begin{bmatrix} \dot{e}_{h_1} \\ \dot{e}_{l_1} \\ \vdots \\ \dot{e}_{h_N} \\ \dot{e}_{l_N} \end{bmatrix} + \Omega_2 \begin{bmatrix} e_{h_1} \\ e_{l_1} \\ \vdots \\ e_{h_N} \\ e_{l_N} \end{bmatrix}$$
(4.27)

with $\Omega_1 = diag[a_{h_1}, a_{l_1}, \cdots, a_{h_N}, a_{l_N}]$ and $\Omega_2 = diag[a_{h_1}\lambda_{h_1}, a_{l_1}\lambda_{l_1}, \cdots, a_{h_N}\lambda_{h_N}, a_{l_N}\lambda_{l_N}].$

Stability analysis w.r.t. the convergence gains

The stability of the system given in eq. 4.27 is proved in two steps; 1) the stability analysis of the state S in the system in eq. 4.25 using a Lyapunov

analysis and; 2) the stability of the convergence error e using the system given in eq. 4.26. The details of the stability demonstration can be found in Section 4.3.1.2

4.4.2 Online reconfiguration gains identification and velocity profile generation

One limitation of the formation reconfiguration based on both the CORM and the E-CORM is their dependency on an optimization process. In fact, the calculation of optimization-based reconfiguration gains may not be adequate for online calculation, especially, when N vehicles are part of the formation and $2 \times N$ gains (decoupled longitudinal and lateral motions are considered) need to be computed. Considering an on-road environment, in one hand, the approach ability to compute and recompute when necessary the reconfiguration gains is mandatory to guarantee the respect of the safety criterion, explicitly in a highly dynamic environment. In the other hand, due to the dynamic nature of the considered scenario, the approach needs to guarantee the continuity of the vehicles' dynamics during the switch from one configuration to another.

Consequently, the FRA-OCS proposed in this work is designed to overcome the CORM limitation in terms of its real-time computation, with an optimization free procedure to compute the reconfiguration gains. As for the FRA-OCS ability to recompute the convergence gains when necessary, under the FRA-OCS formalism the continuity during the switch from one configuration to another is formally ensured. This section presents the procedure employed to write the system of equation that needs to be solved to reconfigure the formation from the initial shape to the final, while taking into account the initial and the desired dynamics of the vehicles part of the formation.

Using eq. 4.24 in eq. 4.25, the convergence model of the formation reconfiguration error is a system of second order linear differential equations, given in eq. 4.28.

$$\ddot{e}_{h_{1}} + (\lambda_{h_{1}} - a_{h_{1}})\dot{e}_{h_{1}} - a_{h_{1}}\lambda_{h_{1}}e_{h_{1}} = 0 \qquad (4.28)$$

$$\ddot{e}_{l_{1}} + (\lambda_{l_{1}} - a_{l_{1}})\dot{e}_{l_{1}} - a_{l_{1}}\lambda_{l_{1}}e_{l_{1}} = 0$$

$$\vdots$$

$$\ddot{e}_{h_{N}} + (\lambda_{h_{N}} - a_{h_{N}})\dot{e}_{h_{N}} - a_{h_{N}}\lambda_{h_{N}}e_{h_{N}} = 0$$

$$\ddot{e}_{l_{N}} + (\lambda_{l_{N}} - a_{l_{N}})\dot{e}_{l_{N}} - a_{l_{N}}\lambda_{l_{N}}e_{l_{N}} = 0$$

The general solution x(t) (representing the general form of e_i and its derivatives) of the system in eq. 4.28 can be written as:

$$x(t) = \alpha_1 e^{\beta_1 t} + \alpha_2 e^{\beta_2 t} \tag{4.29}$$

where β_1 and β_2 are the roots of the second order linear differential equation related to x(t), and α_1 and α_2 are the gains related to the initial and final conditions of the solution.

The system in eq. 4.30 is the proposed velocity profile generator model used to compute the needed vehicle's velocity to reconfigure the formation from the initial shape toward its desired one. The generator model is inspired from eq. 4.29. The latter is used to control the convergence rate of the coordinates h and l with the help of five degrees of freedom (DOFs) K_1 , K_2 , a, λ and c.

$$\mathcal{V}(t) = K_1 e^{at} + K_2 e^{-\lambda t} + c \tag{4.30}$$

with a and λ being the roots of the differential equation in eq. 4.28. K_1 , K_2 and c are the gains used to take into account the initial and final conditions imposed to the velocity generator. The procedure used to solve the system in eq. 4.28 by the identification of the five DOFs of the velocity profile generator is described above:

1. In order to generate a velocity profile that takes into account explicitly the anticipation distance available to perform the merging, it is proposed to introduce the time t_{max} . The latter permits to set the moment where \mathcal{V} reaches its maximum, hence the acceleration $\dot{\mathcal{V}}$ is zero as expressed in eq. 4.31. Consequently, t_{max} permits to dynamically adapt the acceleration dynamic of the vehicles w.r.t. the length of the anticipation zone.

$$\dot{\mathcal{V}}(t_{max}) = aK_1 e^{at_{max}} - \lambda K_2 e^{-\lambda t_{max}} = 0 \tag{4.31}$$

2. The velocity profile generator in eq. (4.30) needs to take into account the initial velocity of the vehicles and the desired one at the end of the reconfiguration.

Thus $c = \mathcal{V}^{init} - K_1 - K_2$ is computed to impose the initial velocity. In order to impose the final velocity $\mathcal{V}(t = t_f) = \mathcal{V}^{end}$, with \mathcal{V}^{end} is the velocity of the reference vehicle, the eq. 4.32 is introduced.

$$K_1 e^{at_f} + K_2 e^{-\lambda t_f} - K_1 - K_2 + \mathcal{V}^{init} - \mathcal{V}^{end} = 0 \qquad (4.32)$$

3. Based on eq. 4.30, the expression of the position P(t) of the vehicle can be written as in eq. 4.33.

$$P(t) = \frac{K_1}{a}e^{at} - \frac{K_2}{\lambda}e^{-\lambda t} + (\mathcal{V}^{init} - K_1 - K_2)t + d$$
(4.33)

The term $d = P_0 - \frac{K_1}{a} + \frac{K_2}{\lambda}$ is used to impose the initial position P_0 of the vehicle at t = 0.

4. Let us define M(t) as the coordinate of the vehicle in the formation and M^{end} is its desired final coordinate. M can be either a longitudinal coordinate or a lateral one. In order to link the proposed convergence model to the coordinates used in the virtual structure formalism, eq. 4.34 is proposed.

$$M(t) = P_{ref} - P(t)$$
 (4.34)

with $P_{ref} = \mathcal{V}_{ref} * t + P_{ref_0}$ is the pose of the reference vehicle and \mathcal{V}_{ref} is the reference vehicle's velocity, said to be constant according to the Frenet based coordinates system used by the virtual structure approach.

To guarantee that the desired final coordinate is reached by the vehicle at $t = t_f$ (i.e., $M(t_f) = M_f$), eq. 4.35 is used.

$$\left(\mathcal{V}_{ref}t_f + P_{ref_0}\right) - \left[\frac{K_1}{a}e^{at_f} - \frac{K_2}{a}e^{-\lambda t_f} + \right]$$
(4.35)

$$\left(\mathcal{V}_{min} - K_1 - K_2\right)t_f + P_0 - \frac{K_1}{a} + \frac{K_2}{\lambda}\right] - M_f = 0 \tag{4.36}$$

To solve equations (4.31), (4.32) and (4.35), a numerical solver is employed. Following this, a prediction step is initiated using the generated velocity profiles and the initial conditions of the formation. Subsequently, the numerical solution is evaluated based on the satisfaction of the safety criterion and the dynamic feasibility (e.g., respect of the acceleration limits, the maximum authorized velocity, etc.).

As an example, when the formation reconfiguration takes into account longitudinal motion, a longitudinal velocity profile is generated based on the initial and final conditions and dynamics (cf. Figure 4.8, Segment A in green). Similarly, when both of the longitudinal and the lateral motions are the responsibility of the formation reconfiguration strategy, both velocity profiles are generated for the the longitudinal and lateral motions (cf. Figure 4.8, Segment B in red).

4.4.3 Simulation results

To access the effectiveness of the proposed FRA-OCS in ensuring the safety criterion and the dynamics smoothness of the formation reconfiguration during the on-ramp merging on highway, the evaluation focuses on analyzing the simulation results of merging scenario similar to the one proposed in Section 4.2 and Section 4.3. The simulation video can be found in https://youtu.be/3gDSq5L3m-c.

The following simulation aims to perform the merging scenario of a formation composed of four vehicles (three highway vehicles and one merging vehicle) illustrated in the initial shape of the formation in Figure 4.14.

The initial position of the merging vehicle is configured to trigger the safety of the formation. The FRA-OCS is expected to merge V_m between V_{hw_2} and V_{hw_3} , V_{hw_1} being the reference vehicle V_R (cf. Figure 4.14 (b)). The desired formation shape is a platoon formation as in Figure 4.14 (c).

The coordinates of the formation are illustrated in Figure 4.15. The green and the red shaded zones represent the formation coordinates in Segment A and Segment B, respectively. To maintain an appropriate longitudinal separation between the vehicles in the desired formation shape, an intervehicle distance corresponding to $\approx 2[s] \times \mathcal{V}[m/s]$ is employed. Since V_m is placed initially approximately at the same longitudinal position as the reference vehicle V_R , its longitudinal coordinate converges from its initial value toward the desired one w.r.t. the Frenet reference frame centered on V_{hw_1} (V_R).



(c)Platoon formation

Figure 4.14. Snapshots of the evolution of the formation shape during the onramp merging scenario obtained with the help of SCANeR studio (Simulation video: https://youtu.be/3gDSq5L3m-c)

As expected the longitudinal coordinate of the merging vehicle V_m increases toward the desired longitudinal coordinate, such as V_m merges between V_{hw_2} and V_{hw_3} . The lateral coordinate of the merging vehicle follows its reference trajectory in Segment A, while it is controlled by the inter-target distance matrix in Segment B.

To evaluate the formation's safety during the reconfiguration, the in-between Euclidean distances are depicted in Figure 4.16. The minimum Euclidean inter-vehicle distance $d_{safe} = \underline{D}_T$ is set to be equal to 10m. Given the initial placement of V_m w.r.t. V_{hw_1} , the Euclidean distance between these two vehicles is below the safety distance before the entry in Segment B, but since they are in two different lanes no collision is occurred. The Euclidean



Figure 4.15. The formation's coordinates

clidean in-between distance increases during the merging while the collision is being solved by the reconfiguration matrix. The rest of the in-between distance plots are all above the safety distance, thus the merging scenario is considered to be safe.

The dynamics of the vehicles during the merging are studied through the velocity profile in Figure 4.17. The initial velocity of the merging vehicle is set to be equal to 60[km/h], smaller than the initial velocity of the highway vehicles (set to be equal to 80[km/h]). Both of the velocities of V_{hw_2} and V_{hw_3} increase to reduce the gap w.r.t. the reference vehicle, to decrease afterwards in order to navigate at the same velocity as the reference velocity in order to respect the equal-distance policy part of the platoon requirement.



Figure 4.16. The Euclidean in-between distances



Figure 4.17. The vehicles' velocity profiles

4.4.4 Conclusion of the FRA-OCS algorithm

Through this section, it was proposed to overcome both of the CORM's and the E-CORM's frameworks limitations. This by introducing the Formation Reconfiguration Approach based on Online Control Strategy (FRA-OCS). The proposed FRA-OCS takes advantages from both of the constrained inter-distance matrix and the road segmentation strategy to perform an online formation reconfiguration. In fact, the non-linear optimization algorithm used to compute the reconfiguration matrix during the formation's merging is replaced with an online strategy. A system of equations based on the formalization of the merging scenario, the initial and the final conditions of the vehicles part of the formation was proposed. The latter is solved using a numerical solver to obtain the needed dynamics for each vehicle part of the formation to perform the merging.

The evaluation of the FRA-OCS algorithm for dynamic formation reconfiguration was carried out in a simulation environment. It showcased the algorithm's ability to adhere to safety criterion and its enhanced flexibility.

4.5 Chapter conclusion

This chapter has extensively examined cooperative formation reconfiguration, presenting and evaluating three distinct frameworks: the Constrained Optimal Reconfiguration Matrix (CORM), the Extended Constrained Optimal Reconfiguration Matrix (E-CORM), and the Formation Reconfiguration Approach based on Online Control Strategy (FRA-OCS).

The CORM framework employs two-step process, utilizing an optimization algorithm to account for environment constraints and a projection approach for safe and feasible merging. While simulation evaluations demonstrated its success in challenging scenarios, it was acknowledged that the CORM has limitations, such as assuming identical dynamics for longitudinal and lateral reconfiguration, impacting its flexibility.

Introducing the E-CORM aimed to address the CORM's lack of flexibility. The extended inter-target distance matrix and road segmentation strategy enhanced motion flexibility, allowing vehicles to precisely follow their global reference path. Evaluation in a simulation environment showcased the algorithm's enhanced flexibility, but reliance on a non-linear optimization algorithm remained a limitation.

In response to the limitations of the CORM and the E-CORM, the FRA-OCS was proposed. This approach combines the constrained inter-target distance matrix and road segmentation strategy for online formation reconfiguration. By replacing the non-linear algorithm with an online strategy based on numerically solving of a system of equations, the FRA-OCS demonstrated success in dynamic formation reconfiguration in simulation environments. While each framework exhibits strengths, it is essential to recognize their specific limitations. CORM faced challenges with different dynamics in longitudinal and lateral reconfiguration, while both of the CORM and the E-CORM relied on a time-consuming optimization algorithm. Additionally, the FRA-OCS is the formation reconfiguration that met the C-MCA objectives.

The formation reconfiguration mechanism part of the planning level translates the passing sequence provided by the decision-making level to perform the merging. The following chapter delves in the multi-behavior decisionmaking strategy used by the decision-making level part of the C-MCA to provide this passing sequence.

CHAPTER 5 Multi-Behavior Decision-Making Strategy

Abstract

This chapter is dedicated to draw the details related to the decisionmaking level part of the proposed Cooperative Multi-Controller Architecture (C-MCA) for MVS performing on-ramp merging on highway, thus under the prisme of safety, passenger comfort, and energy efficiency. The proposed multi-behavior selection strategy predicts the safety assessment during the merging scenario based on a nominal behavior of the MVS. Based on the safety criterion, the optimal behavior of the MVS is chosen, either to perform the merging in a nominal manner, or to overcome a conflicting situation in a cooperative manner ¹.

Contents

5.1	Multi-	Multi-behavior selection strategy									
5.2	Nomin	al behavior									
	5.2.1	Overview of the nominal behavior part of the C-MCA 13	34								
	5.2.2	Simulation results	36								
5.3	Coope	rative behavior	38								
	5.3.1	Passing sequence selection	39								
	5.3.2	Suitable passing sequence	41								
		5.3.2.1 Global cooperation cost function $\ldots \ldots 14$	43								

¹The detailed contributions in this chapter have been the subject of the following publications ITSC'23, MMAR'23 and VAMS'23.

	5.3.2.2 Altruistic passing sequence	145
5.4	Simulation results	146
5.5	Conclusion	155

The decision-making level part of the proposed Cooperative Multi-Controller Architecture (C-MCA) for MVS was designed to answer three main questions:

- 1. What is the nominal dynamic for each vehicle part of the MVS to perform the merging?
- 2. How to predict the potential conflicting merging scenarios when the vehicles are operated by their nominal behavior?
- 3. How to solve the conflicting merging when occurred, with minimum changes imposed to the nominal behavior, while taking advantage of the MVS cooperative ability?

The decision-making level employs a multi-behavior decision-making strategy. Based on the collision risk metric, the proposed strategy depicted in Figure 5.1, has the responsibility of switching between two embedded behaviors. The first objective of this chapter is to delve into the selection strategy used by the decision-making-level. In Section 5.1, the details of the selection strategy are given. The second objective of this chapter is to delve into the details of the nominal behavior through Section 5.2. Additionally, the details the of the cooperative behavior are discussed in Section 5.3. Where the concept of the passing sequence is explained, and an appropriate passing sequence selection approach is proposed.

The performance of the multi-behavior decision-making strategy, serving as the decision-making level within the C-MCA, is assessed through simulations detailed in both: the Section 5.2.2 for the nominal behavior and in Section 5.4 for the cooperative behavior.

5.1 Multi-behavior selection strategy

One of the central objectives of the proposed C-MCA is to ensure that the merging maneuver of the MVS is executed with appropriate dynamics while prioritizing the safety requirements. This involves the design of the MVS's nominal behavior, which is focused on the optimization of the vehicles' dynamics in alignment with their individual objectives (cf. Section 5.2).



Figure 5.1. The proposed multi-mode behavior selection level, part of the overall C-MCA (cf. Figure 3.2)

To initiate the behavior selection process, the first step is to predict the nominal behavior of each vehicle within the MVS (cf. Figure 5.1). Following the nominal behavior generation strategy, this prediction process also extends to forecasting the distances between the vehicles throughout the merging maneuver. This step is crucial in identifying potential collision risks among the vehicles (cf. Figure 5.1).

Subsequently, the safety metric is assessed under two main scenarios, as illustrated in Figure 5.1. In the first scenario, which corresponds to the absence of predicted collisions within the merging zone. Triggered by the nominal behavior predictions, the decision is made to activate the nominal behavior is initially activated, the safety criterion is continuously monitored through a feedback loop. If, at any point, the safety criterion predicts a collision within the merging zone, the decision-making level promptly activates the cooperative behavior to avert and avoid the collision risk. In the second scenario, where a predicted collision risk is present, the C-MCA activates the cooperative behavior (cf. Section 5.3).

In the following section, the details related to the nominal behavior are

given along with corresponding simulation results.

5.2 Nominal behavior

This section is dedicated to draw the explanation related to the nominal behavior part of the multi-behavior decision-making implemented in the C-MCA. Simulation results on an on-ramp merging scenario performed by the nominal behavior are presented in this section.

5.2.1 Overview of the nominal behavior part of the C-MCA

The goal of the nominal mode is to perform the merging maneuver while optimizing each vehicle's dynamics w.r.t. their individual goals (cf. Figure 5.2). For instance, the highway vehicles have an already established highway dynamics, thus their aim is to keep tracking the latter. As for the merging vehicle, it aims to perform the merging maneuver while minimizing the merging time, and to not modify a lot its planned velocity (generally depending on the road velocity limitation).

This mode takes the assumption of a free merging zone. Consequently, no conflicting scenario is considered. It is important to note that this assumption permits to focus only on the vehicles' dynamics, such as their individual goals are satisfied. Consequently, the safety criterion when generating the nominal mode is assessed by the decision-making level.

The nominal behavior of the highway vehicles consists of their initial dynamics through the merging maneuver. In fact, the highway vehicles have an already established dynamic when traveling in the highway (cf. Figure 5.2) (in general calibrated w.r.t. the maximum authorized highway velocity), thus in order to minimize velocity and acceleration changes w.r.t. their initial dynamics, the nominal behavior supposes that the highway vehicles have constant and equal to their initial velocity.

As for the merging vehicle in Figure 5.2, when navigating in the merging secondary road, its velocity is lower than the highway maximum authorized velocity. Since the merging vehicle goal is to perform the merging maneuver while minimizing the required time, the nominal behavior goal for the merging vehicle is to ensure the transition from its initial dynamic to its highway one. In order to establish this transition, a two sub-criteria



Figure 5.2. Illustration of the on-ramp merging on highway scenario seen by the nominal behavior

optimization formalization is proposed taking into account: the necessary merging time and its minimal acceleration profile (cf. eq. 5.1). The latter is correlated to the energy consumption of the merging vehicle while performing the merging maneuver.

The nominal behavior for the merging vehicle uses an optimization process to generate the velocity cycle given to the merging vehicle as an input. This velocity cycle has the responsibility of transitioning from the merging vehicle initial velocity toward the highway maximum authorized velocity (cf. Figure 5.2). The latter is based on a sigmoid function given in eq. 5.1.

$$\mathcal{V}(k) = \mathcal{V}_m^{init} + \frac{\mathcal{V}_m^{desired} - \mathcal{V}_m^{init}}{1 - e^{-\alpha^*(k - \beta^*)}}$$
(5.1)

with \mathcal{V}_m^{init} the initial velocity of the merging vehicle V_m , and $\mathcal{V}_m^{desired}$ its desired velocity. The latter is based on \mathcal{V}_{max} , the maximum authorized velocity by the traffic laws on the traveled segment (cf. Figure 5.2). k represents the sample time. α is the convergence rate of the sigmoid function while β is the mid-point time. α^* and β^* are the optimal parameters that must be found by the optimization process in order to satisfy eq. 5.2.

$$\min_{\substack{\alpha,\beta}} \qquad \omega_1 \frac{TravelTime}{\overline{t}} + \omega_2 \sum_{k=1}^{I_N} \left(\frac{a(k)}{\overline{a}}\right)^2$$
so that $-4[m/s^2] \le a(k) \le 4[m/s^2], \ k \in \{1, I_N\}$

$$\mathcal{V}(k) \le \mathcal{V}_{max}, \qquad k \in \{1, I_N\}$$
(5.2)

The objective function used to find the optimal sigmoid parameters α^* and β^* is given in eq. 5.2, it is composed of two sub-criteria weighted functions

with the help of ω_1 and ω_2 . 1) The time related cost aims to minimize the required time (*TravelTime*) for V_m to perform the merging. 2) The acceleration related cost aims to minimize the variation of the acceleration $a(k), k \in \{1, I_N\}$ (considered lower than the maximal authorized acceleration) during the merging and I_N is the number of time samples during the merging performed by the nominal behavior. Consequently, it results in an energy efficient merging maneuver from the V_m perspective.

 \overline{t} and \overline{a} are the normalization terms. They represent respectively, \overline{t} the maximum time needed to perform the maneuver, which is obtained by imposing to V_m to maintain its initial velocity during all the merging phase, and \overline{a} the maximum acceleration from all the obtained acceleration profile a(k).

5.2.2 Simulation results

In the simulation given in Figure 5.3, it is aimed to perform the merging scenario according to the nominal behavior. The merging vehicle V_m initial pose is set to respect the safety requirement such as the nominal behavior is activated, in addition to a proper spacing of the highway vehicles $V_{hw_{1,..,3}}$ and V_m . The simulation video can be found in https://youtu.be/8Cqw7GEFKd4.



Figure 5.3. Illustration of the on-ramp merging scenario at the vehicles initial state (Simulation video: https://youtu.be/8Cqw7GEFKd4)

Figure 5.4 presents the Euclidean in-between distances. During the merging scenario, it can be noticed that both the distance between the merging vehicle V_m and the highway vehicles V_{hw_2} and V_{hw_3} are above the minimum in-between distance meaning that the safety requirement is respected with the help of the nominal behavior.



Figure 5.4. Nominal behavior: progress of the Euclidean in-between distances

The vehicles' velocity cycles and the corresponding acceleration profiles are depicted in Figure 5.5. In the merging road (where V_m is located initially) the maximum authorized velocity is $80 \, km/h$ (cf. Figure 5.3), consequently, V_m velocity profile converges toward the maximum authorized velocity, while respecting the maximum allowed acceleration by the nominal behavior(cf. Figure 5.5, Acceleration profiles). When V_m enters the highway through the merging zone, the maximum authorized velocity becomes $130 \, km/h$ (cf. Figure 5.3). Consequently, its velocity cycle increases from $80 \, km/h$ toward $130 \, km/h$, to decrease toward the highway velocity established at $110 \, km/h$ in order to guarantee an equal spacing policy w.r.t. the highway vehicles. As for the highway vehicles, their velocity is constant through the performance of the merging scenario as planned by the nominal behavior.

In the simulation results presented above, the safety requirement was respected following the nominal behavior. However, a collision may occur when following the dynamic generated by the nominal behavior. In this case, the multi-behavior decision-making level activates the cooperative behavior. The following section is dedicated to the presentation of the details related to the cooperative behavior.



Figure 5.5. Nominal behavior: The vehicles' velocity and acceleration profiles

5.3 Cooperative behavior

The main goal of this section is to answer the following question: How to solve the conflict when occurred, with minimum changes imposed to the nominal behavior, while taking advantage of the MVS cooperative ability?

In order to solve a conflicting situation, the decision-making focuses on providing a safe passing order of the vehicles in the merging zone. Thus, this section delves into the details related to the selection of this safe passing order. In Section 5.3.1, the global concept of the passing sequence is presented as well as the establishment of the possible passing orders. Followed by Section 5.3.1, where the evaluation and the selection of the desired passing sequence is discussed. This section ends with a dedicated simulation results part, where several simulation scenarios were built to test the cooperative behavior ability to solve conflicting situation even in highly dynamic environments.

5.3.1 Passing sequence selection

The merging zone (cf. Figure 5.7) is a shared topological resource between the highway vehicles and the merging vehicles. The cooperation behavior part of the C-MCA (cf. Figure 5.1 is activated when a collision may occurs in the merging zone, and this if the nominal behavior is applied. Consequently, the cooperation protocol has the responsibility of setting a conflict-free passing sequence sq of the vehicles in the merging zone. The passing sequence sq designs the order on which the vehicles pass the merging zone. Once sq is selected, it will be translated to a desired formation virtual shape with the help of the formation coordinates, which constitute the input of the cooperative trajectory planning level (cf. Chapter 4).

Through the analysis of the prediction based on the nominal behavior, given in Section 5.3.1, it is proposed to delve into the details of the selection of the potential passing sequence. The selected passing sequence is evaluated through a prediction of the MVS corresponding dynamics. Section 5.3.2 delves into the process of the optimal passing sequence.

Potential passing sequences

The selection of the cooperative behavior passing sequence is based on the safety information obtained by the prediction of the vehicles' dynamics following the nominal behavior. In other terms, based on the predicted behavior, the collision is virtually simulated in order to extract the following information: The Collision Time (CT), the Collision Partner (CP), and an image of the merging scenario at CT, the latter is used in order to analyze the potential collisions, build a list of collision partners. Figure 5.6 presents the flowchart illustrating this procedure.

The image of the merging scenario at CT is used in order to build the list of potential passing sequences. The proposed strategy detects all the potential MVS in-between collisions and classify them in term of riskiness². In Figure 5.7, the image of the merging scenario at CT, where the virtual projection of the merging vehicle V_m^{end} and the first collision partner V_{CP_1} are expected to collide at t = CT.

For instance, the list of the potential passing sequences for the scenario in Figure 5.7 can be built as follows:

 $^{^{2}}$ The potential collisions are evaluated and classified in terms of their priority. The classification is based on the distance w.r.t. merging zone.



Figure 5.6. Flowchart of the prediction based on the nominal behavior



Figure 5.7. Image of the merging scenario at t = CT (Collision Time)

- 1. The merging vehicle in-between both of the highway vehicles, sq(1) : V_{CP_1}, V_m, V_{hw_2}
- 2. The merging vehicle in front of both of the highway vehicles, sq(2):

 V_m, V_{CP_1}, V_{hw_2}

3. The merging vehicle in the back of both of the highway vehicles, sq(3): V_{CP_1}, V_{hw_2}, V_m

In crowded and highly dynamic environment such as the highway, the established list of potential passing sequences can be unsafe. Thus, the potential passing sequence list is augmented with additional potential passing sequences that take into account lane change of the highway vehicles.

- 1. V_{CP_1} changes lane such as the merging vehicle takes its place, sq(4): $V_{CP_{1_{1}\to 2}}, V_m, V_{hw_2}$
- 2. V_{CP_2} changes lane such as the merging vehicle has enough space to merge, $sq(5): V_{CP_1}, V_m, V_{hw_{2_{1}\rightarrow 2}}$

Once the list of the potential passing sequences is established, it has to be evaluated. The following section is dedicated to draw the details of the selection of the suitable passing sequence.

5.3.2 Suitable passing sequence

The suitable passing sequence defines the passing order of the vehicles in the merging zone that solves the collision problem occurred when following the MVS nominal behavior, and that imposes the least changes w.r.t. the MVS nominal dynamics. In other terms, the MVS nominal dynamics are used as a reference dynamics in order to find the closest cooperative behavior to the nominal one, by evaluating the cooperative behavior dynamics that allow to perform safely the potential passing sequence safely.

Figure 5.8 presents the flowchart of the selection of the passing sequence through an evaluation process. The evaluation of each potential passing sequence is described in Section 5.3.2.1, it is based on several quantifiable criteria such as the safety and the dynamics related to the passing sequence. The evaluation focuses also on the altruistic character of the chosen passing sequence. The details of the latter are discussed in Section 5.3.2.2.

In order to evaluate the performance of the passing sequence selection algorithm, two illustrative scenarios are studied in Section 5.4.



Figure 5.8. The flowchart of the MVS passing sequence in the merging zone selection strategy

5.3.2.1 Global cooperation cost function

A global cooperation cost function J_G is used to evaluate the level of cooperation related to the different passing sequences sq. It uses the initial conditions of the highway vehicles V_{hw} as a reference to define the cooperation effort, and the nominal behavior (cf. Section 5.2) to compare the performance of the merging vehicles V_m . It can be written as:

$$J_G(sq(j)) = \sum_{i=1}^N \omega_i J_i \tag{5.3}$$

$$\sum_{i=1}^{N} \omega_i = 1 \tag{5.4}$$

 ω_i represents the participation weight given to the vehicle V_i and it respects the expression given in eq. 5.4. $i \in N$ represents the index of the N vehicles part of the considered formation. J_i given in eq. 5.5 is the individual cost function related to each considered vehicle.

$$J_{i=\omega_{safe}J_i^{Safe}+\omega_{Acc}J_i^{Acc}+\omega_{KE}J^{KE}+\omega_{NC}J^{NC}, \quad \forall i \in \mathbb{N}$$

$$(5.5)$$

with:

$$\omega_{safe} + \omega_{Acc} + \omega_{KE} = 1 \tag{5.6}$$

$$\omega = \{\omega_{safe}, \omega_{Acc}, \omega_{KE}, \omega_{NC}\}$$
(5.7)

 ω is the set containing the participation weights given to each sub-cost part of the individual cost J_i . The non-collaborative weight is used in addition of including non-cooperative vehicle part of the C-MCA, to calibrate the cooperation efforts related to every vehicle. Thus, ω_{NC} can only take two states; when the vehicle is said to be cooperative ω_{NC} is null, in contrast, when the vehicle is said to be not cooperative ω_{NC} is equal to big value, allowing to not consider the corresponding solution. The cost J_i is composed of the four components described above:

(a) The safety related cost J_i^{Safe} considers the Euclidean inter-vehicular distances in between the vehicles part of the formation. It can be written as in eq. 5.8:

$$J_{i}^{Safe} = \frac{\sum_{k=1}^{I_{S}} \frac{1}{EucDist\{X_{V_{i}}(k), X_{V_{j}}(k)\}^{\forall j \in N, j \neq i}}}{\sum_{k=1}^{I_{S}} \frac{1}{d_{Safe}}}, \quad \forall i \in N$$
(5.8)

 d_{safety} is the safety distance that needs to be always respected between the vehicles. It corresponds to twice the radius of the safety circle around the vehicle (cf. Figure 5.7). The latter is chosen to take into account the dimensions of the vehicle in addition of a certain offset. I_S represented the time samples needed to perform the merging with the help of the cooperative behavior. It is used here to normalize the safety cost.

(b) The cost related to the dynamics of the vehicles J_m^{Acc} utilizes the acceleration. This cost aims to minimize the gap between the nominal dynamic and the cooperation-based dynamic related to V_m , $\forall m \in \mathbb{M}$. The subset \mathbb{M} with $\mathbb{M} \in N$ contains only the *m*-indexes of the merging vehicles V_m (cf. Figure 5.7).

$$J_m^{Acc} = \left| \frac{1}{I_N} \sum_{k=1}^{I_N} \left[\frac{a_m^{nominal}(k)}{\overline{a_m^{nominal}}} \right]^2 - \frac{1}{I_S} \sum_{k=1}^{I_S} \left[\frac{a_m^S(k)}{\overline{a_m^S}} \right]^2 \right|, \quad \forall m \in \mathbb{M}$$
(5.9)

As for the highway vehicles V_{hw} , their dynamics are represented with the cost J_{hw}^{Acc} , $\forall hw \in \mathbb{H}$. The subset \mathbb{H} with $\mathbb{H} \in N$ contains only the *hw*-indexes of the highway vehicles V_{hw} (cf. Figure 5.7).

$$J_{hw}^{Acc} = \frac{1}{I_S} \sum_{k=1}^{I_S} \left[\frac{a_{hw}^S(k)}{\overline{a_{hw}^S}} \right]^2, \quad \forall \, hw \in \mathbb{H}$$
(5.10)

with a_m^S and a_{hw}^S being respectively, the acceleration profile of the V_m and V_{hw} during the merging scenario under the cooperation behavior. $a_m^{nominal}$ and a_m^S are the maximum acceleration with the nominal behavior (cf. Section 5.2) and with the cooperation behavior, respectively. The latter are used to normalize the acceleration cost.

(c) The cost related to the energy generated by the cooperative behavior is characterized by the kinetic energy used by the vehicles (cf. eq. 5.11 and eq. 5.12). The kinetic energy related cost has two objectives: 1) It gives the energy consumption of the vehicle even at zero acceleration.
2) Since it includes the weight of the vehicle, it allows for instance the distinction of the efforts asked to be done by a truck and a light-weight vehicle.

Same as for the dynamic related cost, the nominal dynamic (cf. Section 5.2) is used to evaluate and minimize the gap between the nominal

behavior dynamic and the cooperation one. Thus, the kinetic energy cost for the merging vehicle is shown in eq. 5.11.

$$J_m^{KE} = \frac{1}{2} m_{V_m} \Big| \frac{1}{I_N} \sum_{k=1}^{I_N} \Big[\frac{\mathcal{V}_m^{nominal}(k)}{\mathcal{V}_m^{nominal}} \Big]^2 - \frac{1}{I_S} \sum_{k=1}^{I_S} \Big[\frac{\mathcal{V}_m^S(k)}{\mathcal{V}_m^S} \Big]^2 \Big|, \quad \forall m \in \mathbb{M}$$

$$(5.11)$$

The kinetic energy related to the highway vehicles is noted J_{hw}^{KE} and it is given in eq. 5.12.

$$J_{hw}^{KE} = \frac{1}{2} m_{V_{hw}} \left(\frac{1}{I_S} \sum_{k=1}^{I_S} \left[\frac{\mathcal{V}_{ref}^S(k) - \mathcal{V}_{hw}^S(k)}{max(\overline{\mathcal{V}_{ref}^S}, \overline{\mathcal{V}_{hw}^S})} \right]^2 \right), \quad \forall hw \in \mathbb{H}$$
 (5.12)

with m_{V_i} being the weight of the vehicle V_i . \mathcal{V}_m^S and \mathcal{V}_{hw}^S are the velocity profiles of both V_m and V_{hw} , respectively, during the merging scenario under the cooperation behavior. \mathcal{V}_{ref}^S is the desired velocity of the V_{hw} . The term $max(\overline{\mathcal{V}_{ref}^S}, \overline{\mathcal{V}_{hw}^S})$ is used for the normalization of the kinetic energy cost.

5.3.2.2 Altruistic passing sequence

The fourth term of the individual cost function is the non-collaborative cost. Its objective is to ensure the avoidance of extensive cooperation efforts from the perspective of the V_{hw} . In fact, the highway MVS are said to be altruistic, in other terms, they are set as collaborative by definition. However, the cost in eq. 5.13 plays the role of the cooperation threshold from the V_{hw} perspective. The threshold represents the maximum cooperation tolerance of the highway vehicle. Consequently, when a passing order sq orders the V_{hw_i} an effort above its effort's threshold, the non-collaborative cost is maximized.

$$\omega_{NC} = \begin{cases} 0 & \text{for } J_{NC} < threshold \\ \text{inf } & \text{for } J_{NC} \ge threshold \end{cases}$$
(5.13)

The non-collaborative cost J_{NC} represents the highway vehicle cooperation effort. In fact, the nominal behavior is said to be the best behavior, thus, J_{NC} aims to quantify the gap separating the dynamics imposed by the cooperative behavior in comparison to the nominal ones. The non-cooperative cost J_{NC} is given in equation in eq. 5.14.

$$J_{NC} = \left| \frac{J_{cooperation} - J_{nominal}}{J_{nominal}} \right|$$
(5.14)

with $J_{cooperation}$ and $J_{nominal}$ being the cooperation-related and the nominal-related costs, respectively. Both of the costs are given in eq. 5.15 and eq. 5.16, respectively.

$$J_{nominal} = \sum_{k=1}^{I_N} \frac{\mathcal{V}(k)_{nominal}}{\overline{\mathcal{V}}_{nominal}} + \sum_{k=1}^{I_N} \frac{a(k)_{nominal}}{\overline{a}_{nominal}}$$
(5.15)

$$J_{cooperation} = \sum_{k=1}^{I_S} \frac{\mathcal{V}(k)_{cooperation}}{\overline{\mathcal{V}_{cooperation}}} + \sum_{k=1}^{I_S} \frac{a(k)_{cooperation}}{\overline{a}_{cooperation}}$$
(5.16)

where $\{\mathcal{V}_{nominal}, a_{nominal}\}$ are the velocity and the acceleration profile generated by the nominal behavior, and $\{\overline{\mathcal{V}_{nominal}}, \overline{a_{nominal}}\}$ are their maximum values. $\{\mathcal{V}_{cooperation}, a_{cooperation}\}$ are the velocity and the acceleration profile generated by the cooperative behavior, and $\{\overline{\mathcal{V}_{cooperation}}, \overline{a_{cooperation}}\}\$ are their maximum values.

5.4 Simulation results

The proposed decision-making level part of C-MCA was designed to select the appropriate behavior of the MVS when performing the merging. As stated above, two behaviors are implemented in the C-MCA, so that the MVS prioritize a nominal merging maneuver when possible. If needed, the MVS is expected to cooperate in order to overcome the collision risk related to the merging, without neglecting the dynamics of the vehicles part of it.

In this section it is proposed through two illustrative scenarios to study and evaluate the decision chosen by the decision-making level. Thus, the C-MCA is asked to solve both scenarios, as well as to evaluate the solution through the prism of the safety, feasibility and energy efficiency related to the MVS. The simulations were conducted using Matlab/Simulink and SCANeR studio simulator 1.2.1.2.

The first scenario aims to test the C-MCA ability to solve a conflicting on-ramp merging on highway. Based on the passing sequence selected by the decision-making level, the E-CORM is tasked to generate the required MVS dynamics needed w.r.t. the passing sequence. The details and the results related to this first scenario are presented in this section.

In addition to find a suitable solution, the C-MCA needs to assess the safety criterion and when needed, online resolve conflicting scenarios. The second scenario was built to test this particularity. Thus, with the help of the cooperative optimization free formation reconfiguration framework seen in Section 4.4, a dynamic scenario is proposed where the C-MCA performed the merging safely.

Scenario 1

The following simulation presents an on-ramp merging on highway performed by four vehicles (three highway vehicles V_{hw_1} , V_{hw_2} , V_{hw_3} and the merging vehicle V_m) being part of the MVS (cf. Figure 5.9). In order to study the C-MCA ability to solve conflicting mergings, the MVS vehicles are placed in positions where the safety criterion is triggered (cf. Figure 5.9). Thus, the cooperation behavior is activated in order to solve the conflict occurred in the merging zone. The simulation video related to this scenario can be found in https://shorturl.at/gjGKL.



Figure 5.9. Scenario 1: Illustration of the on-ramp merging on highway scene. The initial shape of the formation under the communication range of the RSU is represented, along with the desired shape

Figure 5.10 (b.1) shows the formation shape when the merging is being performed by the nominal behavior. In fact, a collision is detected between the merging vehicle and the highway vehicle at t = 3.6s. Consequently, the cooperative behavior is asked to solve the predicted collision. Thus, a list of potential passing sequences was built. In Table 5.1 the results of the best passing sequence is shown.

Several passing sequences were tested and evaluated in terms of their safety, comfort and energy efficiency. The passing sequence selected by the cooperation behavior is $sq = \{V_{hw_1}, V_{hw_2}, V_{hw_3}, V_m\}$ (cf. Table 5.1). Figure 5.10 (b.2) shows the formation shape when merging vehicle enters the merging zone at t = 3.6s. It can noticed that the selected passing sequence



(C): T=14s, snapshot of the merging scenario at the end of reconfiguration

Figure 5.10. Evolution of the formation' shape during the on-ramp merging on high-way

 Table 5.1.
 Scenario 1: Numerical results of the cooperation behavior

sq	V	V_{hw_1}, V_{hw_2}	$_{2}, V_{hw_3}, V_{hw_3}$	m	$V_{hw_1}, V_{hw_2}, V_m, V_{hw_3}$				
J_G		0.1	159		0.1642				
Vehicle id	hw_1	hw_2	m	hw_3	hw_1	hw_2	m	hw_3	
J ^{safe}	0.0705	0.0884	0.1058	0.1031	0.0871	0.0894	0.1451	0.1031	
Jacc	0	0	0.1290	0.0571	0	0	0.1188	0.0971	
J ^{KE}	0	0	0.0574	0.1159	0	0	0.1405	0.1509	

has solved the collision occurred when following the nominal behavior. In Figure 5.10 (c), the formation shape at the end of the merging maneuver is illustrated, where the MVS follows a platoon-based formation shape as expected.

The Euclidean distances within the MVS are given in Figure 5.11, where conflict between the merging vehicle V_m and the highway vehicle V_{hw_3} can be noticed around t = 3.6s (cf. Figure 5.11, in dashed magenta). Thanks to the cooperation behavior part of the C-MCA, the conflict is solved, where the Euclidean distance between V_m and V_{hw_3} is always greater than the minimum allowed safety distance (cf. Figure 5.11, in blue plot).



Figure 5.11. Euclidean in-between distances within the formation

The overall MVS reconfiguration during the merging respects the safety criterion, since the rest of the vehicles Euclidean in-between distances are all greater than the minimum safety distance.

The velocity profiles of the vehicles corresponding to the selected passing sequence sq (cf. Table 5.1) are depicted in Figure 5.12 (a). The velocity profiles respect the maximum authorized velocity (80 km/h in segment A and 130 km/h in segment B). Since the highway vehicle V_{hw_3} is expected to be ahead of the merging vehicle, it has to accelerate in order to reduce its spacing w.r.t. V_{hw_2} . The merging vehicle has a lower initial velocity in comparison to the highway vehicles. Consequently, it has to accelerate,

while respecting the authorized maximum velocity in each traveled segment, in order to respect the selected passing sequence sq. The final velocity of the MVS corresponds to the reference vehicle V_{hw_1} velocity in order to be able to maintain the desired platoon shape with equal in-between distances. The overall acceleration profiles respect the maximum and minimum authorized limits (cf. Figure 5.12 (b)).



Figure 5.12. The vehicles' velocity and acceleration profiles

Scenario 2

The goal of this scenario is to evaluate the capacity of the C-MCA to access the safety criterion of the MVS during the merging maneuver, while being able to solve online conflicting situation when occurred. Consequent, an unforeseen event is introduced to the studied scenario at t = 8s. The scenario is divided into two part, the first part is related to the merging scenario before t = 8s, while the second part focuses on the effects of the unexpected introduced event and its begins after t = 8s. The simulation video can be found in https://youtu.be/u0LRJbOW94M.

The scenario in figure 5.13 uses a formation of five vehicles that participate into the on-ramp merging on highway. The initial position of the merging vehicle V_m is configured to trigger the safety criterion part of the proposed C-MCA, in Figure 5.14 (a) Initial shape of the formation, the initial vehicles positions are illustrated with the help of the snapshot from the SCANeR studio engine. Consequently, it is the cooperative behavior that will generate the vehicles' dynamics in order to perform the merging maneuver. In Table 5.2, the numerical results of the cooperative behavior are shown. The $sq = \{V_{hw_1}, V_{hw_2}, V_m, V_{hw_3}\}$ is the best passing sequence according to the cooperative behavior (cf. Section 5.3).



Figure 5.13. Scenario 2: Illustration of the on-ramp merging on highway scene. The initial shape of the formation under the communication range of the RSU is represented, along with the desired shape (Simulation video: https://youtu.be/u0LRJbOW94M)



Figure 5.14. Illustrative snapshots obtained from the SCANeR studio engine presenting the evolution of the formation during the merging maneuver

sq	V_h	$w_1, V_m,$	V_{hw_2}, V_h	w_3	$V_{hw_1}, V_{hw_2}, V_m, V_{hw_3}$				
J_G		0.1	.96		0.181				
Vehicle id	hw ₁ m		hw ₂	hw ₃	hw ₁	m	m hw ₂		
J^{safe}	0.125	0.15	0.103	0.066	0.109	0.25	0.227	0.063	
Jacc	0	0.11	0.047	0.063	0	0.335	0.025	0.035	
J^{KE}	0.127	0.586	0.292	0.292	0.127	0.102	0.11	0.147	

Table 5.2. Scenario 2.1: Numerical results of the cooperation behavior before t = 8s

However, around t = 8s, the fifth vehicle decides to join the formation (cf. Figure 5.14 (b)). As a result, the initial passing sequence no longer satisfies the safety requirement, prompting the cooperative behavior to recalculate a suitable sq to ensure safe reconfiguration. Four passing sequences were considered by the cooperative behavior part of the C-MCA. In Table 5.3, the numerical results of the two best sq are presented.

Table 5.3. Scenario 2.2: Numerical results of the cooperation behavior after t = 8s

sq	I	V_{hw_1}, V_{hw_1}	v_2, V_m, V	V_{hw_4}, V_{hw}	3	$V_{hw_1}, V_{hw_{2_{1}\rightarrow 2}}, V_m, V_{hw_4}, V_{hw_3}$				
J_G			0.267			0.242				
Vehicle id	hw ₁	m	hw ₂	hw ₃	hw ₄	hw_1	m	hw_2	hw ₃	hw_4
J^{safe}	0.104	0.358	0.227	0.085	0.011	0.101	0.185	0.129	0.106	0.017
J^{acc}	0	0.11	0.135	0.035	0.041	0	0.156	0.089	0.031	0.152
J^{KE}	0.127	0.402	0.188	0.147	0.152	0.127	0.262	0.136	0.153	0.148

The $sq = V_{hw_1}, V_{hw2_{1\to 2}}, V_m, V_{hw_4}, V_{hw_3}$ is the best passing sequence. The V_{hw_2} is asked to change the lane from *lane* 1 to *lane* 2 (cf. Figure 5.14 (c) Formation shape at the merging zone). The simulation results obtained following the selected sq are presented in the following figures.

In Figure 5.15, the longitudinal and the lateral coordinates of the formation are presented. The green shaded part is related to the merging vehicle traveling in segment A, while the red shaded part related to the segment B. The inclusion of the fifth vehicle at t = 8s stimulated the reactivity of the C-MCA, leading to a switch in the configuration at t = 8s (cf. Figure 5.15). The continuity of the reconfiguration is ensured formally by the optimization-free formation reconfiguration framework presented in Section 4.4. The selected sq requires a lane change of V_{hw_2} , what can be seen in its lateral coordinate plot. The merging vehicle exits the merging road to join the highway desired lane according to its lateral coordinate plot. The desired shape of the formation is a platoon formation for the vehicles traveling on the same lane with equal spacing w.r.t. the reference vehicle



Figure 5.15. Formation coordinates during the on-ramp merging on highway

The Euclidean in-between distances are presented in Figure 5.16. The highway vehicle V_{hw_2} and the merging vehicle travel in two different lanes in the desired formation shape, but placed according to the same longitudinal coordinate w.r.t. the reference vehicle V_{hw_1} . Consequently, the Euclidean distance between these two vehicles is above d_{safe} , due to the existence of only the lateral separation (two different lanes). The rest of the Euclidean distances are all greater than d_{safe} , thus the merging can be considered as a safe maneuver.

The velocity profiles of the vehicles part of the MVS performing the merging



Figure 5.16. The Euclidean in-between distances

are displayed in Figure 5.17. As can be seen, the formation switch at t = 8s created a dynamic switch in the merging vehicle velocity profile. The rest of the vehicles' velocity profiles respect the maximum authorized velocity in the traveled segment.



Figure 5.17. The vehicles' velocity profiles

5.5 Conclusion

This chapter has provided an in-depth exploration of the multi-behavior decision-making strategy within the proposed C-MCA. The decision-making strategy is structured around two distinct behaviors:

- The nominal behavior: This behavior is specifically crafted to execute the merging maneuver while ensuring the fulfillment of individual vehicle goals.
- The cooperative behavior: Activated by the decision-making level when the safety requirement is not met by the nominal behavior, the cooperative behavior aims to address merging conflicts by generating a safe passing sequence. The particularity of the generated passing sequence lies in the selection process governing its generation. Notably, the cooperative behavior is proposed to produce MVS dynamics that maintain the satisfaction of individual vehicle goals, while solving the merging conflict. Thanks to the altruistic passing sequence selection strategy that permits to exclude the passing sequence that do not meet the vehicle's individual goals.

The performance of the proposed multi-behavior decision-making strategy was evaluated through simulation scenarios, where both of the nominal and the cooperative behavior have been activated.
General conclusion and future works

General conclusion

The research work presented in this PhD manuscript delves into the challenging yet auspicious domain of Multi-Vehicle System (MVS) in the context of autonomous transportation. The overarching aim of the study has been to address the challenges associated with MVS coordination in complex and dynamic scenarios, particularly focusing on on-ramp merging and navigation in formation on highway.

The investigation into the context highlighted both the benefits and challenges posed by the widespread use of transportation systems. Notably, human errors have been identified as the main cause of accidents, motivating the exploration of autonomous vehicles (AVs) technology as a promising solution. The focus shifted towards MVS, in fact, the identified challenging scenarios require a motion coordination ability that the AVs do not explicitly include part of their abilities. This AVs limitation sets the stage for the presentation of the MVS paradigm, emphasizing the recognition of their potential to enhance safety, passenger comfort, and energy efficiency in various scenarios.

The conceptual idea behind this PhD work is to take advantage of the MVS cooperative ability to tackle the challenging scenario of on-ramp merging on highway. In other terms, the formation control concept part of the MVS paradigm was used to synchronize the vehicles' motion during the merging, while satisfying both of the individual vehicles' objectives and the MVS overarching goal. Consequently, threefold objectives were identified: (1) Formalizing the problem of MVS on-ramp merging on dense highway, (2) Developing navigation strategies for MVS in formation suited for the tackled scenario, and (3) Design a cooperative and altruistic decision-making level

that promotes the MVS advantages. These objectives collectively led to the design of the proposed Cooperative Multi-Controller Architecture (C-MCA).

The C-MCA, originally inspired from the foundational multi-controller architecture, focuses on both of the decision-making and the planning levels to mitigate the challenges related to MVS on-ramp merging. The decisionmaking level is based on the proposed multi-behavior decision-making strategy. This strategy comprises a nominal behavior, crafted for executing the merging maneuver while ensuring the vehicle's individual goal. The latter is activated when the generated vehicle dynamics satisfy the safety requirement. Otherwise, the C-MCA takes the responsibility of solving the merging conflict. To this aim, the altruistic passing sequence selection strategy is proposed, designed to satisfy both of the individual vehicles' goals and the MVS overarching goal, it decides on the MVS vehicles' passing sequence in the merging zone. This decision is motivated through the prism of safety, passengers' comfort, and energy efficiency.

Under the planning level, both of the C-MCA main behaviors have their assigned controller, designed to meet their specifications. For instance, the cooperative behavior controller's goal is to translate the passing sequence to the MVS vehicles' dynamics. The translation is based on the dynamic formation reconfiguration approach. The first proposed reconfiguration approach, named Constrained Optimal Reconfiguration Matrix (CORM), rooted on the virtual structure approach for formation modeling, and the proposed constrained inter-target distance matrix for formation control, which has several limitations, mainly in highly dynamic and structure environment. These limitations were the motivation of the second formation reconfiguration approach, named Extended Constrained Optimal Reconfiguration Matrix (E-CORM). The E-CORM mitigates the CORM's limited flexibility with the help of the proposed trajectory segmentation strategy, but still depends on a time-consuming optimization algorithm that does not align with the objectives of this work. Consequently, the third proposed approach, Formation Reconfiguration Approach based on Online Control Strategy (FRA-OCS), took advantages from both of the CORM and E-CORM to online control the formation reconfiguration according to the selected passing sequence.

Perspectives and future works

The diverse contributions presented in this PhD manuscript could lead to the introduction of a novel approach for understanding MVS navigation in intricate scenarios requiring motion synchronization, notably in situations such as on-ramp merging on highways. Consequently, the developments made in this PhD work are expected to facilitate the exploration and expansion of several research areas within the realms of MVS navigation in complex scenarios, formation control, and multi-criteria decision-making. The subsequent section provides a concise overview of the key research endeavors we intend to undertake in the near future.

Extend the cooperative multi-controller architecture to meet the robustness requirement

Our primary focus was on addressing the challenge of MVS safely navigating on-ramp merging on highways. To tackle this, we developed the C-MCA, drawing inspiration from the AV control architecture and the foundational multi-controller architecture. Our objective was to emphasize decision-making and planning levels to effectively handle the specific scenario. Enhancing the robustness of the C-MCA emerged as a crucial aspect in our effort to improve its genericity, with two key factors demanding attention: uncertainty and communication-related issues.

Robustness entails constructing a control architecture capable of handling multiple sources of uncertainty, whether preexisting or introduced during processing. Uncertainty may arise from various origins, such as local or distant sources of uncertainty, regarding for instance sensor observations in the AV or uncertainty in the environmental scenario communicated by peer vehicles. These uncertainties persist not only when addressing individual vehicles but also when dealing with a group of vehicles represented by the MVS. Consequently, accounting for uncertainty becomes imperative in the decision-making process. Establishing a decision-making layer capable of, (i) estimating the current safety state considering uncertainty, (ii) assessing the risk of future situations and propagating uncertainty, and (iii) conducting risk assessment and management to devise a fail-safe strategy when both nominal and cooperative behaviors fall short of ensuring the safety requirement.

Furthermore, the robustness of the control architecture is closely linked to its dependency to communication. Communication introduces challenges such as delays in sending and receiving information, packet losses, and potential cyber attacks. To mitigate these communication-related issues, a prudent approach involves minimizing the control architecture's reliance on communication. When constructing a control architecture that utilizes communication, efforts should be made to minimize it to the bare minimum, employing tools like architecture distribution and predictions of the vehicle behavior using approaches such as game theory.

Improve the metrics part of the multi-criteria decision-making level

To determine an optimal passing sequence for the MVS within the merging zone, the decision-making level employs a multi-criteria objective function. This function's assessment relies on three sub-criteria: safety, evaluated using the Euclidean distance; passenger comfort, assessed through acceleration changes; and energy efficiency, estimated by considering both acceleration and kinetic energy.

In the context of enhancing the C-MCA efficiency, an avenue worth exploring involves refining the metrics used to evaluate the multi-criteria objective function. Specifically, for safety evaluation, adopting the Extended Time-To-Collision (E-TTC) [116] metric can be beneficial. E-TTC provides comprehensive information on safety between two vehicles by incorporating not only the relative Euclidean distance but also information about relative velocity.

Regarding energy estimation, an enhanced energy consumption estimator based on the Electrical Vehicle (EV) model was developed ³. This estimator leverages a detailed low-level modeling of the EV to generate an estimation of the state of charge variable during normalized velocity cycles (e.g., NEDC, WLTP). Unfortunately, due to time constraints, the results of this improved estimator could not be incorporated into the findings if this PhD manuscript.

Extend the contributions to other scene representations

The following potential use-cases have been identified for investigation to validate the versatility of the proposed C-MCA:

³The improved energy consumption estimator was developed by L. Midelet, L. Saidi, L. Adouane and R. Talj

- Multi-lane CACC: the proposed C-MCA was successfully used in order to perform platoon navigation. One extension of this application is by tackling multi-lane formation navigation (multi-lane CACC).
- Intersection crossing: in addition to on-ramp merging, intersections are also considered as challenging scenarios in the literature. In fact, intersections and on-ramp merging have many similarities such as overlapping trajectories and bottlenecks caused by idling vehicles. Thus, motion coordination through the passing sequence selection can be considered to overcome the intersection crossing challenges.

Simulation and experimentation

The majority of the proposed methodologies have been substantiated primarily through extensive simulation work. Consequently, there is a compelling need to transition the proposed Formation Reconfiguration Approach base on an Online Control Strategy and the multi-behavior decisionmaking from simulated environments to practical implementations on multiple real-world vehicles or within a large-scale simulation grounded in realworld traffic data. This transition presents a host of technical challenges, including ensuring the reliable implementation of software components and addressing the functional safety concerns inherent to automobile operating systems.

Moreover, uncertainties surround these hierarchical layers, each comprising diverse components for collaborative functionality. Executing scenarios resembling on-ramp situations, even without the high dynamics' characteristic of highway scenarios, necessitate either a controlled test environment equipped with an on-ramp or the legal authorization to use public on-ramps. For a more pragmatic approach, research could be conducted using smallscale vehicles within a confined highway environment. As highlighted in the state-of-the-art (Chapter 2), real-world experiments involving MVS technology particularly and AV globally remain a persisting challenge.

Appendices

Appendix A: The Road Side Unit (RSU)

According to [134], the RSU architecture is composed of two main modules, the perception module and the communication module.



Figure .18. Road Side Unit (RSU)

- (a) The RSU perception module: It is equipped with an array of sensors such as cameras, RADAR, and LiDAR. The RSU significantly extends the range and reliability of perception compared to relying solely on the on-board sensors of the individual vehicles. This augmentation of perception capabilities plays a pivotal role in enhancing the overall safety and efficiency of the MVS.

For instance, in [60], the authors introduced a novel application of the RSU involving a 360-degree LiDAR, transforming the RSU into a LiDAR-enhanced infrastructure. Similarly, in [156], a collaborative map-aided tracking system was proposed, combining an on-board Li-DAR sensor and a road side vision system. This fusion of sensor data allowed for the broadcasting of detected objects, contributing to an improved situation awareness. Moreover, the work in [99], viewed the sensor infrastructure as a valuable tool for supporting various vehicles activities. This could involve enhancing agent-based vision systems or optimizing traffic management through more established motion coordination.

 (b) The RSU communication module: It utilizes stationary perception sensors that gather object detection and tracking data, and through a data fusion process, it derives detailed object characteristics such as position, heading, and speed for each detected object. This capability allows the RSU to provide rich and comprehensive information about the detected object.

Additionally, the RSU employs the V2X communication technologies as discussed in section 3.3.1.2 to broadcast the objects data as in Figure .18. Presently, V2X systems predominantly rely on two technologies: Dedicated Short-Range Communications (DSRC) and cellular-based V2X technology. These communication mechanisms enable the RSU to exchange critical information with nearby vehicle (cf. Figure .18); within the communication range of the communication devices, facilitating enhanced situational awareness and coordination with the MVS.

Appendix B: Safety insurance strategy part of the inter-target distance matrix

The inter-target error $e_{f_{23}}$ is defined as given in [241]:

$$e_{f_{23}} = f_2 - f_3$$

= $-f_2^{end} + f_2 + f_3^{end} - f_3 + f_2^{end} - f_3^{end}$
= $-e_{f_2} + e_{f_3} + e_{f_{23}}^{end}$ (17)

The inter-target distance can be expressed by:

$$d_T^2 = e_{f_{23}}^T e_{f_{23}} \tag{18}$$

eq. 18 is derivated in order to obtain the minimum inter-target distance:

$$\frac{d(d_T^2)}{dt} = 0 \tag{19}$$

eq. 18 is replaced in eq. 19:

$$2e_{f_{23}}^T \dot{e}_{f_{23}} = 0 \tag{20}$$

eq. 20 can be written using the derivative of eq. 17 and eq. 4.5 as follows:

$$e_{f_{23}}^{T} \dot{e}_{f_{23}} = 0$$

$$e_{f_{23}}^{T} [\dot{e}_{f_{2}} - \dot{e}_{f_{3}}] = 0$$

$$e_{f_{23}}^{T} [(a_{2} + a_{23})e_{f_{2}} - (a_{3} - a_{23})e_{f_{3}}] = 0$$
(21)

Using eq. 17 in eq. 21, it is obtained:

$$e_{f_{23}}^{T}[(a_{2}+a_{23})e_{f_{2}}-(a_{3}-a_{23})(e_{f_{2}}+e_{f_{23}}-e_{f_{23}}^{end})]=0$$

$$e_{f_{23}}^{T}[(a_{2}-a_{3}+a_{23})e_{f_{2}}+(a_{23}-a_{3})e_{f_{23}}-(a_{23}-a_{3})e_{f_{23}}^{end}]=0$$
 (22)

Using eq. 19 and eq. 22, the condition linking the minimum inter-target distance \underline{D}_T and gains of the reconfiguration matrix A can be formulated by:

$$e_{f_{23}}^{T} \left[\frac{a_2 - a_3 + a_{23}}{a_3 - a_{23}} e_{f_2} + e_{f_{23}}^{end} \right] \ge \underline{D}_T^2$$
(23)

Thus, the values of a_2 , a_3 , and a_{23} must be chosen to satisfy eq. 23.

Appendix C: Simulation tools

Developing an MVS necessitates utilizing accurate simulation tools. These tools typically provide confident vehicle models and environments, which are well-suited for analyzing, validating, and optimizing autonomous systems. In [222], the authors conducted a comprehensive review of the latest advancements and analyses concerning simulation tools.

The algorithms proposed in this PhD study underwent testing within a simulation environment. Specifically, the decision, planning, and control modules of the C-MCA framework were developed using Matlab (cf. Figure .19), with the simulation executed through Simulink.

Initially, the Unreal Engine 3D was employed to recreate the on-ramp merging scenario. To bridge the gap between Simulink and the Unreal Engine, a connecting interface was established, facilitating the integration of the simulation and 3D components, such as vehicle motion. In the Unreal Engine tier (see Figure .19), the 3D model and environment utilized for the simulation were constructed based on the Unreal Engine platform. The simulation environment represents a response to the on-ramp merging scenario within the module, with virtual vehicles comprising models of the vehicles and sensor modules. Lastly, the bridge module facilitates communication between Simulink and the Unreal Engine tier.



Figure .19. The overall simulation architecture

The virtual vehicle module within Unreal Engine employs a kinematic vehi-

cle model, which represents a significant limitation. To address this drawback, SCANeR studio was utilized. Besides its integration of an advanced dynamic model of the vehicle. SCANeR stands out as one of the most comprehensive solutions for automotive simulation, enabling users to configure, prepare, execute simulations, and analyze outcomes effectively. Its versatility proved especially valuable for scenarios involving multiple vehicles, such as MVS. Integration of the SCANeR studio simulation solution followed the same architectural framework as previously depicted (refer to Figure .20).



Figure .20. The overall simulation architecture

Bibliography

- [1] The european automobile manufacturers association. url https://www.acea.auto.
- [2] The guidehouse insight. url https://guidehouseinsights.com.
- [3] How many cars are there in the world in 2023? url https://hedgescompany.com/blog/2021/06/how-many-cars-are-there-in-the-world/: :text=CLICK
- [4] Loes Aarts and Gerben Feddes. European truck platooning challenge. In Proceedings of the HVTT14: International Symposium on Heavy Vehicle Transport Technology, Rotorua, New Zealand, pages 15–18, 2016.
- [5] Khadige Abboud, Hassan Aboubakr Omar, and Weihua Zhuang. Interworking of dsrc and cellular network technologies for v2x communications: A survey. *IEEE transactions on vehicular technology*, 65(12):9457–9470, 2016.
- [6] Dhaminda B. Abeywickrama, Marco Mamei, and Franco Zambonelli. Engineering collectives of self-driving vehicles: The sota approach. In Tiziana Margaria and Bernhard Steffen, editors, *Leveraging Appli*cations of Formal Methods, Verification and Validation. Distributed Systems, pages 79–93, Cham, 2018. Springer International Publishing.
- [7] Sunil Adhau, Murali Lal Mittal, and Abhinav Mittal. A multi-agent system for distributed multi-project scheduling: An auction-based negotiation approach. *Engineering Applications of Artificial Intelligence*, 25(8):1738–1751, 2012.
- [8] National Highway Traffic Safety Administration. Summary report: Standing general order on crash reporting for automated driving systems. National Highway Traffic Safety Administration, pages 1–9, 2022.

- [9] Muhamad Hariz Muhamad Adnan, Mohd Fadzil Hassan, Izzatdin Aziz, and Irving V Paputungan. Protocols for agent-based autonomous negotiations: A review. In 2016 3rd International Conference on Computer and information sciences (ICCOINS), pages 622–626. IEEE, 2016.
- [10] Lounis Adouane. Architectures de Contrôle Comportementales et Réactives pour la Coopération d'un groupe de Robots Mobiles. PhD thesis, Université de Franche-Comté, 2005.
- [11] Lounis Adouane. Autonomous vehicle navigation: from behavioral to hybrid multi-controller architectures. CRC Press, 2016.
- [12] Lounis Adouane. Toward fully autonomous vehicle navigation: From behavioral to hybrid multi-controller architectures. In 2017 11th International Workshop on Robot Motion and Control (RoMoCo), pages 85–98. IEEE, 2017.
- [13] Lounis Adouane and Nadine Le Fort-Piat. Behavioral and distributed architecture of control for minimalist mobile robots. *Journal européen* des systèmes automatisés, 40(2):177–196, 2006.
- [14] Manzoor Ahmed, Muhammad Ayzed Mirza, Salman Raza, Haseeb Ahmad, Fang Xu, Wali Ullah Khan, Qing Lin, and Zhu Han. Vehicular communication network enabled cav data offloading: A review. *IEEE Transactions on Intelligent Transportation Systems*, 2023.
- [15] Assad Alam, Bart Besselink, Valerio Turri, Jonas Mårtensson, and Karl H Johansson. Heavy-duty vehicle platooning for sustainable freight transportation: A cooperative method to enhance safety and efficiency. *IEEE Control Systems Magazine*, 35(6):34–56, 2015.
- [16] Rachid Alami, Raja Chatila, Sara Fleury, Malik Ghallab, and Félix Ingrand. An architecture for autonomy. *The International Journal* of Robotics Research, 17(4):315–337, 1998.
- [17] Mani Amoozadeh, Hui Deng, Chen-Nee Chuah, H Michael Zhang, and Dipak Ghosal. Platoon management with cooperative adaptive cruise control enabled by vanet. *Vehicular communications*, 2(2):110– 123, 2015.
- [18] Shunsuke Aoki and Ragunathan Rajkumar. A merging protocol for self-driving vehicles. In 2017 ACM/IEEE 8th International Conference on Cyber-Physical Systems (ICCPS), pages 219–228. IEEE, 2017.

- [19] Maytheewat Aramrattana, Jérôme Detournay, Cristofer Englund, Viktor Frimodig, Oscar Uddman Jansson, Tony Larsson, Wojciech Mostowski, Víctor Díez Rodríguez, Thomas Rosenstatter, and Golam Shahanoor. Team halmstad approach to cooperative driving in the grand cooperative driving challenge 2016. *IEEE Transactions on Intelligent Transportation Systems*, 19(4):1248–1261, 2018.
- [20] Ronald C Arkin. Behavior-based robotics. MIT press, 1998.
- [21] Ramin Arvin, Asad J Khattak, Mohsen Kamrani, and Jackeline Rio-Torres. Safety evaluation of connected and automated vehicles in mixed traffic with conventional vehicles at intersections. *Journal of Intelligent Transportation Systems*, 25(2):170–187, 2020.
- [22] Mohamad Ali Assaad. An overview on systems of systems control: general discussions and application to multiple autonomous vehicles. PhD thesis, Université de Technologie de Compiègne, 2019.
- [23] Mohamad Ali Assaad, Reine Talj, and Ali Charara. Cooperative lateral maneuvers manager for multi-autonomous vehicles. In 2018 IEEE International Conference on Systems, Man, and Cybernetics (SMC), pages 2651–2656. IEEE, 2018.
- [24] Mohamad Ali Assaad, Reine Talj, and Ali Charara. A system of systems framework: Cooperative maneuvers manager for autonomous vehicles. In 2018 13th Annual Conference on System of Systems Engineering (SoSE), pages 92–97. IEEE, 2018.
- [25] Mohamad Ali Assaad, Reine Talj, and Ali Charara. Autonomous driving as system of systems: roadmap for accelerating development. In 2019 14th Annual Conference System of Systems Engineering (SoSE), pages 102–107. IEEE, 2019.
- [26] 5G Automotive Association. A visionary roadmap for advanced driving use cases, connectivity technologies, and radio spectrum needs. https://5gaa.org/content/uploads/2020/09/A-Visionary-Roadmap-for-Advanced-Driving-Use-Cases-Connectivity-Technologies-and-Radio-Spectrum-Needs.pdf.
- [27] Pierre Avanzini, Benoît Thuilot, and Philippe Martinet. Urban vehicle platoon using monocular vision: Scale factor estimation. In 2010 11th International Conference on Control Automation Robotics & Vision, pages 1803–1808. IEEE, 2010.

- [28] Tanveer Awal, Lars Kulik, and Kotagiri Ramamohanrao. Optimal traffic merging strategy for communication-and sensor-enabled vehicles. In 16th International IEEE Conference on Intelligent Transportation Systems (ITSC 2013), pages 1468–1474. IEEE, 2013.
- [29] Claudine Badue, Rânik Guidolini, Raphael Vivacqua Carneiro, Pedro Azevedo, Vinicius B. Cardoso, Avelino Forechi, Luan Jesus, Rodrigo Berriel, Thiago M. Paixão, Filipe Mutz, Lucas de Paula Veronese, Thiago Oliveira-Santos, and Alberto F. De Souza. Self-driving cars: A survey. Expert Systems with Applications, 165:113816, 2021.
- [30] Saeed Asadi Bagloee, Madjid Tavana, Mohsen Asadi, and Tracey Oliver. Autonomous vehicles: challenges, opportunities, and future implications for transportation policies. *Journal of modern transportation*, 24:284–303, 2016.
- [31] Alexander Bahr, Matthew R Walter, and John J Leonard. Consistent cooperative localization. In 2009 IEEE International Conference on Robotics and Automation, pages 3415–3422. IEEE, 2009.
- [32] Ali Balador, Alessandro Bazzi, Unai Hernandez-Jayo, Idoia de la Iglesia, and Hossein Ahmadvand. A survey on vehicular communication for cooperative truck platooning application. *Vehicular Communications*, 35:100460, 2022.
- [33] T. Balch and R.C. Arkin. Behavior-based formation control for multirobot teams. *IEEE Transactions on Robotics and Automation*, 14(6):926–939, 1998.
- [34] Behzad Bayat, Naveena Crasta, Alessandro Crespi, António M Pascoal, and Auke Ijspeert. Environmental monitoring using autonomous vehicles: a survey of recent searching techniques. *Current opinion in biotechnology*, 45:76–84, 2017.
- [35] Ahmed Benzerrouk. Architecture de contrôle hybride pour systèmes multi-robots mobiles. Theses, Université Blaise Pascal - Clermont-Ferrand II, April 2011.
- [36] Ahmed Benzerrouk, Lounis Adouane, Laurent Lequievre, and Philippe Martinet. Navigation of multi-robot formation in unstructured environment using dynamical virtual structures. In 2010 IEEE/RSJ International Conference on Intelligent Robots and Systems, pages 5589–5594. IEEE, 2010.

- [37] Ahmed Benzerrouk, Lounis Adouane, and Philippe Martinet. Stable navigation in formation for a multi-robot system based on a constrained virtual structure. *Robotics and Autonomous Systems (RAS)*, 62(12):1806 – 1815, 2014.
- [38] Carl Bergenhem, Qihui Huang, Ahmed Benmimoun, and Tom Robinson. Challenges of platooning on public motorways. In 17th world congress on intelligent transport systems, pages 1–12, 2010.
- [39] Carl Bergenhem, Steven Shladover, Erik Coelingh, Christoffer Englund, and Sadayuki Tsugawa. Overview of platooning systems. In Proceedings of the 19th ITS World Congress, Oct 22-26, Vienna, Austria (2012), 2012.
- [40] Vincent L Bernardin Jr, Theodore Mansfield, Benjamin Swanson, Hadi Sadrsadat, and Sumit Bindra. Scenario modeling of autonomous vehicles with trip-based models. *Transportation Research Record*, 2673(10):261–270, 2019.
- [41] David Bevly, Xiaolong Cao, Mikhail Gordon, Guchan Ozbilgin, David Kari, Brently Nelson, Jonathan Woodruff, Matthew Barth, Chase Murray, Arda Kurt, et al. Lane change and merge maneuvers for connected and automated vehicles: A survey. *IEEE Transactions on Intelligent Vehicles*, 1(1):105–120, 2016.
- [42] Youssef Bichiou and Hesham Rakha. Vehicle platooning: An energy consumption perspective. In 2020 IEEE 23rd International Conference on Intelligent Transportation Systems (ITSC), pages 1–6. IEEE, 2020.
- [43] Nicola Bicocchi, Marco Mamei, Andrea Sassi, and Franco Zambonelli. On recommending opportunistic rides. *IEEE Transactions on Intelli*gent Transportation Systems, PP:1–11, 04 2017.
- [44] Jeremy Blum and Azim Eskandarian. The threat of intelligent collisions. IT professional, 6(1):24–29, 2004.
- [45] Martin Buehler, Karl Iagnemma, and Sanjiv Singh. The DARPA Urban Challenge: Autonomous Vehicles in City Traffic. Springer Publishing Company, Incorporated, 1st edition, 2009.
- [46] Giacomo Cabri, Luca Gherardini, and Manuela Montangero. Auctionbased crossings management. In Proceedings of the 5th EAI International Conference on Smart Objects and Technologies for Social Good, pages 183–188, 2019.

- [47] David Camacho, Fernando Fernández, and Miguel A Rodelgo. Roboskeleton: An architecture for coordinating robot soccer agents. *Engineering Applications of Artificial Intelligence*, 19(2):179–188, 2006.
- [48] Y Uny Cao, Andrew B Kahng, and Alex S Fukunaga. Cooperative mobile robotics: Antecedents and directions. *Robot colonies*, pages 7–27, 1997.
- [49] Jason Carbaugh, Datta N Godbole, and Raja Sengupta. Safety and capacity analysis of automated and manual highway systems. *Transportation Research Part C: Emerging Technologies*, 6(1-2):69– 99, 1998.
- [50] CARLA. Carla open-source simulator for autonomous driving research. https://carla.org.
- [51] Dustin Carlino, Stephen D Boyles, and Peter Stone. Auction-based autonomous intersection management. In 16th International IEEE Conference on Intelligent Transportation Systems (ITSC 2013), pages 529–534. IEEE, 2013.
- [52] CarSim. Carsim: Mechanical simulation. https://www.carsim.com/ products/carsim/index.php.
- [53] Constantin F Caruntu, Catalin Braescu, Anca Maxim, Razvan C Rafaila, and Alexandru Tiganasu. Distributed model predictive control for vehicle platooning: A brief survey. In 2016 20th International Conference on System Theory, Control and Computing (ICSTCC), pages 644–650. IEEE, 2016.
- [54] Linguo Chai, Baigen Cai, Wei ShangGuan, and Jian Wang. Connected and autonomous vehicles coordinating method at intersection utilizing preassigned slots. In 2017 IEEE 20th international conference on intelligent transportation systems (ITSC), pages 1–6. IEEE, 2017.
- [55] Eric Chan. Sartre automated platooning vehicles. *Towards innovative freight and logistics*, 2:137–150, 2016.
- [56] Eric Chan, Peter Gilhead, Pavel Jelinek, Petr Krejci, and Tom Robinson. Cooperative control of sartre automated platoon vehicles. In 19th its world congressertico-its europeeuropean commissionits americaits asia-pacific, 2012.

- [57] Chih-Fu Chang and Li-Chen Fu. A formation control framework based on lyapunov approach. In 2008 IEEE/RSJ International Conference on Intelligent Robots and Systems, pages 2777–2782, 2008.
- [58] Alia Chebly, Reine Talj, and Ali Charara. Coupled longitudinal/lateral controllers for autonomous vehicles navigation, with experimental validation. *Control Engineering Practice*, 88:79–96, 2019.
- [59] Danjue Chen, Soyoung Ahn, Madhav Chitturi, and David A Noyce. Towards vehicle automation: Roadway capacity formulation for traffic mixed with regular and automated vehicles. *Transportation research* part B: methodological, 100:196–221, 2017.
- [60] Jingrong Chen, Sheng Tian, Hao Xu, Rui Yue, Yuan Sun, and Yuepeng Cui. Architecture of vehicle trajectories extraction with roadside lidar serving connected vehicles. *Ieee Access*, 7:100406–100415, 2019.
- [61] Lei Chen and Cristofer Englund. Cooperative intersection management: A survey. *IEEE transactions on intelligent transportation sys*tems, 17(2):570–586, 2015.
- [62] Yang Quan Chen and Zhongmin Wang. Formation control: a review and a new consideration. In 2005 IEEE/RSJ International conference on intelligent robots and systems, pages 3181–3186. IEEE, 2005.
- [63] Fang-Chieh Chou, Steven E Shladover, and Gaurav Bansal. Coordinated merge control based on v2v communication. In 2016 IEEE Vehicular Networking Conference (VNC), pages 1–8. IEEE, 2016.
- [64] Erik Coelingh, Jonas Nilsson, and Jude Buffum. Driving tests for self-driving cars. *IEEE Spectrum*, 55:40–45, 2018.
- [65] Jean-Michel Contet, Franck Gechter, Pablo Gruer, and Abderrafiaa Koukam. Bending Virtual Spring-Damper: A Solution to Improve Local Platoon Control. In David Hutchison, Takeo Kanade, Josef Kittler, Jon M. Kleinberg, Friedemann Mattern, John C. Mitchell, Moni Naor, Oscar Nierstrasz, C. Pandu Rangan, Bernhard Steffen, Madhu Sudan, Demetri Terzopoulos, Doug Tygar, Moshe Y. Vardi, Gerhard Weikum, Gabrielle Allen, Jarosław Nabrzyski, Edward Seidel, Geert Dick van Albada, Jack Dongarra, and Peter M. A. Sloot, editors, *Computational Science – ICCS 2009*, volume 5544, pages 601–610. Springer Berlin Heidelberg, Berlin, Heidelberg, 2009.

- [66] Jin Cui, Lin Shen Liew, Giedre Sabaliauskaite, and Fengjun Zhou. A review on safety failures, security attacks, and available countermeasures for autonomous vehicles. Ad Hoc Networks, 90:101823, 2019.
- [67] Rongxin Cui, Ji Guo, and Bo Gao. Game theory-based negotiation for multiple robots task allocation. *Robotica*, 31(6):923–934, 2013.
- [68] Issam W Damaj, Dina K Serhal, Lama A Hamandi, Rached N Zantout, and Hussein T Mouftah. Connected and autonomous electric vehicles: Quality of experience survey and taxonomy. *Vehicular Communications*, 28:100312, 2021.
- [69] Alaa Daoud, Gauthier Picard, Hiba Alqasir, Paolo Gianessi, and Flavien Balbo. Communication-wise comparison of the online resource allocation methods in cav fleets. *Procedia Computer Science*, 220:299–306, 2023.
- [70] Trevor Darrell, Thomas West, and all. California path program biennial report 2018-2020. In University of California, Berkeley Institute of Transportation Studies, 2020.
- [71] Schrank David, Eisele Bill, Lomax Tim, and Bak Jim. 2015 urban mobility scorecard. *Texas AM Transportation Institute*, pages 1–82, 2015.
- [72] Kakan C Dey, Li Yan, Xujie Wang, Yue Wang, Haiying Shen, Mashrur Chowdhury, Lei Yu, Chenxi Qiu, and Vivekgautham Soundararaj. A review of communication, driver characteristics, and controls aspects of cooperative adaptive cruise control (cacc). *IEEE Transactions on Intelligent Transportation Systems*, 17(2):491–509, 2015.
- [73] Mario di Bernardo, Alessandro Salvi, and Stefania Santini. Distributed consensus strategy for platooning of vehicles in the presence of time-varying heterogeneous communication delays. *IEEE Transactions on Intelligent Transportation Systems*, 16(1):102–112, 2015.
- [74] Nelson L Diaz, Adriana Carolina Luna, Juan C Vasquez, and Josep M Guerrero. Centralized control architecture for coordination of distributed renewable generation and energy storage in islanded ac microgrids. *IEEE Transactions on Power Electronics*, 32(7):5202–5213, 2016.
- [75] Jishiyu Ding, Li Li, Huei Peng, and Yi Zhang. A rule-based cooperative merging strategy for connected and automated vehicles. *IEEE*

Transactions on Intelligent Transportation Systems, 21(8):3436–3446, 2019.

- [76] Jing Dong, Erik Nelson, Vadim Indelman, Nathan Michael, and Frank Dellaert. Distributed real-time cooperative localization and mapping using an uncertainty-aware expectation maximization approach. In 2015 IEEE international conference on robotics and automation (ICRA), pages 5807–5814. IEEE, 2015.
- [77] Kurt Dresner and Peter Stone. Multiagent traffic management: A reservation-based intersection control mechanism. In Autonomous Agents and Multiagent Systems, International Joint Conference on, volume 3, pages 530–537. Citeseer, 2004.
- [78] Kevin Duan, Chris McDaniel, Amanda Muller, Brett Yokota, and Jan Kleissl. Effects of highway slipstreaming on california gas consumption, 2007.
- [79] Fábio Duarte and Carlo Ratti. The impact of autonomous vehicles on cities: A review. *Journal of Urban Technology*, 25(4):3–18, 2018.
- [80] Amandine Duboz, Andromachi Mourtzouchou, Monica Grosso, Viktoriya Kolarova, Rubén Cordera, Sophie Nägele, Maria Alonso Raposo, Jette Krause, Ada Garus, Christine Eisenmann, et al. Exploring the acceptance of connected and automated vehicles: focus group discussions with experts and non-experts in transport. *Transportation* research part F: traffic psychology and behaviour, 89:200–221, 2022.
- [81] Arturo Dávila and Mario Nombela. Sartre safe road trains for the environment reducing fuel consumption through lower aerodynamic drag coefficient. In *SAE Brasil 2011 Congress and Exhibit.* SAE International, oct 2011.
- [82] Madeleine El Zaher, Jean-Michel Contet, Franck Gechter, and Abderrafiaa Koukam. Echelon platoon organisation: A distributed approach based on 2-spring virtual links. In Artificial Intelligence: Methodology, Systems, and Applications: 15th International Conference, AIMSA 2012, Varna, Bulgaria, September 12-15, 2012. Proceedings 15, pages 246–255. Springer, 2012.
- [83] David Elliott, Walter Keen, and Lei Miao. Recent advances in connected and automated vehicles. *journal of traffic and transportation* engineering (English edition), 6(2):109–131, 2019.

- [84] Stanford ENGINEERING. Junior takes runner-up and a place in technology history. https://engineering.stanford.edu/news/ junior-takes-runner-and-place-technology-history.
- [85] Cristofer Englund, Lei Chen, Jeroen Ploeg, Elham Semsar-Kazerooni, Alexey Voronov, Hoai Hoang Bengtsson, and Jonas Didoff. The grand cooperative driving challenge 2016: boosting the introduction of cooperative automated vehicles. *IEEE Wireless Communications*, 23(4):146–152, 2016.
- [86] Azim Eskandarian, Chaoxian Wu, and Chuanyang Sun. Research advances and challenges of autonomous and connected ground vehicles. *IEEE Transactions on Intelligent Transportation Systems*, 22(2):683– 711, 2019.
- [87] Niclas Evestedt, Erik Ward, John Folkesson, and Daniel Axehill. Interaction aware trajectory planning for merge scenarios in congested traffic situations. In 2016 IEEE 19th International Conference on Intelligent Transportation Systems (ITSC), pages 465–472. IEEE, 2016.
- [88] Asif Faisal, Md Kamruzzaman, Tan Yigitcanlar, and Graham Currie. Understanding autonomous vehicles. *Journal of transport and land use*, 12(1):45–72, 2019.
- [89] John T Feddema, Chris Lewis, and David A Schoenwald. Decentralized control of cooperative robotic vehicles: theory and application. *IEEE Transactions on robotics and automation*, 18(5):852–864, 2002.
- [90] Shuo Feng, Yi Zhang, Shengbo Eben Li, Zhong Cao, Henry X Liu, and Li Li. String stability for vehicular platoon control: Definitions and analysis methods. *Annual Reviews in Control*, 47:81–97, 2019.
- [91] NHTSA's National Center for Statistics and Analysis. 2016 fatal motor vehicle crashes: Overview. https://crashstats.nhtsa.dot. gov/Api/Public/ViewPublication/812456.
- [92] Alexander Frötscher and Bernhard Monschiebl. C-roads: Elements of c-its service evaluation to reach interoperability in europe within a wide stakeholder network: Validation steps and comparative elements used in a living lab environment in austria. pages 1–5, 06 2018.
- [93] Takehiko Fujioka. Its demonstration by the university of tokyo team at demo 2000. *KOYO ENGINEERING JOURNAL*, pages 3–9, 2002.

- [94] Chen Gao, Xiao He, Hongli Dong, Hongjian Liu, and Guangran Lyu. A survey on fault-tolerant consensus control of multi-agent systems: trends, methodologies and prospects. *International Journal of Sys*tems Science, 53(13):2800–2813, 2022.
- [95] Huan Gao, Yanqing Cen, Bo Liu, Xianghui Song, Hongben Liu, and Jia Liu. A collaborative merging method for connected and automated vehicle platoons in a freeway merging area with considerations for safety and efficiency. *Sensors*, 23(9), 2023.
- [96] Gazebo. Gazebo. https://gazebosim.org/home.
- [97] OLASUBOMI GBENJO. What is a connected car? https: //www.makeuseof.com/connected-cars-pros-and-cons/#:~: text=The%20Benefits%20of%20Connected%20Cars,relevant% 20data%20it%20picks%20up.
- [98] Andreas Geiger, Martin Lauer, Frank Moosmann, Benjamin Ranft, Holger Rapp, Christoph Stiller, and Julius Ziegler. Team annieway's entry to the 2011 grand cooperative driving challenge. *IEEE Transactions on Intelligent Transportation Systems*, 13(3):1008–1017, 2012.
- [99] Florian Geissler, Sören Kohnert, and Reinhard Stolle. Designing a roadside sensor infrastructure to support automated driving. In 2018 21st International Conference on Intelligent Transportation Systems (ITSC), pages 703–708. IEEE, 2018.
- [100] Amrita Ghosal and Mauro Conti. Security issues and challenges in v2x: A survey. *Computer Networks*, 169:107093, 2020.
- [101] Anouck Renée Girard, J Borges De Sousa, and J Karl Hedrick. An overview of emerging results in networked multi-vehicle systems. In *Proceedings of the 40th IEEE Conference on Decision and Control* (*Cat. No. 01CH37228*), volume 2, pages 1485–1490. IEEE, 2001.
- [102] Michael A Goodrich, Alan C Schultz, et al. Human-robot interaction: a survey. Foundations and Trends[®] in Human-Computer Interaction, 1(3):203-275, 2008.
- [103] G. Grilli. Data dissemination in vehicular networks. Ph.D. dissertation, Dept. Comput. Sci., Syst. Prod., Univ. Rome Tor Vergata, Rome, Italy, 2010.
- [104] Dominique Gruyer, Valentin Magnier, Karima Hamdi, Laurène Claussmann, Olivier Orfila, and Andry Rakotonirainy. Perception,

information processing and modeling: Critical stages for autonomous driving applications. *Annual Reviews in Control*, 44:323–341, 2017.

- [105] Jacopo Guanetti, Yeojun Kim, and Francesco Borrelli. Control of connected and automated vehicles: State of the art and future challenges. *Annual reviews in control*, 45:18–40, 2018.
- [106] Qiangqiang Guo, Li Li, and Xuegang Jeff Ban. Urban traffic signal control with connected and automated vehicles: A survey. Transportation research part C: emerging technologies, 101:313–334, 2019.
- [107] Levent Güvenç, Ismail Meriç Can Uygan, Kerim Kahraman, Raif Karaahmetoglu, Ilker Altay, Mutlu Sentürk, Mümin Tolga Emirler, Ahu Ece Hartavi Karci, Bilin Aksun Guvenc, Erdinç Altug, et al. Cooperative adaptive cruise control implementation of team mekar at the grand cooperative driving challenge. *IEEE transactions on intelligent transportation systems*, 13(3):1062–1074, 2012.
- [108] Sohan Gyawali, Shengjie Xu, Yi Qian, and Rose Qingyang Hu. Challenges and solutions for cellular based v2x communications. *IEEE Communications Surveys Tutorials*, 23(1):222–255, 2021.
- [109] Bernhard Häfner, Vaibhav Bajpai, Jörg Ott, and Georg A Schmitt. A survey on cooperative architectures and maneuvers for connected and automated vehicles. *IEEE Communications Surveys & Tutorials*, 24(1):380–403, 2021.
- [110] Saqib Hakak, Thippa Reddy Gadekallu, Praveen Kumar Reddy Maddikunta, Swarna Priya Ramu, M Parimala, Chamitha De Alwis, and Madhusanka Liyanage. Autonomous vehicles in 5g and beyond: A survey. Vehicular Communications, page 100551, 2022.
- [111] John Harding, Gregory Powell, Rebecca Yoon, Joshua Fikentscher, Charlene Doyle, Dana Sade, Mike Lukuc, Jim Simons, Jing Wang, et al. Vehicle-to-vehicle communications: readiness of v2v technology for application. Technical report, United States. National Highway Traffic Safety Administration, 2014.
- [112] Andreas Hegyi, Bart De Schutter, and Johannes Hellendoorn. Optimal coordination of variable speed limits to suppress shock waves. *IEEE Transactions on intelligent transportation systems*, 6(1):102– 112, 2005.

- [113] Bassem Hichri, Lounis Adouane, Jean-Christophe Fauroux, et al. Cooperative lifting and transport by a group of mobile robots. In *International Symposium on Distributed Autonomous Robotic Systems*, DARS 2014, 2014.
- [114] Yulong Huang, Yonggang Zhang, Bo Xu, Zhemin Wu, and Jonathon A Chambers. A new adaptive extended kalman filter for cooperative localization. *IEEE Transactions on Aerospace and Electronic Systems*, 54(1):353–368, 2017.
- [115] Bernhard Häfner, Vaibhav Bajpai, Jörg Ott, and Georg A. Schmitt. A Survey on Cooperative Architectures and Maneuvers for Connected and Automated Vehicles. *IEEE Communications Surveys & Tutorials*, 24(1):380–403, 2022.
- [116] Dimia Iberraken. Safe Trajectories and Sequential Bayesian Decision-Making Architecture for Reliable Autonomous Vehicle Navigation. Theses, Université Clermont Auvergne, October 2020.
- [117] SAE International. Autodrive challenge. https://www.sae.org/ attend/student-events/autodrive-challenge.
- [118] Sae International. Taxonomy and definitions for terms related to driving automation systems for on-road motor vehicles. SAE Int., 4970(724):1–5, 2018.
- [119] Superseding JA1012 JAN2002. Surface vehicle/aerospace recommended practice. 2010.
- [120] Dongyao Jia, Kejie Lu, Jianping Wang, Xiang Zhang, and Xuemin Shen. A survey on platoon-based vehicular cyber-physical systems. *IEEE communications surveys & tutorials*, 18(1):263–284, 2015.
- [121] Dongyao Jia, Kejie Lu, Jianping Wang, Xiang Zhang, and Xuemin Shen. A survey on platoon-based vehicular cyber-physical systems. *IEEE Communications Surveys Tutorials*, 18(1):263–284, 2016.
- [122] Huifu Jiang, Jia Hu, Shi An, Meng Wang, and Byungkyu Brian Park. Eco approaching at an isolated signalized intersection under partially connected and automated vehicles environment. *Transportation Re*search Part C: Emerging Technologies, 79:290–307, 2017.
- [123] Li Jin, Mladen Cicic, Karl H Johansson, and Saurabh Amin. Analysis and design of vehicle platooning operations on mixed-traffic highways. *IEEE Transactions on Automatic Control*, 2020.

- [124] Ronald K Jurgen. V2V/V2I communications for improved road safety and efficiency, volume 154. SAE International, 2012.
- [125] Georgios Karagiannis, Onur Altintas, Eylem Ekici, Geert Heijenk, Boangoat Jarupan, Kenneth Lin, and Timothy Weil. Vehicular networking: A survey and tutorial on requirements, architectures, challenges, standards and solutions. *IEEE communications surveys & tutorials*, 13(4):584–616, 2011.
- [126] Anna Karmańska. The benefits of connected vehicles within organizations. Procedia Computer Science, 192:4721–4731, 2021.
- [127] Prabhjot Kaur, Samira Taghavi, Zhaofeng Tian, and Weisong Shi. A survey on simulators for testing self-driving cars. In 2021 Fourth International Conference on Connected and Autonomous Driving (Metro-CAD), pages 62–70. IEEE, 2021.
- [128] Pooja Kavathekar and YangQuan Chen. Vehicle platooning: A brief survey and categorization. In International Design Engineering Technical Conferences and Computers and Information in Engineering Conference, volume 54808, pages 829–845, 2011.
- [129] Justin Kek. Alphabet's waymo: A technical study on autonomous vehicle tech. https://justinkek.medium.com/alphabets-waymo-atechnical-study-on-autonomous-vehicle-tech-c128180ab2c5.
- [130] John B Kenney. Dedicated short-range communications (dsrc) standards in the united states. *Proceedings of the IEEE*, 99(7):1162–1182, 2011.
- [131] Ahmed Khalifa, Olivier Kermorgant, Salvador Dominguez, and Philippe Martinet. Vehicles platooning in urban environment: Consensus-based longitudinal control with limited communications capabilities. In 2018 15th international conference on control, automation, robotics and vision (icarcv), pages 809–814. IEEE, 2018.
- [132] Ahmed Khalifa, Olivier Kermorgant, Salvador Dominguez, and Philippe Martinet. Vehicles platooning in urban environments: Integrated consensus-based longitudinal control with gap closure maneuvering and collision avoidance capabilities. In 2019 18th European Control Conference (ECC), pages 1695–1701. IEEE, 2019.
- [133] Ahmed Khalifa, Olivier Kermorgant, Salvador Dominguez, and Philippe Martinet. Platooning of car-like vehicles in urban environments: Longitudinal control considering actuator dynamics, time de-

lays, and limited communication capabilities. *IEEE Transactions on Control Systems Technology*, pages 1–8, 2020.

- [134] Zine El Abidine Kherroubi. Novel off-board decision-making strategy for connected and autonomous vehicles (Use case: Highway on-ramp merging).(Nouvelle stratégie de prise de décision débarquée pour les véhicules connectés et autonomes (Cas d'utilisation: insertion sur autoroute)). PhD thesis, Claude Bernard University Lyon 1, France, 2020.
- [135] Behrokh Khoshnevis and George Bekey. Centralized sensing and control of multiple mobile robots. Computers & industrial engineering, 35(3-4):503-506, 1998.
- [136] Seong-Woo Kim, Wei Liu, Marcelo H Ang, Emilio Frazzoli, and Daniela Rus. The impact of cooperative perception on decision making and planning of autonomous vehicles. *IEEE Intelligent Transportation Systems Magazine*, 7(3):39–50, 2015.
- [137] Gregor Klančar, Drago Matko, and Sašo Blažič. A control strategy for platoons of differential drive wheeled mobile robot. *Robotics and Autonomous Systems*, 59(2):57–64, 2011.
- [138] Jaeho Lee. Reactive-system approaches to agent architectures. In International Workshop on Agent Theories, Architectures, and Languages, pages 132–146. Springer, 1999.
- [139] Jaeho Lee, Marcus J Huber, Edmund H Durfee, and Patrick G Kenny. Um-prs: An implementation of the procedural reasoning system for multirobot applications. In NASA. Johnson Space Center, Conference on Intelligent Robotics in Field, Factory, Service and Space (CIRFFSS 1994), Volume 2, number AIAA PAPER 94-1297-CP, 1994.
- [140] Shengbo Eben Li, Yang Zheng, Keqiang Li, and Jianqiang Wang. An overview of vehicular platoon control under the four-component framework. In 2015 IEEE Intelligent Vehicles Symposium (IV), pages 286–291, 2015.
- [141] Yongfu Li, Zongyu Qin, Hao Zhu, Srinivas Peeta, and Xinbo Gao. Platoon control of connected vehicles with heterogeneous model structures considering external disturbances. *Green Energy and Intelligent Transportation*, 1(3):100038, 2022.

- [142] Trista Lin, Hervé Rivano, and Frédéric Le Mouël. A survey of smart parking solutions. *IEEE Transactions on Intelligent Transportation* Systems, 18(12):3229–3253, 2017.
- [143] Hao Liu, Xingan David Kan, Steven E Shladover, Xiao-Yun Lu, and Robert E Ferlis. Modeling impacts of cooperative adaptive cruise control on mixed traffic flow in multi-lane freeway facilities. *Transportation Research Part C: Emerging Technologies*, 95:261–279, 2018.
- [144] Wei Liu, Min Hua, Zhiyun Deng, Zonglin Meng, Yanjun Huang, Chuan Hu, Shunhui Song, Letian Gao, Changsheng Liu, Bin Shuai, et al. A systematic survey of control techniques and applications in connected and automated vehicles. *IEEE Internet of Things Journal*, 2023.
- [145] Seng W Loke. Cooperative automated vehicles: A review of opportunities and challenges in socially intelligent vehicles beyond networking. *IEEE Transactions on Intelligent Vehicles*, 4(4):509–518, 2019.
- [146] Guillaume Lozenguez, Lounis Adouane, Aurélie Beynier, Philippe Martinet, and Abdel-Illah Mouaddib. Map partitioning to approximate an exploration strategy in mobile robotics. In Advances on Practical Applications of Agents and Multiagent Systems: 9th International Conference on Practical Applications of Agents and Multiagent Systems, pages 63–72. Springer, 2011.
- [147] Lukas Luft, Tobias Schubert, Stergios I Roumeliotis, and Wolfram Burgard. Recursive decentralized collaborative localization for sparsely communicating robots. In *Robotics: science and systems*. New York, 2016.
- [148] Jiaqi Ma, Xiaopeng Li, Steven Shladover, Hesham A Rakha, Xiao-Yun Lu, Ramanujan Jagannathan, and Daniel J Dailey. Freeway speed harmonization. *IEEE Transactions on Intelligent Vehicles*, 1(1):78–89, 2016.
- [149] Lifeng Ma, Zidong Wang, Qing-Long Han, and Yurong Liu. Consensus control of stochastic multi-agent systems: a survey. *Science China Information Sciences*, 60:1–15, 2017.
- [150] Zachary MacHardy, Ashiq Khan, Kazuaki Obana, and Shigeru Iwashina. V2x access technologies: Regulation, research, and remaining challenges. *IEEE Communications Surveys & Tutorials*, 20(3):1858–1877, 2018.

- [151] Sumbal Malik, Manzoor Ahmed Khan, and Hesham El-Sayed. Collaborative autonomous driving—a survey of solution approaches and future challenges. *Sensors*, 21(11):3783, 2021.
- [152] Stefano Mariani, Giacomo Cabri, and Franco Zambonelli. Coordination of autonomous vehicles: taxonomy and survey. ACM Computing Surveys (CSUR), 54(1):1–33, 2021.
- [153] Dan Marinescu, Jan Čurn, Mélanie Bouroche, and Vinny Cahill. Onramp traffic merging using cooperative intelligent vehicles: A slotbased approach. In 2012 15th International IEEE Conference on Intelligent Transportation Systems, pages 900–906, 2012.
- [154] Gian Luca Mariottini, Fabio Morbidi, Domenico Prattichizzo, George J Pappas, and Kostas Daniilidis. Leader-follower formations: Uncalibrated vision-based localization and control. In *Proceedings* 2007 IEEE International Conference on Robotics and Automation, pages 2403–2408. IEEE, 2007.
- [155] Jonas Mårtensson, Assad Alam, Sagar Behere, Muhammad Altamash Ahmed Khan, Joakim Kjellberg, Kuo-Yun Liang, Henrik Pettersson, and Dennis Sundman. The development of a cooperative heavy-duty vehicle for the gcdc 2011: Team scoop. *IEEE Transactions on Intelligent Transportation Systems*, 13(3):1033–1049, 2012.
- [156] Stefano Masi, Philippe Xu, Philippe Bonnifait, and Sio-Song Ieng. Augmented perception with cooperative roadside vision systems for autonomous driving in complex scenarios. In 2021 IEEE International Intelligent Transportation Systems Conference (ITSC), pages 1140– 1146. IEEE, 2021.
- [157] Moneim Massar, Imran Reza, Syed Masiur Rahman, Sheikh Muhammad Habib Abdullah, Arshad Jamal, and Fahad Saleh Al-Ismail. Impacts of autonomous vehicles on greenhouse gas emissions—positive or negative? International Journal of Environmental Research and Public Health, 18(11):5567, 2021.
- [158] Carlos Massera Filho, Marco H Terra, and Denis F Wolf. Safe optimization of highway traffic with robust model predictive control-based cooperative adaptive cruise control. *IEEE Transactions on Intelligent Transportation Systems*, 18(11):3193–3203, 2017.
- [159] Mathworks. Automated driving toolbox. https://fr.mathworks. com/help/driving/.

- [160] Gereon Meyer and Stefan Deix. Research and innovation for automated driving in germany and europe. *Road vehicle automation*, pages 71–81, 2014.
- [161] James B Michael, Datta N Godbole, John Lygeros, and Raja Sengupta. Capacity analysis of traffic flow over a single-lane automated highway system. *Journal of Intelligent Transportation System*, 4(1-2):49–80, 1998.
- [162] François Michaud, Dominic Létourneau, Matthieu Guilbert, and Jean-Marc Valin. Distributed motion coordination of multiple robots. In Proceedings of the 2002 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), Lausanne, Switzerland, pages 2740–2745, 2002.
- [163] Vicente Milanés, Steven E. Shladover, John Spring, Christopher Nowakowski, Hiroshi Kawazoe, and Masahide Nakamura. Cooperative adaptive cruise control in real traffic situations. *IEEE Transactions on Intelligent Transportation Systems*, 15(1):296–305, 2014.
- [164] Johan Miorner and Michaela Trippl. Embracing the future: path transformation and system reconfiguration for self-driving cars in west sweden. *European Planning Studies*, 27(11):2144–2162, 2019.
- [165] Umberto Montanaro, Shilp Dixit, Saber Fallah, Mehrdad Dianati, Alan Stevens, David Oxtoby, and Alexandros Mouzakitis. Towards connected autonomous driving: review of use-cases. *Vehicle system* dynamics, 57(6):779–814, 2019.
- [166] Sara Moridpour, Majid Sarvi, and Geoff Rose. Lane changing models: a critical review. *Transportation letters*, 2(3):157–173, 2010.
- [167] Dominik Moser, Roman Schmied, Harald Waschl, and Luigi del Re. Flexible spacing adaptive cruise control using stochastic model predictive control. *IEEE Transactions on Control Systems Technology*, 26(1):114–127, 2017.
- [168] Dominik Moser, Harald Waschl, Harald Kirchsteiger, Roman Schmied, and Luigi Del Re. Cooperative adaptive cruise control applying stochastic linear model predictive control strategies. In 2015 European Control Conference (ECC), pages 3383–3388. IEEE, 2015.
- [169] Mujahid Muhammad and Ghazanfar Ali Safdar. 5g-based v2v broadcast communications: A security perspective. Array, 11:100084, 2021.

- [170] Richard M. Murray. Recent Research in Cooperative Control of Multivehicle Systems. Journal of Dynamic Systems, Measurement, and Control, 129(5):571–583, 05 2007.
- [171] Elnaz Namazi, Jingyue Li, and Chaoru Lu. Intelligent intersection management systems considering autonomous vehicles: A systematic literature review. *IEEE Access*, 7:91946–91965, 2019.
- [172] Rudy R Negenborn and Jose Maria Maestre. Distributed model predictive control: An overview and roadmap of future research opportunities. *IEEE Control Systems Magazine*, 34(4):87–97, 2014.
- [173] Jianjun Ni, Yinan Chen, Yan Chen, Jinxiu Zhu, Deena Ali, and Weidong Cao. A survey on theories and applications for self-driving cars based on deep learning methods. *Applied Sciences*, 10(8):2749, 2020.
- [174] Zifei Nie and Hooman Farzaneh. Adaptive cruise control for ecodriving based on model predictive control algorithm. *Applied Sciences*, 10(15):5271, 2020.
- [175] Fabrice R Noreils. Toward a robot architecture integrating cooperation between mobile robots: Application to indoor environment. The international journal of robotics research, 12(1):79–98, 1993.
- [176] Christopher Nowakowski, Jessica O'Connell, Steven E Shladover, and Delphine Cody. Cooperative adaptive cruise control: Driver acceptance of following gap settings less than one second. In *Proceedings* of the human factors and ergonomics society annual meeting, volume 54, pages 2033–2037. SAGE Publications Sage CA: Los Angeles, CA, 2010.
- [177] Hiroshi Ohta and Brendan F.D. Barrett. Politics of climate change and energy policy in japan: Is green transformation likely? *Earth* System Governance, 17:100187, 2023.
- [178] Markos Papageorgiou, Elias Kosmatopoulos, and Ioannis Papamichail. Effects of variable speed limits on motorway traffic flow. *Transportation Research Record*, 2047(1):37–48, 2008.
- [179] Lynne E Parker. Alliance: An architecture for fault tolerant, cooperative control of heterogeneous mobile robots. In Proceedings of IEEE/RSJ international conference on intelligent robots and systems (IROS'94), volume 2, pages 776–783. IEEE, 1994.

- [180] Lynne E Parker. L-alliance: Task-oriented multi-robot learning in behavior-based systems. Advanced Robotics, 11(4):305–322, 1996.
- [181] Lynne E Parker. Alliance: An architecture for fault tolerant multirobot cooperation. *IEEE transactions on robotics and automation*, 14(2):220-240, 1998.
- [182] Scott Drew Pendleton, Hans Andersen, Xinxin Du, Xiaotong Shen, Malika Meghjani, You Hong Eng, Daniela Rus, and Marcelo H Ang Jr. Perception, planning, control, and coordination for autonomous vehicles. *Machines*, 5(1):6, 2017.
- [183] Jeroen Ploeg, Elham Semsar-Kazerooni, Alejandro I Morales Medina, Jan FCM de Jongh, Jacco van de Sluis, Alexey Voronov, Cristofer Englund, Reinder J Bril, Hrishikesh Salunkhe, Álvaro Arrúe, et al. Cooperative automated maneuvering at the 2016 grand cooperative driving challenge. *IEEE Transactions on Intelligent Transportation* Systems, 19(4):1213–1226, 2017.
- [184] Jeroen Ploeg, Steven Shladover, Henk Nijmeijer, and Nathan van de Wouw. Introduction to the special issue on the 2011 grand cooperative driving challenge. *IEEE Transactions on Intelligent Transportation* Systems, 13(3):989–993, 2012.
- [185] Xiangjun Qian, Arnaud de La Fortelle, and Fabien Moutarde. A hierarchical model predictive control framework for on-road formation control of autonomous vehicles. In 2016 IEEE Intelligent Vehicles Symposium (IV), pages 376–381, 2016.
- [186] Zhihua Qu. Cooperative control of dynamical systems: applications to autonomous vehicles, volume 3. Springer, 2009.
- [187] Zhihua Qu, Jing Wang, and Richard A. Hull. Cooperative control of dynamical systems with application to autonomous vehicles. *IEEE Transactions on Automatic Control*, 53(4):894–911, 2008.
- [188] Rajesh Rajamani. Vehicle dynamics and control. Springer Science & Business Media, 2011.
- [189] Asim Rasheed, Haleemah Zia, Farhan Hashmi, Umair Hadi, Warda Naim, and Sana Ajmal. Fleet & convoy management using vanet. *Journal of Computer Networks*, 1(1):1–9, 2013.
- [190] Andreas Rauch, Felix Klanner, Ralph Rasshofer, and Klaus Dietmayer. Car2x-based perception in a high-level fusion architecture for

cooperative perception systems. In 2012 IEEE Intelligent Vehicles Symposium, pages 270–275. IEEE, 2012.

- [191] Ioannis Rekleitis, Gregory Dudek, and Evangelos Milios. Probabilistic cooperative localization and mapping in practice. In 2003 IEEE International Conference on Robotics and Automation (Cat. No. 03CH37422), volume 2, pages 1907–1912. IEEE, 2003.
- [192] Wei Ren and Randal W Beard. *Distributed consensus in multi-vehicle cooperative control*, volume 27. Springer, 2008.
- [193] Jackeline Rios-Torres, Andreas Malikopoulos, and Pierluigi Pisu. Online optimal control of connected vehicles for efficient traffic flow at merging roads. In 2015 IEEE 18th International Conference on Intelligent Transportation Systems, pages 2432–2437, 2015.
- [194] Jackeline Rios-Torres and Andreas A Malikopoulos. Automated and cooperative vehicle merging at highway on-ramps. *IEEE Transactions* on Intelligent Transportation Systems, 18(4):780–789, 2016.
- [195] Jackeline Rios-Torres and Andreas A. Malikopoulos. A survey on the coordination of connected and automated vehicles at intersections and merging at highway on-ramps. *IEEE Transactions on Intelligent Transportation Systems*, 18(5):1066–1077, 2017.
- [196] ROS. Ros. https://www.ros.org.
- [197] Francisca Rosique, Pedro J Navarro, Carlos Fernández, and Antonio Padilla. A systematic review of perception system and simulators for autonomous vehicles research. *Sensors*, 19(3):648, 2019.
- [198] Ankur Sarker, Haiying Shen, Mizanur Rahman, Mashrur Chowdhury, Kakan Dey, Fangjian Li, Yue Wang, and Husnu S Narman. A review of sensing and communication, human factors, and controller aspects for information-aware connected and automated vehicles. *IEEE transactions on intelligent transportation systems*, 21(1):7–29, 2019.
- [199] Andrea Sassi and Franco Zambonelli. Coordination infrastructures for future smart social mobility services. *IEEE Intelligent Systems*, 29(5):78–82, 2014.
- [200] SCANR. Scaner-avsimulation. https://www.avsimulation.com/ scaner/.

- [201] Paolo Schito et al. Numerical and experimental investigation on vehicles in platoon. SAE International Journal of Commercial Vehicles, 5(2012-01-0175):63-71, 2012.
- [202] Gunther K. Schmidt and Bernd Posch. A two-layer control scheme for merging of automated vehicles. In *The 22nd IEEE Conference on Decision and Control*, pages 495–500, 1983.
- [203] W.J. Scholte, P.W.A. Zegelaar, and H. Nijmeijer. A control strategy for merging a single vehicle into a platoon at highway on-ramps. *Transportation Research Part C: Emerging Technologies*, 136:103511, 2022.
- [204] Wilko Schwarting, Javier Alonso-Mora, and Daniela Rus. Planning and decision-making for autonomous vehicles. Annual Review of Control, Robotics, and Autonomous Systems, 1:187–210, 2018.
- [205] G. Schöner, M. Dose, and C. Engels. Dynamics of behavior: Theory and applications for autonomous robot architectures. *Robotics and Autonomous Systems*, 16(2):213–245, 1995. Moving the Frontiers between Robotics and Biology.
- [206] Antonio Sciarretta, Giovanni De Nunzio, and Luis Leon Ojeda. Optimal ecodriving control: Energy-efficient driving of road vehicles as an optimal control problem. *IEEE Control Systems Magazine*, 35(5):71– 90, 2015.
- [207] Yunli Shao and Zongxuan Sun. Robust eco-cooperative adaptive cruise control with gear shifting. In 2017 American Control Conference (ACC), pages 4958–4963. IEEE, 2017.
- [208] Steven E Shladover, Christopher Nowakowski, Xiao-Yun Lu, and Robert Ferlis. Cooperative adaptive cruise control: Definitions and operating concepts. *Transportation Research Record*, 2489(1):145–152, 2015.
- [209] Bruno Siciliano, Oussama Khatib, and Torsten Kröger. Springer handbook of robotics, volume 200. Springer, 2008.
- [210] Reid Simmons, Sanjiv Singh, David Hershberger, Josue Ramos, and Trey Smith. First results in the coordination of heterogeneous robots for large-scale assembly. In *Experimental Robotics VII*, pages 323–332. Springer, 2001.
- [211] Mechnical Simulation. Trucksim autonomous platooning.
- [212] Katrin Sjoberg, Peter Andres, Teodor Buburuzan, and Achim Brakemeier. Cooperative intelligent transport systems in europe: Current deployment status and outlook. *IEEE Vehicular Technology Maga*zine, 12(2):89–97, 2017.
- [213] Nathan Sorensen and Wei Ren. A unified formation control scheme with a single or multiple leaders. In 2007 American Control Conference, pages 5412–5418, 2007.
- [214] Thomas Stanger and Luigi del Re. A model predictive cooperative adaptive cruise control approach. In 2013 American Control Conference, pages 1374–1379. IEEE, 2013.
- [215] Mitja Stiglic, Niels Agatz, Martin Savelsbergh, and Mirko Gradisar. Enhancing urban mobility: Integrating ride-sharing and public transit. Computers & Operations Research, 90:12–21, 2018.
- [216] SUMO. Simulation of urban mobility (sumo). https://sumo.dlr. de/docs/index.html.
- [217] Xiaoqiang Sun, F Richard Yu, and Peng Zhang. A survey on cybersecurity of connected and autonomous vehicles (cavs). *IEEE Transactions on Intelligent Transportation Systems*, 23(7):6240–6259, 2021.
- [218] Synopsys. The 6 levels of vehicle autonomy explained. https://www. synopsys.com/automotive/autonomous-driving-levels.html#e.
- [219] Alireza Talebpour, Hani S Mahmassani, and Samer H Hamdar. Speed harmonization: Evaluation of effectiveness under congested conditions. *Transportation research record*, 2391(1):69–79, 2013.
- [220] VIRGINIA TECH. Virginia tech team qualifies as darpa urban challenge semi-finalist. https://news.vt.edu/articles/2007/08/ 2007-429.html.
- [221] Sebastian Thrun, Mike Montemerlo, Hendrik Dahlkamp, David Stavens, Andrei Aron, James Diebel, Philip Fong, John Gale, Morgan Halpenny, Gabriel Hoffmann, Kenny Lau, Celia Oakley, Mark Palatucci, Vaughan Pratt, Pascal Stang, Sven Strohband, Cedric Dupont, Lars-Erik Jendrossek, Christian Koelen, Charles Markey, Carlo Rummel, Joe van Niekerk, Eric Jensen, Philippe Alessandrini, Gary Bradski, Bob Davies, Scott Ettinger, Adrian Kaehler, Ara Nefian, and Pamela Mahoney. *Stanley: The Robot That Won the DARPA Grand Challenge*, pages 1–43. Springer Berlin Heidelberg, Berlin, Heidelberg, 2007.

- [222] Kailin Tong, Zlatan Ajanovic, and Georg Stettinger. Overview of tools supporting planning for automated driving. In 2020 IEEE 23rd International Conference on Intelligent Transportation Systems (ITSC), pages 1–8. IEEE, 2020.
- [223] A Toschi, M Sanic, J Leng, Q Chen, C Wang, and M Guo. Characterizing perception module performance and robustness in productionscale autonomous driving syste. Proceedings of the 16th IFIP WG 10.3 International Conference on Network and Parallel Computing, pages 235–247, 2019.
- [224] The European Land Robot Trial. Elrob- the european land robot trial. https://www.elrob.org.
- [225] Sadayuki Tsugawa. An introduction to demo 2000: The cooperative driving scenario. *IEEE Intelligent Systems*, 15(04):78–79, 2000.
- [226] Sadayuki Tsugawa. An overview on an automated truck platoon within the energy its project. *IFAC Proceedings Volumes*, 46(21):41– 46, 2013. 7th IFAC Symposium on Advances in Automotive Control.
- [227] Sadayuki Tsugawa. Results and issues of an automated truck platoon within the energy its project. In 2014 IEEE Intelligent Vehicles Symposium Proceedings, pages 642–647. IEEE, 2014.
- [228] Sadayuki Tsugawa, Shin Kato, and Keiji Aoki. An automated truck platoon for energy saving. In 2011 IEEE/RSJ international conference on intelligent robots and systems, pages 4109–4114. IEEE, 2011.
- [229] Sadayuki Tsugawa, Shin Kato, Kiyohito Tokuda, Takeshi Matsui, and Haruki Fujii. A cooperative driving system with automated vehicles and inter-vehicle communications in demo 2000. In *ITSC 2001.* 2001 IEEE Intelligent Transportation Systems. Proceedings (Cat. No. 01TH8585), pages 918–923. IEEE, 2001.
- [230] Sadayuki Tsugawa, Shin Kato, Kiyohito Tokuda, Takeshi Matsui, and Hayato Fujii. Demo 2000 cooperative driving: an overview. In 8th World Congress on Intelligent Transport Systems ITS America, ITS Australia, ERTICO (Intelligent Transport Systems and Services-Europe), 2001.
- [231] Valerio Turri, Oscar Flärdh, Jonas Mårtenssont, and Karl H Johansson. Fuel-optimal look-ahead adaptive cruise control for heavy-duty vehicles. In 2018 Annual American Control Conference (ACC), pages 1841–1848. IEEE, 2018.

- [232] Elisabeth Uhlemann. Introducing connected vehicles [connected vehicles]. *IEEE vehicular technology magazine*, 10(1):23–31, 2015.
- [233] Carnegie Mellon University. Driverless vehicle makes its debut. https://www.cmu.edu/homepage/practical/2007/winter/ meet-the-boss.shtml.
- [234] Ardalan Vahidi and Antonio Sciarretta. Energy saving potentials of connected and automated vehicles. Transportation Research Part C: Emerging Technologies, 95:822–843, 2018.
- [235] Bart Van Arem, Cornelie JG Van Driel, and Ruben Visser. The impact of cooperative adaptive cruise control on traffic-flow characteristics. *IEEE Transactions on intelligent transportation systems*, 7(4):429–436, 2006.
- [236] Pravin Varaiya. A question about hierarchical systems. In System theory: modeling, analysis and control, pages 313–324. Springer, 2000.
- [237] Matteo Vasirani and Sascha Ossowski. A market-inspired approach for intersection management in urban road traffic networks. *Journal* of Artificial Intelligence Research, 43:621–659, 2012.
- [238] J. M. Vilca Ventura. Safe and flexible hybrid control architecture for the navigation in formation of a group of vehicles. PhD thesis, Université Blaise Pascal-Clermont-Ferrand II, 2015.
- [239] J. M. Vilca Ventura, Lounis Adouane, and Youcef Mezouar. A novel safe and flexible control strategy based on target reaching for the navigation of urban vehicles. *Robotics and Autonomous Systems*, 70:215– 226, 2015.
- [240] J. M. Vilca Ventura, Lounis Adouane, and Youcef Mezouar. Adaptive leader-follower formation in cluttered environment using dynamic target reconfiguration. In *Distributed Autonomous Robotic Systems: The* 12th International Symposium, pages 237–254. Springer, 2016.
- [241] J. M. Vilca Ventura, Lounis Adouane, and Youcef Mezouar. Stable and flexible multi-vehicle navigation based on dynamic inter-target distance matrix. *IEEE Transactions on Intelligent Transportation* Systems, 20(4):1416–1431, 2019.
- [242] Yu A Vershinin and Yao Zhan. Vehicle to vehicle communication: dedicated short range communication and safety awareness. In 2020 Systems of Signals Generating and Processing in the Field of on Board Communications, pages 1–6. IEEE, 2020.

- [243] Wierwille W., W., Hanowski R., J., Hanley J., M., et al. Identification and evaluation of driver errors: Overview and recommendation. *Technical report*, 2002.
- [244] Zia Wadud, Don MacKenzie, and Paul Leiby. Help or hindrance? the travel, energy and carbon impacts of highly automated vehicles. *Transportation Research Part A: Policy and Practice*, 86:1–18, 2016.
- [245] Chunping Wang, Jiaqi Wang, Ping Wu, and Jinfeng Gao. Consensus problem and formation control for heterogeneous multi-agent systems with switching topologies. *Electronics*, 11(16):2598, 2022.
- [246] Fei-Yue Wang, Nanning Zheng, Li Li, Jingmin Xin, Xiao Wang, Linhai Xu, Bin Tian, Guozheng Wu, Zhaotian Zhang, Chenghong Wang, et al. China's 12-year quest of autonomous vehicular intelligence: The intelligent vehicles future challenge program. *IEEE Intelligent Transportation Systems Magazine*, 13(2):6–19, 2021.
- [247] Jiadai Wang, Jiajia Liu, and Nei Kato. Networking and communications in autonomous driving: A survey. *IEEE Communications* Surveys & Tutorials, 21(2):1243–1274, 2018.
- [248] Ziran Wang, Yougang Bian, Steven E Shladover, Guoyuan Wu, Shengbo Eben Li, and Matthew J Barth. A survey on cooperative longitudinal motion control of multiple connected and automated vehicles. *IEEE Intelligent Transportation Systems Magazine*, 12(1):4–24, 2019.
- [249] Ziran Wang, Guoyuan Wu, and Matthew J Barth. A review on cooperative adaptive cruise control (cacc) systems: Architectures, controls, and applications. In 2018 21st International Conference on Intelligent Transportation Systems (ITSC), pages 2884–2891. IEEE, 2018.
- [250] Ziran Wang, Guoyuan Wu, Peng Hao, Kanok Boriboonsomsin, and Matthew Barth. Developing a platoon-wide eco-cooperative adaptive cruise control (cacc) system. In 2017 IEEE Intelligent Vehicles Symposium (IV), pages 1256–1261. IEEE, 2017.
- [251] Hermann Winner, Stephan Hakuli, Felix Lotz, and Christina Singer. Handbook of driver assistance systems. Springer International Publishing Amsterdam, The Netherlands:, 2014.
- [252] L. Wischhof, A. Ebner, and H. Rohling. Information dissemination in self-organizing intervehicle networks. *IEEE Transactions on Intelligent Transportation Systems*, 6(1):90–101, 2005.

- [253] Lars Wischhof, André Ebner, and Hermann Rohling. Information dissemination in self-organizing intervehicle networks. *IEEE Trans*actions on intelligent transportation systems, 6(1):90–101, 2005.
- [254] Wouter Klein Wolterink, Geert Heijenk, and Georgios Karagiannis. Automated merging in a cooperative adaptive cruise control (cacc) system. In *Fifth ERCIM workshop on eMobility*, page 23. Citeseer, 2011.
- [255] G Wu, P Hao, Z Wang, Y Jiang, and al. Eco-approach and ecodeparture planning study. *Technical Report, Crash Avoidance Metrics Partnership*, pages 1–16, 2021.
- [256] Yongjie Xue, Chuan Ding, Bin Yu, and Wensa Wang. A platoon-based hierarchical merging control for on-ramp vehicles under connected environment. *IEEE Transactions on Intelligent Transportation Systems*, 23(11):21821–21832, 2022.
- [257] Hao Yang, Hesham Rakha, and Mani Venkat Ala. Eco-cooperative adaptive cruise control at signalized intersections considering queue effects. *IEEE Transactions on Intelligent Transportation Systems*, 18(6):1575–1585, 2016.
- [258] Kai Yang, Xiaolin Tang, Yechen Qin, Yanjun Huang, Hong Wang, and Huayan Pu. Comparative study of trajectory tracking control for automated vehicles via model predictive control and robust h-infinity state feedback control. *Chinese Journal of Mechanical Engineering*, 34(1):1–14, 2021.
- [259] Yang Quan Chen and Zhongmin Wang. Formation control: a review and a new consideration. pages 3181–3186. IEEE, 2005.
- [260] Ekim Yurtsever, Jacob Lambert, Alexander Carballo, and Kazuya Takeda. A survey of autonomous driving: Common practices and emerging technologies. *IEEE access*, 8:58443–58469, 2020.
- [261] Shin'ichi Yuta and Suparerk Premvuti. Coordinating autonomous and centralized decision making to achieve cooperative behaviors between multiple mobile robots. In *Proceedings of the IEEE/RSJ international* conference on intelligent robots and systems, volume 3, pages 1566– 1574. IEEE, 1992.
- [262] Linlin Zhang, Feng Chen, Xiaoxiang Ma, and Xiaodong Pan. Fuel economy in truck platooning: a literature overview and directions for future research. *Journal of Advanced Transportation*, 2020, 2020.

- [263] Fei Zhao, Yu Liu, Jian Wang, and Li Wang. Distributed model predictive longitudinal control for a connected autonomous vehicle platoon with dynamic information flow topology. In *Actuators*, volume 10, page 204. MDPI, 2021.
- [264] Li Zhao and Jian Sun. Simulation framework for vehicle platooning and car-following behaviors under connected-vehicle environment. *Procedia-Social and Behavioral Sciences*, 96:914–924, 2013.
- [265] Yang Zheng, Shengbo Eben Li, Jianqiang Wang, Dongpu Cao, and Keqiang Li. Stability and scalability of homogeneous vehicular platoon: Study on the influence of information flow topologies. *IEEE Transactions on intelligent transportation systems*, 17(1):14–26, 2015.
- [266] Yang Zheng, Shengbo Eben Li, Jianqiang Wang, Keqiang Li, et al. Influence of information flow topology on closed-loop stability of vehicle platoon with rigid formation. In 17th International IEEE Conference on Intelligent Transportation Systems (ITSC), pages 2094–2100. IEEE, 2014.
- [267] Mofan Zhou, Xiaobo Qu, and Sheng Jin. On the impact of cooperative autonomous vehicles in improving freeway merging: a modified intelligent driver model-based approach. *IEEE Transactions on Intelligent Transportation Systems*, 18(6):1422–1428, 2016.
- [268] Zhongbao Zhou, Ling Lin, Helu Xiao, Chaoqun Ma, and Shijian Wu. Stochastic network dea models for two-stage systems under the centralized control organization mechanism. *Computers & Industrial Engineering*, 110:404–412, 2017.
- [269] Hui Zhu and Zhigang Yang. Simulation of the aerodynamic interaction of two generic sedans moving very closely. In 2011 International Conference on Electric Information and Control Engineering, pages 2595–2600. IEEE, 2011.
- [270] Jie Zhu, Said Easa, and Kun Gao. Merging control strategies of connected and autonomous vehicles at freeway on-ramps: a comprehensive review. Journal of Intelligent and Connected Vehicles, 5(2):99– 111, 2022.
- [271] Zheng Ze Zhu. Hierarchical and Hybrid Decisional Control Architecture for Cooperative Navigation of Autonomous Vehicles in Complex Environments/Situations. PhD thesis, Université Clermont Auvergne, 2022.

Ai-je agi comme il le fallait ?
Tu as agi comme il le fallait, dit le garçon nommé Corbeau. Tu as fait ce qui était juste. Personne n'aurait pu agir aussi bien que toi. Tu es le garçon le plus courageux du monde réel, tu sais.
Mais je ne sais toujours pas ce que cela signifie, vivre, dis-je.
Regarde le tableau, déclare-t-il. Et écoute le vent
... Tu devrais dormir un peu, dit le garçon nommé Corbeau. Quand tu te réveilleras, tu feras partie d'un monde nouveau.

> Kafka sur rivage Haruki Murakami

Abstract

The widespread use of Intelligent Transportation Systems (ITS) primarily stems from the imperative to mitigate human errors contributing to accidents. However, addressing the "individual" Automated Vehicles (AVs) alone remains insufficient, given that a number of situations require the "coordination" of the relative movements of AVs. Within the Multi-Vehicle System (MVS) paradigm, AVs derive advantages from connectivity and road preview information. Consequently, they can sense more accurately, process more information, and can be more tightly controlled. The collaborative assessment of safety within MVS allows for the establishment of advanced collision-avoidance strategies, particularly in challenging scenarios such as intersection crossings and on-ramp mergings. Moreover, MVS technology facilitates the reduction of gaps between vehicles, enhancing road capacity and traffic flow. The MVS's shorter response time enables improved control of AV dynamics, paving the way for promising energy-centric strategies. The main aim of the research done in this Ph.D. manuscript is to propose a safe and energy efficient decision/control architecture for MVS that navigates in dynamic and complex environment such as on-ramp merging and multi-road navigation in highway. on the multi-controller foundational control architecture a Cooperative Multi-Controller Architecture (C-MCA) is proposed. The first part of the proposed C-MCA deals with the decision-making level. This level involves a multi-behavior decision-making strategy, responsible for activating the MVS' suitable behavior based on the safety metric. Two distinct behaviors are proposed: (a) The nominal behavior, designed for executing the merging scenario while adhering to the individual goals of the MVS vehicles. It is activated when no collision risk is detected, (b) The cooperative behavior, activated by the decision-making level when both of the safety and the energy efficiency requirement are not satisfied by the nominal behavior. Its objective is to solve the merging conflict by generating a safe and energy efficient passing order of the MVS vehicles in the merging zone. The second part of the proposed C-MCA focuses on the local trajectory planning level. Each behavior is assigned a dedicated controller tailored to meet its specific requirements. For instance, the cooperative behavior controller has the responsibility of translating the vehicle's passing order into feasible dynamics (i.e., trajectory, velocity, etc.). The translation task is facilitated through the proposed dynamic formation reconfiguration strategy. In essence, the approach leverages the multi-vehicle system's formation navigation capabilities to conceptualize the merging challenge as a formation reconfiguration problem. A formalization of the formation reconfiguration problem is presented, employing a flexible and generic formal approach. The proposed dynamic formation reconfiguration strategy, employs virtual dynamic targets to ensure a secure reshaping of the formation toward the desired configuration based on the selected passing order. The overall architecture's performance is assessed through co-simulation using Matlab/Simulink and SCANeR studio.

Keywords: Highway cooperative navigation, Multi-vehicle system, Multi-controller architecture,

Multi-behavior decision-making strategy, Dynamic formation reconfiguration strategy.

Résumé

L'engouement important pour les systèmes de transport intelligent est justifié principalement par l'impératif de réduire, voire annihiler, les erreurs humaines induisant les accidents. Cependant, s'attaquer uniquement aux Véhicules Autonomes (VAs) "individuels" demeure insuffisant, étant donné que plusieurs situations nécessitent la "coordination" des mouvements relatifs des VAs. Dans le paradigme du Système Multi-Véhicules (SMV), les VAs bénéficient de l'information issue de leur connectivité. Par conséquent, ils peuvent détecter plus précisément, traiter davantage d'informations et être contrôlés de manière plus précise. L'évaluation collaborative de la sécurité au sein du SMV permet l'établissement de stratégies avancées en matière d'évitement de collision, en particulier dans des scénarios complexes tels que les croisements au sein d'intersections et les insertions sur les entrées d'autoroute. De plus, la technologie SMV facilite la réduction des espaces entre les véhicules, améliorant ainsi la capacité et la fluidité du trafic routier. Le temps de réponse plus court du SMV permet un meilleur contrôle de la dynamique des VAs, ouvrant la voie à des stratégies énergétiques prometteuses. L'objectif principal des travaux de recherche constituant ce manuscrit de doctorat est de proposer une architecture de décision/contrôle sûre et peu énergivore pour le SMV naviguant dans des environnements dynamiques et complexes. Inspirée des architectures multi-contrôleurs, une Architecture Multi-Contrôleurs Coopérative est proposée. La première partie de l'architecture proposée concerne le niveau de prise de décision. Ce niveau, impliquant une stratégie de prise de décision à plusieurs comportements, est responsable de l'activation du comportement approprié du SMV en fonction de la métrique de sécurité. Deux comportements distincts sont proposés : (a) Le comportement nominal, conçu pour réaliser le scénario d'insertion tout en respectant les objectifs individuels des véhicules formant le SMV, est activé lorsqu'aucun risque de collision est détecté, (b) Le comportement coopératif est activé par le niveau de prise de décision lorsque l'exigence de sécurité n'est pas satisfaite par le comportement nominal. Son objectif est de résoudre le conflit lors de l'insertion en générant un ordre de passage sûr et économe énergétiquement pour les véhicules composant le SMV dans la zone d'insertion. La deuxième partie de l'architecture proposée se concentre sur le niveau de planification de trajectoire locale. Chaque comportement se voit attribuer un contrôleur dédié, concu pour répondre à ses besoins spécifiques. Par exemple, le contrôleur du comportement coopératif est chargé de traduire l'ordre de passage des véhicules en dynamiques réalisables (trajectoire, vitesse, etc.). Cette tâche d'obtention de dynamiques réalisables est facilitée par la stratégie de reconfiguration dynamique de la formation proposée. En substance, l'approche tire parti des capacités de navigation en formation du SMV pour conceptualiser des manœuvres coopératives ayant trait d'insertion en milieu autoroutier, comme un problème de reconfiguration de formation. Une formalisation du problème de reconfiguration de la formation est présentée, utilisant une approche formelle, flexible et générique. La stratégie proposée utilise des cibles dynamiques virtuelles pour garantir une reconfiguration sécurisée de la formation, ceci, de sa forme initiale vers la configuration souhaitée par rapport à l'ordre de passage sélectionné. La performance de l'architecture globale a été évaluée par co-simulation en utilisant Matlab/Simulink et SCANeR studio.

Mots-clés : Navigation coopérative en milieu autoroutier, Système multi-véhicules, Architecture multi-contrôleurs, Stratégie multi-comportementales de prise de décision, Stratégie de reconfiguration dynamique d'une formation.